# Modestly Modular vs. Massively Modular Approaches to Phonology

## Vanessa McIntosh

A thesis submitted for the degree of

**Doctor of Philosophy** 

School of English Literature, Language and Linguistics

February 2020

#### Abstract

This thesis considers the extent to which phonology (that is, the phonological processor) can be considered a module of the mind. It is divided into two parts. In the first, an approach of 'modest' modularity owing to Fodor (1983) is explored. In the second, the 'massive' modularity model, due to evolutionary psychologists in general, but Caruthers (2006a) in particular, is examined. Whilst for Fodor (1983, 2000) the mind is only modular around its periphery (i.e. only its input and output systems are modules), for massive modularists the mind is modular through and through, up to and including its central capacities. The two authors, therefore, by extension differ in their definitions of modularity: Fodor (1983, 2000) sees 'informational encapsulation' as being essential to modularity, whereas for Carruthers (2006) domain specificity is much more important. The thesis concludes that whether phonology is a module or not then depends on the definition of modularity, for although a substance-free phonology which has no phonetic grounding could count as strong evidence for the informational encapsulation (and therefore the modularity) of phonology by Fodor's (1983) standards, some aphasiology data has shown that semantic treatments can remediate phonological word finding difficulties in aphasia, which would be indicative that phonology is not domain-specific, and therefore amodular in the terms of massive modularists like Carruthers (2006a).<sup>1</sup> In order to answer whether phonology is modular, then, we must first define, once and for all, what modularity (and indeed phonology) means. Until then, the debate remains, and so does my resolve to settle it.

<sup>&</sup>lt;sup>1</sup> This result is challenging to Jackendoff's (2002) parallel architecture of the language faculty too, I argue, which requires the phonological (integrative) module to be *both* informationally encapsulated and domain specific.

## Dedication

To John. I dedicate this thesis to you.

#### Acknowledgements

I wish I had an 80,000 word limit for this acknowledgements section alone, as in this short space I doubt I'll ever do you all justice. Now of course it's customary that one thanks their primary supervisor first, but S.J. really does deserve to come at the top of this page for many more reasons than that. I've worried all the way through that he would leave before I did but alas, we stuck together 'til the bitter end, with him retiring the very same day I submitted. THANK YOU UNIVERSE.

I wouldn't be here if he hadn't have introduced me to Fodor's (1983) modularity of mind thesis in the third year of my BA (Hons), and I definitely wouldn't be here if it wasn't for the lol catz on his lecture slides (j.k. – although, they were canny good). I've been turning up at his office door unannounced for 6 years now with a knock, knock and an 'areet', and truth be told, I'm rather going to miss it. He's been a supervisor, a mentor – hell – even a friend (?). If you haven't had the pleasure of meeting the man yet I strongly urge you to slide into his inbox. My fondest memories of Newcastle Uni have to do with him (even if I have avoided every single social 'cause I can't people).

Special thanks are also due to my secondary supervisor Joel Wallenberg. He's a Jack of all trades and master of all (no, trust me – he is). He's one *the* smartest people I've had the pleasure of picking the brains of and is super sweet too, which is always a bonus, right? I'm so glad he came on board when we switched around my supervision. The quantitative part of my thesis won't have been possible if it weren't for Joel (I hadn't touched statistics for 10 years!), and nor would my musings on language change. Thank you, thank you, thank you for all that you've done (and been) – you really are a genius.

I've also had help from (and many a girl talk with) Danielle Turton, and for her I'll forever be grateful. I loved teaching on her (and Joel and Rachel's) modules, but I'm especially thankful for the times I got to teach phonology (thanks, Geoff!) to supplement my AHRC funding. Big up the Arts and Humanities Research Council and the Northern Bridge Doctoral Training Partnership also, by the way, for it was through them that this research was funded. I still can't quite believe I won a scholarship (I've never won anything else in my life), and you better believe that that'll be something I tell the grandchildren.

Thanks also to Julie Morris (from the School of Education, Communication and Language Sciences (ECLS) at Newcastle University) for her secondary supervision in the first couple years. She really helped me get a good grasp of the speech and language therapy literature, and the thesis is better informed for it. I'd like to direct my thanks to a number of others over at ECLS: David Howard, Janet Webster, Ghada Khattab – you've all been bomb – and all those at the Tavistock Aphasia Centre, of course, where I was fortunate enough to observe therapists at work.

I'd also like to thank the NHS. Although I wasn't granted ethical approval to conduct a study of my own with anomic aphasics just yet (and probably rightly so as I'm not a qualified clinican!) they kindly allowed me to sit in on other people's so I could get a feel for how it all goes down. I'm also keen to give thanks for the treatment I myself have received at the hands of the NHS. Without their support (and the support of S.J. and Joel), I hand on heart don't know whether I would have been mentally or physically well enough to bring this project to completion.

I sadly didn't manage to meet with Jerry Fodor (online or off), but I was lucky enough to talk to Peter Carruthers on a couple of occasions, and I'm as happy as Larry I did. Thanks also, of course, to Pavel Iosad. Not only are he and Geoff Poole (once again, thanks Geoff!) my examiners (Pavel external and Geoff internal), his thesis was another S.J. introduced me to in Topics in Phonological Theory in 2013 and that was my first encounter of substance-free phonology. He's replied promptly to every email of mine that's been sent his way, and I'm so appreciative he's taken the time (and sent me a proof of his 2017 publication for free, come to think of it!). Thanks also, actually, to Josef Fruehwald who's in the Edinburgh gang too. He's been just as approachable – so many thanks for that, Joe!

Thank you to my parents, Angela and Michael, who together are even more protective than the patron saint of protection himself (homeboy Archangel Michael). They've saw me through some tough times, over the past 4 years, and have kept me warm and well-lit in the 'back room' where most of my writing's took place (I know, I know – God help their heating and leccy bills!). My Mam is the best sounding board there ever did be, and my Dad the best cooker of tea that'll be ready in 20 minutes. Neither of them know what the hell I'm talking about half the time, but both of them hate me pacing while I do it (sorry). Last, but by no means least, I'd like to thank John Richardson and the speech and language therapists at Walkergate Rehabilitation Centre where he was for a year after his brain injury. Not much good came of my partner ending up in a coma (to say the very least, lol), but I hope that this thesis is one of those things, and points us in the direction we need to go, to improve treatment for phonological anomia.

### **Table of Contents**

Abstract	i
Dedication	ii
Acknowledgements	iii
Table of Contents	vii
List of Tables and Figures	xii
Chapter 1: Introduction	1
Chapter 2: An Introduction to Modularity	7
1. Introduction	7
2. Natural modules vs. artificial modules	7
3. Modularity in cognitive science	9
3.1 Computational cognitive science vs. neural cognitive science	9
3.2 Cognitive modules vs. neural modules	11
3.2.1 Function vs. structure	11
3.2.2 Closing the gap between functional and structural modules	13
3.3 Cognitive modules: Some differences	14
3.3.1 Computational modules	15
3.3.1.1 Non-Darwinian modules	15
3.3.1.2 Darwinian modules	17
4. Concluding remarks	18
Chapter 3: Modest Modularity vs. Massive Modularity	21
1. Introduction	21
2. Modest modularity	22
2.1 Fodorian modules	22
2.1.1 Features of Fodorian modules	23
2.1.1.1 Dissociability and localisability	23
2.1.1.2 Mandatoriness, speed and superficiality	25
2.1.1.3 Innateness and ontogenetic determinism	27
2.1.1.4 Domain specificity	28
2.1.1.5 Inaccessibility and encapsulation	29
2.2 What is and isn't modular for Fodor	32

2.2.1 Peripheral modularity	36
2.2.2 Central nonmodularity	37
3. Massive modularity	39
3.1 Evidence for modularity	40
3.1.1 Developmental evidence	40
3.1.2 Pathological evidence	41
3.1.3 Experimental evidence	41
3.1.4 Evidence from evolutionary psychologists	43
3.2 Carruthers' arguments for massive modularity	44
3.2.1 The argument from design	44
3.2.1.1 Biological modules	44
3.2.1.2 Is the mind a biological module?	46
3.2.1.2 Carruthers' argument from design summarised	48
3.2.2 The argument from animals	48
3.2.2.1 Types of module	50
3.2.2.1.1 Learning modules	50
3.2.2.1.2 Desire modules	51
3.2.2.1.3 Other modules	51
3.2.3 The argument from computational tractability	52
3.2.3.1 Narrow-scope vs. wide-scope encapsulation	53
3.3 Carrutherian modules	56
4. Concluding remarks	57
Chapter 4: A Modestly Modular Approach to Phonology	58
1. Introduction	58
2. Substance-free phonology	59
2.1 Some assumptions about substance-free phonology	61
2.2 Some existing arguments for a substance-free phonology	61
2.2.1 Explanatory adequacy	61
2.2.2 Multiple modalities	63
2.2.3 Emergence	63
2.2.3.1 Emergent features	63
2.2.3.2 Emergent patterns	65
2.2.4 Diachrony	70
2.2.5 Gradual phonologisation	71
2.3 Alternative arguments for a substance-free phonologyviii	76

2.3.1 Spontaneous phonologisiation	76
2.3.2 The enhancement-of-clipping hypothesis	83
3. Concluding remarks	86
Chapter 5: A Massively Modular Approach to Phonology	87
1. Introduction	87
2. Models of spoken word production	88
2.1 One-step models	92
2.1.1 Morton's (1970, 1979) logogen model	92
2.2 Two-step models	94
2.2.1 Semantic and phonological levels	94
2.2.1.1 Evidence for independent semantic and phonological levels	94
2.2.1.1.1 Evidence from normal speakers	94
2.2.1.1.1.1 Speech errors	94
2.2.1.1.1.2 Tip of the tongue	95
2.2.1.1.2 Evidence from aphasic speakers	97
2.2.2 Butterworth's (1989) semantic lexicon model	98
2.2.3 Levelt et al.'s (1999) WEAVER++ model	100
2.2.4 Evidence that the lemma (syntax) is independent of phonology	102
2.2.4.1 Tip of the tongue studies	102
2.2.4.2 Neuropsychological studies	102
2.2.5 Activation within two-step models	103
2.2.5.1 Stemberger (1985) and Dell's (1986) interactive activation models	104
2.3 Phonological encoding	106
2.3.1 Shattuck-Hufnagel's (1979) interactive activation account of phonological encoded	ding
	106
2.3.1.1 How speech errors are classified	106
2.3.1.2 How speech errors arise	108
2.3.2 Dell's (1989) interactive activation account of phonological encoding	109
2.4 Concluding remarks	111
2.4.1 An alternative account owing to Jackendoff (2002)	112
2.4.1.1 The role of syntax	112
2.4.1.2 The integrative and interface modules	113
2.4.1.2.1 The interfaces	115
2.4.1.2.1.1 A side note on the phonetics-phonology interface	116
2.4.1.3 The lexicon and lexical entries	117

2.4.1.4 Lexical processing	120
3. Anomia	123
3.1 Semantic anomia	126
3.1.1 Impairment in access to conceptual-semantic representations	128
3.1.2 Permanent erasure of conceptual-semantic representations from long-term	memory
	130
3.2 Phonological anomia	133
3.2.1 Impairment in access to output lexicons from the lexical-semantic system	134
3.2.2 Impaired phoneme assembly	135
4. Anomia therapy	138
4.1 Data source and sampling	138
4.2 Data analysis	149
4.2.1 A qualitative analysis of the data	149
4.2.1.1 Primarily semantic tasks	149
4.2.1.1.1 Marshall et al. (1990)	149
4.2.1.1.2 Marshall et al. (1998)	152
4.2.1.1.3 Nettleton and Lesser (1991)	155
4.2.1.2 Semantic and phonological tasks	159
4.2.1.2.1 Best et al. (1997)	159
4.2.1.2.2 Hillis (1989)	162
4.2.1.2.3 Hillis and Caramazza (1994)	164
4.2.1.2.4 Pederson et al. (2001)	165
4.2.1.2.5 Raymer and Ellsworth (2002)	166
4.2.1.3 Primarily phonological tasks	169
4.2.1.3.1 Fink et al. (2002)	169
4.2.1.3.2 Hickin et al. (2002a)	170
4.2.1.3.3 Hickin et al. (2002b)	172
4.2.1.3.4 Jokel and Rochon (1998)	174
4.2.2 Outcome classification (treated vs. untreated)	177
4.2.2.1 Marshall et al. (1990)	177
4.2.2.2 Marshall et al. (1998)	178
4.2.2.3 Nettleton and Lesser (1991)	178
4.2.2.4 Best et al. (1997)	179
4.2.2.5 Hillis (1989)	180
4.2.2.6 Hillis and Caramazza (1994)	180

4.2.2.7 Raymer and Ellsworth (2002)	180
4.2.2.8 Hickin et al. (2002a)	181
4.2.2.9 Hickin et al. (2002b)	181
4.2.2.10 Jokel and Rochon (1998)	181
4.2.3 A quantitative analysis of the case studies	182
4.2.3.1 Statistical significance testing	182
4.2.3.2 Results and discussion	183
4.2.3.2.1 The effects on statistical significance	185
4.2.3.2.1.1 Sample size	185
4.2.3.2.1.2 Factors that affect naming	186
4.2.3.2.1.2.1 Word frequency and length	186
4.2.3.2.1.2.2 Orthography	187
4.2.3.2.2 The effects of publication bias	188
5. Concluding remarks	189
Chapter 6: Conclusion	191
References	196

## List of Tables and Figures

## Tables

Table 2.1 A table showing two different approaches to studying cognitive science10
Table 4.1 Opaque interaction between /ay/ raising and flapping (adapted from Fruehwald
2013: 117)
Table 4.2 Median /ay/ durations by context (adapted from Fruehwald 2013: 117)78
Table 5.1 KE's performance on semantic tasks (adapted from Hillis et al. 1990: 203)
Table 5.2 A breakdown of the 33 single case (and series of single case) studies of word
retrieval in Nickels (2002) that I've excluded from my analysis and my reasons for doing so
Table 5.3 IS's scores (out of 25) for the four groups at successive testing (adapted from
Marshall et al. 1990) 152
Table 5.4 Unprimed and primed naming in CM (adapted from Marshall et al. 1998: 99)
Table 5.5 Pre- and post-therapy naming of treated and control items (adapted from Marshall et
al. 1998: 100)
Table 5.6 Subjects (adapted from Nettleton and Lesser 1991: 145) 158
Table 5.7 Patient 1's performance on lexical tasks (where the stimuli for each task was 144
names for or pictures of concepts
Table 5.8 Number of items correct for WR on the Florida Noun/Verb Battery, a set of lexical
tasks for noun and verb comprehension and retrieval (adapted from Raymer and Ellsworth
2002: 1035)
Table 5.9 Background information for the participants included in the study, including
aphasia type as assessed by the Comprehensive Aphasia Test (Swinburn et al. 2002) (adapted
from Hickin et al. 2002a: 986) 170
Table 5.10 Participants' performances on a number of background assessments (adapted from
Hickin et al. 2002a: 987) 171
Table 5.11 Language assessment in percentages prior to therapies (from Hickin et al. 2002b:
72) 173
Table 5.12 A breakdown of the 10 single case studies of word retrieval relevant to this thesis

Table 5.13 A breakdown of the 9 single case studies of word retrieval that I've included in my
analysis 182
Table 5.14 The effects of semantic therapy and phonological therapy at treating phonological
anomia in the sample 183
Table 5.15 The effects of semantic therapy and phonological therapy at treating semantic
anomia in the sample 184
Table 5.16 The effects of semantic therapy and phonological therapy at treating phonological
anomia in the sample (excluding those phonological therapies that made use of orthography)
Table 5.17 The hypothetical effects of semantic therapy and phonological therapy at treating
phonological anomia if the data patterned differently 190
Table 6.1 The hypothetical effects of semantic therapy and phonological therapy at treating
phonological anomia if the data patterned the same, but the sample were ten times the size

## Figures

Figure 2.1 Hebb's (1949) neural sub-structure schema (adapted from Schnelle 2010: 9)	)
	12
Figure 2.2 The different types of modules and their relationships with one another	20
Figure 3.1 Functional zones in the brain (adapted from Daroff et al. 2012: 95)	24
Figure 3.2 A schematic illustrating the domain specificity of speech perception where (	(a)
illustrates isolated resonances and (b) those same resonances producing [d] and [g] in a	l
speechlike context (adapted from Mattingly and Liberman 1988: 69)	29
Figure 3.3 A Müller-Lyer illusion with typical central shaft and inward and outward fin	15
	32
Figure 3.4 The path of sensory information (Raven 2006: 1104)	34
Figure 3.5 Smell (Raven 2006: 1110)	36
Figure 3.6 The duck-rabbit figure (Jastrow 1899: 312)	55
Figure 4.1 Distribution of contextual variants ([Vx] and [Vy]) of a hypothetical vowel	
(Freuhwald 2013: 61)	71
Figure 4.2 Independent targets of phonetic implementation produced by phonological	
differentiation (Fruehwald 2013: 62)	72
Figure 4.3 The effect of coarticulation on shifting productions from intended targets	
(Fruehwald 2013: 63)	73

Figure 4.4 The interaction of phonological feature spreading and diachronic phonetic change
(Fruehwald 2013: 64)
Figure 4.5 The interaction of phonetic coarticulation and diachronic phonetic change
(Fruehwald 2013: 64)
Figure 4.6 The reanalysis of phonetic coarticulation as phonological feature spreading, and it
effect on the rate of phonetic change
Figure 4.7 Nucleus to glide trajectories in Victoria, British Columbia (Fruehwald 2013: 199)
Figure 4.8 /ay/ height by date of birth and context
Figure 4.9 Model estimates of /ay/ height, faceted by surface vs. flapped realisations
Figure 4.10 Model estimates of /ay/ height, faceted by /t/ vs /d/
Figure 4.11 The stem-level, word-level and phrase-levels of phonology (adapted from
Bermudez-Otero and Trousdale 2012: 700)
Figure 5.1 A depiction of the Wernicke-Lichtheim model of language processing and the
different aphasia profiles accounted by the model (Laine and Martin 2006: 5)
Figure 5.2 The logogen model of spoken word production (Nickels 1997: 11)
Figure 5.3 The proposed stages of spoken word production: Butterworth's (1989) semantic
lexicon model (from Nickels 1997: 21)
Figure 5.4 The types of process involved in speaking according to WEAVER++ (Roelofs
2000: 73)
Figure 5.5 Stemberger's (1985) interactive activation model (adapted from Stemberger 1985)
Figure 5.6 Dell's (1986) interactive activation model (adapted from Dell 1986).
Word/morpheme level nodes are initially activated from semantic representations that aren't
in the diagram. At the phoneme level on = onset, nu = nucleus and co = coda 105
Figure 5.7 Dell's lexical and word shape networks (adapted from Dell 1989). The intended
word here is 'cat' as indicated by the flag on the listing of 'cat' at the word level in the lexica
network. The dotted lines in the diagram represent connections between the lexical network
and the word shape network, and the arrows between phoneme categories in the word shape
network their sequence of activation. As in Dell's (1986) model, On = onset, Nu = nucleus
and Co = coda
Figure 5.8 Jackendoff (2002)'s model of the language faculty (Jackendoff 2002: 199)
Figure 5.9 Chomsky's model of language (ten Hacken 2019: 207) 120

#### **Chapter 1: Introduction**

Great attention has been paid to complex systems in the sciences, and for good reason. The phenomena we wish to understand are all complex, and the problems we need to solve in order to understand them are even more so. Take a balloon, for example, filled with helium gas. The balloon itself isn't necessarily complex, but how it interacts with the earth's atmosphere (e.g. how it floats because the gas which fills it (helium) is lighter than the air it displaces (which is, of course, mostly made up of nitrogen, oxygen, argon and carbon dioxide) is (Simon 2005)). The source of the balloon's complexity, therefore, isn't in itself, but in its *interaction* with a more complex system that is the earth's atmosphere (Simon 2005).

It might be fair to say then, that complexity arises when systems interact with each other in ways that amount to something more than they could ever be by themselves (Simon 2005); when they are more than just the sum of their parts, because they're a part of something else as well. Before the big bang the universe was not particularly complex. All there was in the cosmos was a gravitational singularity, that is, a point in time and space where gravity is infinite, space-time curves infinitely and the laws of physics as we know them cease to exist (Hawking 1988). It was after the fact that complexity arose, when stars (and then stellar galaxies) were formed, and solar systems within them. Our planet, the earth, is of course complex – both in terms of the elements that make it up, and its place in the solar system (and therefore the Milky Way).

It has been known for some time now that the complex systems we observe in the natural world have hierarchical structures; that they are made up of complex subsystems, which themselves are made up of complex subsystems, and so on and so forth (Simon 2005). The smaller subsystems interact with each other of course (or else they wouldn't form larger subsystems or systems) but what's interesting to this project in particular is that the frequencies associated with the subsystem interactions drop stealth the higher up the hierarchy you go (Simon 2005).

We call complex systems with this property nearly decomposable (Simon and Ando 1961), and these have been the subject of scientific research for tens to hundreds of years (Callebaut

and Rasskin-Gutman 2005). A question of interest to *cognitive* scientists especially though is this one:

If complexity in our universe at all levels generally takes this hierarchical nearly decomposable (or modular) form, is this true of a product of the universe - the mind - as well? Is the human mind (and its components) modular?

Jerry Fodor was among the first to try and answer this question and in his 1983 publication *The Modularity of Mind* he claimed that the mind is indeed modular, or it is at least around its periphery in the input and output systems. Evolutionary psychologists, however, believe Fodor's (1983, 2000) thesis of mind modularity to be too modest, and that the central systems are modular as well. The idea that the mind is modular up to and including its central capacities is what we call the massive modularity hypothesis. It was articulated and defended by Peter Carruthers in 2006, and it is against these two theses of mind modularity (Fodor's (1983, 2000) modest thesis and Carruthers' (2006a) massive one) that my research question is based.

I question in particular whether the phonological processor is modular. Being part of the language faculty (and therefore one of the input systems) both Fodor (1983, 2000) and Carruthers (2006a) would argue that it is, but they each have different definitions of modularity – and there lies the crux of the problem. For Fodor (2000), a modest modularist, informational encapsulation is the defining feature of modularity. Evolutionary psychologists, on the other hand, place greater emphasis on domain specificity in their definitions of modules in their massive modularity theories (e.g. Sperber 1994; Cosmides and Tooby 1994; Pinker 1997; Carruthers 2006a).

This thesis will, therefore, be split into two halves. After providing an introduction to modularity in general in Chapter 2 and drawing differences between modest modularity and massive modularity in Chapter 3, I will explore whether phonology can be considered a modest or massive module of the mind. In Chapter 4 I will consider the extent to which phonology can be considered informationally encapsulated and therefore a modest module and in Chapter 5 the extent to which it can be considered domain specific and therefore a massive one.

I do this, because there seems to be even less of a consensus among phonological practitioners at present, than there was forty or fifty years ago, and – whilst it would be naive to think that there ever will be *full* agreement as to the modularity or nonmodularity of phonology – I would hope that my work (which explores two different definitions of what it means to be modular) will contribute to the discussion in at least some small way. I do hope, however, that we do one day find an answer to this question – on the one hand, to inform linguistic theory, and on the other to better inform the field of aphasiology so that treatments of phonological anomia (which is an impaired ability to access words in and retrieve them from the mind that's due to a deficit at the phonological level of language processing) can be improved.

Returning to my previous point, there are a number of reasons for the lack of consensus concerning modularity among phonologists Hannahs and Bosch (2018) argue, which I will give an overview of very briefly here in order for one to see where my research sits within the wider research context. The first (and perhaps most obvious one) is Optimality Theory (or OT) (Prince and Smolensky 1993), which caused controversy among generativists (Hannahs and Bosch 2018). The framework on which it was built called into question some of the claims made within the generative phonology field. These included (but of course weren't limited to) one's reliance on rules to mediate between underlying and surface forms and the theory that the two (abstract underlying representations and concrete surface representations, that is) are opaquely related (Hannahs and Bosch 2018).

The development of OT, however, is of course not the *only* contributing factor. Again during the 1980s/early 1990s (which was around the time as OT's advent too), phonologists began to question some of the assumptions of abstractness underlying the (then-current) approaches to generative grammar. Within OT, e.g., there was an endeavour to ensure that constraints were grounded in phonetics (e.g. Pierrehumbert 2000; Archangeli and Pulleyblank 1994; Hayes et al. 2004). Outside of OT, there was a trying to tie phonology and phonetics together (as was the case with articulatory phonologists (e.g. Browman and Goldstein 1986), usage-based phonologists (e.g. Bybee 2003) and exemplar theorists (e.g. Bybee 2006; Johnson 2007) (Hannahs and Bosch 2018).

Two other recent developments are likely to have led to this fragmentation within the field, Hannahs and Bosch (2018) argue, however – sociophonetics and database studies. Sociophonetics, they say, having grown out of the Labovian sociolinguistic paradigm (in which sociolinguistic variation – such as the alternation of the allophones [n] and [ŋ] in the pronunciations of words like *singing* with and without g-dropping, respectively – is examined from the perspective of socio-economic contexts) generally has a heavier focus on phonetics than it does on phonology. For most sociophoneticians, the phonological system itself is of little interest, they point out, though this is, of course, a bit of a sweeping statement. Indeed there are (albeit a small number of) phonologists interested in interpreting sociophonetic variation in the light of phonological theory (Fruehwald 2013, 2016). This is something I touch on in Chapter 4.

The other recent development, database studies, has led to an interest in 'big' data and, for some practitioners, an assumption that the absence of big data to support a theoretical position invalidates it entirely (Hannahs and Bosch 2018). Whilst this seems to be a stance taken by empirical scientists in general, I should point out that the problem with this line of thinking in phonology in particular is that, from a competence/performance perspective, surface forms cannot always account for what is going on underlyingly.

There is more to be understood in phonology than what can be *observed* and so - if we don't engage with the systems underlying those observations and ask questions based on what we've found (no matter how big or small our datasets are) - we run the risk of missing important pieces of the puzzle, Hannahs and Bosch (2018) point out. For me, both empirical concreteness and theoretical abstraction have an important part to play in phonology, and so I draw on (but, note: don't rely on) empirical data from language change in Chapter 4, and speech and language pathology in Chapter 5, in order to inform my theoretical thinking. As Hannahs and Bosch (2018) put it:

[w]hile empirical concreteness has contributed enormously to our understanding of what is possible in human language, there is value also in focused exploration into more abstract elements of language: elements that we cannot see or measure, such as phonological structures, non-surface-true generalizations, and relationships that can only be inferred through theoretical analysis

To circle back to my previous point, I make my main contributions to knowledge in Chapters 4 and 5. In Chapter 4, I conclude that phonology can be considered substance-free and therefore can be considered an informationally encapsulated module of the mind according to Fodor's (1983, 2000) definition of (modest) modularity. In Chapter 5, on the other hand, I present evidence from the field of aphasiology which is suggestive that phonology is not domain specific (and conclude that it is, therefore, *amodular* to the minds of massive

modularists (like Carruthers (2006)) who demand domain specificity in their definitions of modularity).

Also in Chapter 5 (and 6), I show how my results are relevant to a broader range of questions than those about the modularity of phonology. Firstly, I argue that the models of the language faculty in the clinical literature are limited in that they neither offer a clear theory of the lexicon and lexical processing, nor do they account for the communication between the semantic and phonological submodules they posit in their proposals. Secondly, I argue that Jackendoff's (2002) parallel architecture model of the language faculty better does this, for Jackendoff (2002) there aren't just two component parts to the grammar (semantics and phonology) there are three (semantics, syntax and phonology) which are all, what he calls, integrative modules, and that they each communicate with one another via interface modules, which transform the output of one module it interfaces into an input interpretable by the other it interfaces and vice versa. I also argue that Jackendoff's (2002) theory of the phoneticsphonology interface (or interfaces) is superior to the one owing to Fodor (1983, 2000) and colleagues because theirs, being innate, cannot account for cross-linguistic variation. Jackendoff's (2002) theory, which does not assume innateness, can account for this better, I stress. It also provides an answer to the question of where, in the language faculty, is damaged when disorders of phonetic encoding/phonemic assembly arise - that answer being, in the phonology-phonetics interface.

Towards the end of the thesis I point out, however, that the way in which Jackendoff (2002) models the language faculty requires the phonological processor to be both informationally encapsulated (as Fodor (1983, 2000) does) and domain specific (as massive modularists like Carruthers (2006a) do) and so the data I collect and analyse in Chapter 5 casts doubt on not only phonology's domain specificity, but also Jackendoff's (2002) model's representativeness of phonology. To finish, I endorse a roughly Fodorian view that phonology is modular in that it is informationally encapsulated but not domain specific, albeit tentatively. Further research should be conducted with large sample sizes I stress, to confirm or deny whether this is indeed the case.

I think it is safe to say that this thesis is an ambitious one, for it considers a question to which many a body of research is relevant. It necessarily crosses the disciplines of philosophy of mind and language, linguistics, cognitive psychology, evolutionary psychology, language science and – to some degree – natural science and computer science, because, as Caruthers

(2006a) points out, we all seem to be asking the same questions about modularity but taking different routes in order to answer them, which is just journeying us further and further away from each other (and most likely the truth). I firmly believe that the only way we'll reach a clearer conclusion about the modularity of phonology than we have right now is to bring together different domains of inquiry, and so I hope that the conclusions I come to and questions I ask at the end of this cross-disciplinary thesis provide a useful starting point for the research of others to bring about the knowledge transfer between all fields that is so desperately needed for us to move forward. This research is unique in a number of ways, but particularly so in how broad a scope it has.

To begin, I describe the difference between the natural modularity we see in biology and the artificial modularity we see in technology, and distinguish between types of biological modules (detailing the difference between modules of mind and brain and other biological modules seen in our species and others, the difference between cognitive modules and neural modules, representative and computational modules and Darwinian and non-Darwinian modules) to demonstrate where the Fodorian and Carrutherian modules discussed in this thesis sit within the wider research context.

I provide a surface overview of the ways in which they are similar to (both computational, cognitive, natural modules of mind) and different (one is Darwinian, and the other is not) from one another, and the ways in which they are different from other modules that are e.g., representational rather than computational like them, neural rather than cognitive like them, biological, but not having to do with the mind or brain like them, and technological instead of natural like them, so that we have a rough picture of what Fodorian and Carrutherian modules are and aren't before I dive into a more in-depth discussion of Fodor's and Carruthers' understandings of modularity. Now with that being said, let us turn to Chapter 2.

#### **Chapter 2: An Introduction to Modularity**

#### 1. Introduction

Describing the world in terms of the modular organisation of its parts dates as far back as 1543 when Copernicus' *De Revolutionibus Orbium Coelestium (On the Revolutions of the Heavenly Spheres)* marked the beginning of the scientific revolution (Callebaut 2005). It inspired Leibniz's and Kant's work on faculty psychology in the 17<sup>th</sup> and 18<sup>th</sup> centuries as well as Gall's work on phrenology in the 18<sup>th</sup> and 19<sup>th</sup>, and even now, in 2019, modularity is a driving force of much scientific research. In biology, both structural and functional modules are recognised, and engineers, having seen the value of modularity as it exists naturally in biological systems, have come to make use of modular design principles themselves in the technological domain (Wilhem 1997). In the cognitive sciences, modules as systems of representation have been proposed by the likes of Chomsky (1965, 1988), and in 1983 Fodor launched a debate on whether the mind is made up of functional modules – one that has, of course, continued to the present day. Neuroscientists have asked similar questions to the ones that cognitive scientists have asked about the mind about the brain, but I will have no more to say about those here, for that goes beyond the scope of this thesis.

This chapter provides some conceptual foundations for the enterprise that follows along the lines of Callebaut (2005). I first describe the various meanings and uses of the word 'modularity' (for example as *explanans* in the case of natural modules or *explanandum* in the case of artificial modules). I then provide as general a definition as possible, that applies to the wide range of contexts in which it appears in section 2. I will then describe the differences between the fields of computational cognitive science and neural cognitive science and their modules in section 3.1 before looking in some detail at the differences between cognitive and neural modules in section 3.2. Lastly, I will outline the two types of cognitive modules that have been explored in the cognitive science literature: (non-Darwinian) computational modules (in section 3.3.1.1) and Darwinian computational modules (in section 3.3.1.2). These will be the focus of the thesis, and so the chapter after this one (i.e., 3) will be devoted to their discussion.

#### 2. Natural modules vs. artificial modules

In our world, modular systems – whether they be natural ones (as is the case in biology and the cognitive and neurosciences) or artificial ones (as is the case in technology) – abound

(Simon 1969). They have been discovered by a number of researchers working in a number of different research domains, but despite the differences in these researchers' definitions of modularity (there are many), they all seem to be in agreement that modules are 'unit[s] that [are] component part[s] of [...] larger system[s] and yet possessed of [their] own structural and/or functional identit[ies]' (Moss 2001: 91).

Consider some examples from biology and technology. In biological organisms, cells come together with cells of similar structures to perform shared functions as tissues; tissues come together to create organs and organs, organ systems (Ritter 1998). The heart, for example, is made up of cardiac tissue (which is itself made up of cardiac cells) and functions as part of the cardiovascular system to pump blood around the body (Farley et al. 2012). The cardiovascular system works together with two other systems (the pulmonary and the systemic) as part of the circulatory one, to transport through the blood oxygen and essential nutrients *to* cells and waste products *from* them (Farley et al. 2012). The parts and systems of cars (which are of course technological artefacts), on the other hand, are susceptible to independent breakdown and repair because of their modular architectures. Damage to a car's brake pedal would affect the functionality of its brake system, for example, of which it is a part, but not its exhaust or steering systems, of which it is not. The functionalities of those, the exhaust and steering systems, that is, would remain intact.

It is important to bear two things in mind here, however. Firstly, the structural and functional modularities mentioned by Moss (2001) are two very different things, and so don't necessarily map neatly onto one another (I'll return to this point in more detail in section 3.2.1 where distinctions between neural modules (i.e., modules of structure) and cognitive modules (i.e., modules of function) will be drawn). Secondly, one must bear in mind that different research domains make use of quite different research strategies. Modules are said to be 'top-down', for example, when they are the product of a researcher breaking down a complex system in a reverse engineering fashion to gain an insight into its inner workings. This is the type of module that is typically found in biology and is especially common in cognitive science (Bechtel and Richardson 1993). 'Top-down' modules aren't empirically observed but brought forth as *explanans* (Moss 2001; Eble 2005) in straightforward Popperian (1935, 1959) fashion. Examples of the 'bottom-up' research strategy in which small, simple, subsystems are pieced together to give rise to a larger, more complex system can be found in technological arenas, on the other hand, such as engineering, as well as in neuroscientific

research (Barbutti et al. 1993; Fontana and Buss 1994; Adami 2002; Husken et al. 2002). In this context, modularity arises as an *explanandum* as opposed to as *explanans* (Bolker 2000).

Now, given that this is a thesis on cognitive modularity and not neural modularity (or technological modularity, for that matter), and since we've established that modules in the cognitive sciences are all 'top-down' – I will have no more to say about the distinction I've drawn between top-down and bottom-up research strategies here. To this thesis, what matters most is what constitutes a module in the eyes of the researchers who posited them not how their characterisations came to be, and so this will be the focus of the subsections that follow.

#### **3. Modularity in cognitive science**

Just as developmentalists and evolutionists disagree about what it means to be a module and for a system to be modular in biology, the thesis that the mind is modular in its architecture has triggered a tremendous amount of often heated debate in the cognitive sciences as well (Callebaut and Rasskin-Gutman 2005). Fodor (1983) is often cited as having had the most impact on recent theorising on these matters, for an entirely new discipline, evolutionary psychology (which rests on a massively modular conception of cognitive architecture, that is, that the mind is composed largely (or even entirely, actually) of computational modules (Callebaut and Rasskin-Gutman 2005)) was born out of cognitive scientists contesting Fodor's (1983) thesis that the mind is less modular than that (see section 3.3). One of the most obvious reasons why modularity is such a controversial topic in the cognitive sciences, however, is that there are two different (and conflicting) approaches to the study of cognitive scientific research: computational cognitive science and neural cognitive science (Calabretta and Parisi 2005), and so I'll now turn my attention to that.

#### 3.1 Computational cognitive science vs. neural cognitive science

The theoretical paradigm of computational cognitive science (or cognitivism) is based on an analogy of the mind as the software of a computer, according to which the mind is a system that manipulates symbols computationally (Newell and Simon 1976). More recently, though, a different kind of cognitive science called neural cognitive science (or as some prefer, connectionism (Rumelhart and McClelland 1986)) has arisen, that rejects the analogy of the mind as computer software, instead interpreting our cognitive capacities using theoretical models known as neural networks that are inspired by both the structure and the functioning of the nervous system (Rumelhart and McClelland 1986). For connectionists, the mind is not a computational system of symbol manipulation; rather, it is the result of the many

interactions that take place among neurons in a neural network. For them, the mind 'consists entirely of quantitative processes in which physiochemical causes produce physiochemical effects' (Calabretta and Parisi 2005: 309).

Cognitivists and connectionists also differ in their assumptions about modularism and nativism. While computational cognitive science tends to be strongly modularistic, neural cognitive science does not (Calabretta and Parisi 2005). Meanwhile, whilst cognitivists tend to be nativists, connectionists tend not to be (Calabretta and Parisi 2005), as Table 2.1 depicts:

Two types of cognitive science			
Computational	Mind as a	Modularist	Nativist
cognitive science or	computational		
cognitivism	system of symbol		
	manipulation		
Neural cognitive	Mind as the global	Anti-modularist	Anti-nativist
science or	result of the many		
connectionism	physicochemical		
	interactions that take		
	place in a network of		
	neurons		

Table 2.1 A table showing the two different approaches to studying cognitive science

But alas, although cognitivists can be differentiated from connectionists in terms of their subscriptions to modular and nonmodular, and nativist and anti-nativist, ideals, one should note that cognitivists are divided on some things as well, in particular on their views of the adaptive nature of inherited traits. Evolutionary psychologists, for example, believe that cognitive modules are one of the many biological modules, and so think of the modular structure of the mind as being the result of evolutionary pressures from the environment as the modular structures of biological systems are. They, therefore, embrace a strong form of adaptationism as well as nativism (Calabretta and Parisi 2005); not only do they believe that modules of the mind are a part of a human's genetic material, they believe that they arose as a consequence of evolution (Calabretta and Parisi 2005).

Not all cognitivists subscribe to this idea, however. For example, while Noam Chomsky sees the mind as a modular system that has a specific subsystem that is specialised for language (Chomsky 1965, 1988), he does not believe that language in humans emerged under any specific evolutionary pressure (Fodor 2000). Nor does Fodor (1998), actually, who is in fact in favour of a strong form of nonadaptive modularism (Fodor 1998). I will leave the debate between proponents of adaptive modularism and nonadaptive modularism here for now but will return to it when I come to describe the differences between the cognitive modules in more detail in section 3.3.

#### 3.2 Cognitive modules vs. neural modules

Although a cognitive scientist (a.k.a. Fodor 1983) popularised the term 'module', the concept is not a new one. It is not uncommon for the complex systems of the mind *or* the brain to be analysed in terms of collections of less complex subsystems. In the neurosciences, there has been a long-standing interest in decomposing into component structures called modules as well (Meyering 1994).

I should probably point out that when I speak of neural modules in this section, I don't mean the modules of the artificial neural networks modelled by connectionists that are described in section 3.1, since, as I said there, they tend to be nonmodular in their architecture. What I mean by neural modules in this context are the natural modules of brains themselves rather than the modules of artificial models of them. In this section, then, I draw a distinction between the (more functional) modules of the mind that are proposed by cognitivists and the (more structural) modules of the brain that are proposed by neuroscientists.

#### 3.2.1 Function vs. structure

By the end of the nineteenth century neurologists Wernicke, Sherrington and Cajal had introduced a new and empirically based theory of the brain according to which its functioning took place in a cellular system of neurons (the brain's operative units (Meyering 1994)) which were arranged into more or less functional groups (Schnelle 2010). This was the most influential view of the brain's architecture until the turn of the twentieth century, which brought with it enough empirical and conceptual progress to lead Hebb (1949) to propose a neural sub-structure like the one in Figure 2.1:



Figure 2.1 Hebb's (1949) neural sub-structure schema (adapted from Schnelle 2010: 9)

According to Hebb (1949), cellular organisations operate in very different areas of the cortex, which is itself divided into four anatomically distinct areas: the frontal lobe, the parietal lobe, the occipital lobe and the temporal lobe, which each contribute to different sub-aspects of more globally organised functions such as perception, memory and motor control (Kandel et al. 1995). Hebb (1949) individuates the modules of the brain as anatomical entities that come together in their operations. This is supported by research in more recent years (e.g. Lettvin et al. 1959; Powell and Mountcastle 1959; Kilmer et al. 1969; Hubel and Wiesel 1974) and so in general, it seems fair to say that the concept of a module in the neurosciences is one of *structure* rather than of function (Arbib 1987).

By contrast, the models of mind modularity that are proposed by cognitivists are characterised in a fundamentally *functional* way. Particularly prevalent in the cognitive science literature is the one developed by Fodor (1983). He began with a cognitive analysis of global psychological functions (such as linguistic and perceptual behaviour) and developed a functional taxonomy of them based on the characteristic operational differences between the systems involved in their functioning (Meyering 1994).

For cognitivists, modules are *postulated* rather than *observed* entities, then (Calabretta and Parisi 2005). They are components of theories of mind, which hypothesise that the mind can be broken down into a number of functionally independent components that work together to explain some phenomenon of interest (Calabretta and Parisi 2005). This was evidenced in formal linguistics of the Chomskyan variety, which observed the linguistic judgements of native speakers to interpret that there are three autonomous modules that work together in linguistic competence: syntax, semantics and phonology. This purely theoretical notion of a module is defined and defended by Fodor (1983) in *The Modularity of Mind*. Evolutionary psychologists, who tend to disagree with Fodor (1983) about the extent to which the mind is modular and as a consequence, his definition of a module (a point I will come back to in section 3.3), have a cognitivist orientation too. They conceive of the mind as a collection of specialised and genetically inherited adaptive modules (which are, again, like Fodorian modules, in that they have a cognitivist orientation in that they are functionally independent (Calabretta and Parisi 2005)).

#### 3.2.2 Closing the gap between functional and structural modules

There is a tendency for humans to fill in the unspecified details of ontologically ambiguous objects (Boyer 2001; Bloom 2004). This is particularly apparent in our responses to theories that postulate information structures but leave all of the parameters other than those of the theory itself unstipulated (as in the spirit of Chomsky (1965), for example, who theorised that the grammar is made up of a syntactic, a semantic, and a phonological component only) (Barrett 2006). It is important to recognise, however, that underspecification must be taken seriously if theories like Chomsky's are to have any real value (Barrett 2006). It is a mistake to criticise a theory about psychological entities by filling in parameters that weren't meant to be filled in in the first place and then question the plausibility of the postulated structure based on what one has filled in (Barrett 2006). While it might well be reasonable to ask of Chomsky where in the brain syntax, semantics and phonology are located, that question is in many ways irrelevant to whether his hypothesis is correct or not because it was framed in terms of information, not in terms of brain structure. As Marr (1982) rightly pointed out, while there is a relationship between hypotheses about information structure and hypotheses about brain structure, that relationship is an asymmetric one, and there are many ways in which an information structure might be instantiated.

Although, then, one can look for correspondences between the two types of modules (i.e. the theorised modules of cognitive science and the observed modules of the brain), one cannot assume that there will be any. The brain's organisation into structural modules by neuroscientists mightn't marry up with the cognitive scientists' compartmentalisation of the mind into functional modules, but that does not necessarily mean that either of the theories must be wrong as they are each built on entirely different foundations.

#### 3.3 Cognitive modules: Some differences

Ever since the cognitive revolution in psychology began some 60 or so years ago with Chomsky's (1959) review of Skinner's (1957) *Verbal Behavior*, the evidence in favour of cognition having a modular architecture – which is, by way of explanation, to say 'that cognition is subserved by a number of innately channelled [...] systems whose operations are largely independent of, and inaccessible to, the rest of the mind' (Carruthers and Chamberlain 2000) – has been mounting up. Initially the evidence only supported Fodor's (1983) conception of cognition being modular around the edges (i.e. the idea that only the input and output systems of the mind, that is, those responsible for the five senses of opthalmoception, audioception, gustaception, olfaception and tactioception and the system responsible for our linguistic competence) are modular (Carruthers and Chamberlain 2000). More recently, however, there has been evidence (from e.g. Atran (1990), Baron-Cohen (1995) and Sperber et al. (1995)) to suggest that the central systems (i.e. those charged with the generation of beliefs from perception or indeed, other beliefs) are modular in structure as well (Carruthers and Chamberlain 2000).

Recall also from section 3.1 that while cognitive modularism is typically associated with nativism, not every cognitive modularist has taken an evolutionary perspective (Carruthers and Chamberlain 2000). Chomsky (1988) and Fodor (1998), for example, have been inclined to ignore all evolutionary theorising, thinking of modules as mere by-products of the expansion of the hominid neocortex instead (Carruthers and Chamberlain 2000). Evolutionary psychologists, on the other hand, share a different position, that evolution by natural selection is the only non-cultural explanation we have for the development of organised, functional complexity (as Pinker and Bloom (1990) and Pinker (1994) so decisively point out) (Carruthers and Chamberlain 2000).

In order to make sense of the various interpretations of modularity in cognitive science, Gobet (2015) helpfully distinguished two meanings of modularity: the *functional* (for example the

modules proposed by Fodor (1983) that will be covered in section 3.3.1.1) and evolutionary psychologists (which will be the focus of section 3.3.1.2)) and the *knowledge meaning* (read 'representational') which has some kinship with the notion of modularity used by Chomsky (1965, 1988), but not his later work.

This is not dissimilar to Samuels' (2000) earlier classification of Fodor's (1983) and evolutionary psychologists' modules as computational modules, but while Fodorian and Carrutherian modules bear similarities to one another in terms of their functionality, they differ in that the modules of evolutionary psychology are *Darwinian* (take an evolutionary perspective) whereas Fodorian (1983) modules are not. These distinctions are especially relevant to those who wish to assess the evidence for the existence of computational modules in cognition as I do, and so the following sections will serve to dichotomise them further.

#### 3.3.1 Computational modules

#### 3.3.1.1 Non-Darwinian modules

Despite Chomsky's (1965, 1988) work on modularity in linguistics, it was Fodor (1983)'s seminal *Modularity of Mind* that dominated the theoretical discussions in philosophy and psychology, and provided the modern origins of the more general modular models of cognition we know today (Lyons 2001; Carruthers et al. 2005). Fodor (1983) argued that there is a trichotomous architecture to cognition in that it's composed of: *transducers* which serve to convert stimuli into signals to be used in processing, *central systems* which are responsible for the likes of general inference, reasoning, and the generation of beliefs and desires and *input systems* which act as intermediaries between the sensory transducers' signals and converting them into hypotheses about the external world in a format that the central system is able to operate on (Meyering 1994). The primary function of input systems Fodor points out is 'to so represent the world as to make it accessible to thought' (1983: 40).

According to Fodor (1983), there is a world of difference between the rational computations of an open cognitive system (i.e. central cognition) and the unconscious, automatic computations of a closed perceptual one (i.e. the part of cognition that is made up of input systems). The cognitive abilities of input systems, Fodor (1983) stresses, differ from those of the central system in two ways. Firstly, input systems are domain specific in that they can only generate hypotheses with respect to a very limited class of distal properties (Meyering 1994). Secondly, modules are informationally encapsulated in that their internal processes are impervious to influence from the rest of cognition, or, to put it more simply than that, that information outside the module is inaccessible from within it (Callebaut 2005; Carruthers et al. 2005).

By contrast, he says, central systems are characterised by properties that are diametrically opposed to domain specificity and informational encapsulation. For Fodor (1983), central systems are Quinean (in that the hypotheses they develop such as scientific theories, for example, are evaluated in light of everything else that a person happens to believe, and therefore their confirmations are global, as opposed to local, phenomena) and isotropic (in that in forming those sorts of scientific theories, everything a person knows may in principle be relevant for the development and confirmation of those ideas) (Meyering 1994).

In light of these characterisations, Fodor (1983) went on to argue that the input systems (which are situated around the mind's periphery) are modular while central processors are not, for according to Fodor (1983), modules must be nine things:

- (1) fast in their processing;
- (2) mandatory in their operations;
- (3) computationally shallow;
- (4) inaccessible to the rest of cognition;
- (5) associated with fixed neural architecture and as a result
- (6) exhibit characteristic and specific breakdown patterns;
- (7) have ontogenies that seem to be endogenously determined and follow a characteristic pace and sequencing, as well as being
- (8) domain specific and
- (9) informationally encapsulated.

There is indeed overwhelming evidence for each of those features in the sensory and linguistic systems (which will be described in due course in Chapter 3), but Fodor has since softened his requirements for modularity somewhat, and now conceives of modules as 'computational system[s] with [...] proprietary database[s which] operate[...] to map [their] characteristic inputs onto [...] characteristic outputs [and] in the course of doing so, [have] informational resources [that] are restricted to what is in [their] proprietary database' (Fodor 2000: 63). For Fodor (2000), then, the *sine qua non* of modularity is informational encapsulation: cognitive processes are modular precisely because they exhibit encapsulation,

and central cognition (which does not), is resolutely ammodular as a result (Carruthers et al. 2005).

Other researchers (e.g. Tooby and Cosmides 1992; Pinker 1997; Scholl and Leslie 1999; Carruthers 2003a, b; 2006a) have increasingly argued otherwise, that the mind is more modular than the Fodorian perspective of modularity allows. In doing so, they have been required to adjust Fodor's (1983) definition of a module somewhat, though; I will elaborate on this in section 3.2.1.2 of this chapter, and even more so in Chapter 3.

Before drawing this section to a close, I would like to point out that the computational modules described above are a very different sort of structure than knowledge meaning (or representational) modules. Whilst knowledge meaning modules are systems of representations and so are, in a sense, inert in that they won't eventuate in behaviour until cognitive mechanisms manipulate their representations, computational modules are the processing devices that do that very thing (Samuels 2000). Computational modules are typically assumed in the literature to be symbol (or representation, it makes more sense to say) manipulating devices that take representations held in knowledge meaning systems as inputs and manipulate them according to formally specifiable rules in order to generate output representations (Pylyshyn 1984; Segal 1996). They are, therefore, importantly different with regards to the functional roles they play in our cognitive economy, as well as with respect to their features.

#### 3.3.1.2 Darwinian modules

Whereas Fodor's (1983, 1998) view of the mind like Chomsky's (1965, 1988) was and is anti-Darwinian in that it is opposed to the idea that modules are naturally selected for (Carruthers and Chamberlain 2000), evolutionary psychologists (e.g. Carroll 1988; Garfield 1991; Barkow et al. 1992; Hirschfield and Gelman 1994; Sperber 1994, 2002; Charland 1995; Segal 1996) and their philosophical associate Pinker (1997) tend to subscribe to the very much Darwinian-like adaptationist view of the Modern Synthesis that's due to Williams (1966) and Dawkins (1976). This views variety between organisms as being due, in part, to their genetic material adapting as a response to their ever-changing environments as they fight for survival in them (Callebaut 2005).

This adaptationist view of the mind is one of the four central tenets of evolutionary psychology (the others being *computationalism, nativism* and *massive modularity*) (Samuels

1998). The first, computationalism, is the view of the mind as an information processing device that can be likened to 'a computer made out of organic components rather than silicon chips' (Cosmides et al. 1992: 7). Thus, Darwinian modules are not Chomskyan modules but rather a kind of computational module. The second claim of evolutionary psychology is that much of the human mind's structure is innate, which rejects the familiar empiricist proposal that the human mind consists of little more than general-purpose learning mechanisms (the view that has been dominant in psychology for most of the twentieth century) in favour of the nativist stance that is associated with Chomsky and his followers (Samuels 2000). The third fundamental hypothesis that evolutionary psychologists endorse is that the mind is massively modular. This is the view that the human mind is largely (or perhaps even entirely) made up of Darwinian modules that comprise both the peripheral systems (which Fodor (1983) argued are modular) and the central capacities (which Fodor (1983) argued are not) (Cosmides and Tooby 1992; Gigerenzer 1994; Leslie 1994; Pinker 1994; Sperber 1994, 2002; Baron-Cohen 1995; Samuels 1998, 2000).

The final tenet of evolutionary psychology, adaptationism, meanwhile, is roughly the view that cognitive modularity is the product of evolution (Samuels 2000). Its advocates suggest that human minds were designed by natural selection in order to solve adaptive problems (e.g. 'evolutionary recurrent problem[s] whose solution promoted reproduction, however long or indirect the chain by which it did so' (Cosmides and Tooby 1994: 87)) and so for them the modules that make up our minds are simply adaptations that were 'invented by natural selection during the species' evolutionary history to product adaptive ends in the species' natural environment' (Tooby and Cosmides 1995: xiii). Its advocates wonder why it is uncontroversial that the nonpsychological modules of organisms (for example those in the eyes or the liver) are generally best understood as adaptations (Sperber 2002), but it is controversial to argue that psychological modules may have arisen through evolution as well (Callebaut 2005). I too have asked myself this question.

#### 4. Concluding remarks

The aim of this introductory chapter was not to spell out what modularity in the cognitive sciences is in greater detail than is necessary for a proper understanding of what follows, nor was it to survey the various lines of criticism that have been addressed to any of the notions of modularity discussed here. Rather, this chapter was written to provide the reader with a bit of background about modularity and how different researchers in different research domains define it in order to situate my own work within the wider research context. I'd also like to

point out that what was discussed in section 3.3 by no means exhausts the ways in which the term 'module' is used in contemporary cognitive science – for a more comprehensive review, one should see Segal (1996) - but what was necessary for this thesis on computational modularity was to draw a distinction between Chomskyan modules (which are said to be systems of mental representations and so will not be paid any more attention to from here on out) and mechanisms that are computer-like in character and are so-called computational modules (which will, of course, be the focus of it). Computational modules can be thought of as being Darwinian (i.e. those that are adaptive and proposed by evolutionary psychologists) or non-Darwinian (those that are proposed by Fodor (1983) and are not). Since this thesis aims to assess the evidence for a) whether phonology can be considered a computational module of the mind in the Fodorian (1983) sense of the word and b) whether it can be conceived as a computational module according to Carruthers (an evolutionary psychologist)'s (2006a) definition of a modularity, Chapter 3 will be devoted to discussing what it means to be a module according to Fodor (1983) and Carruthers (2006a) more thoroughly. The following figure demonstrates the ways in which all of the modules reviewed are related. Of course, the ones I am concerned with in the thesis are the bottom two: non-Darwinian (or Fodorian modules) and Darwinian modules (i.e. those proposed by evolutionary psychologists):



Figure 2.2 The different types of modules and their relationships with one another

#### **Chapter 3: Modest Modularity vs. Massive Modularity**

#### 1. Introduction

Recall from Chapter 2 that the modularity hypotheses of today derived, first and foremost, from Chomsky's (1965) work on generative grammar. In *Aspects of the Theory of Syntax* he made two central claims about the architectural structure of the language faculty: (a) that it is a module of the mind distinct from ones responsible for, say, music and mathematics and (b) that it can itself be divided into a number of submodules relating to language meanings, structures and sounds (the semantic, syntactic and phonological modules, respectively) (McGilvray 2005). The thesis was revived by Fodor in 1983 with his publication of the *Modularity of Mind,* in which it was extended to all of the low-level cognitive processes subserved by the input and output systems around the mind's edges. This has launched a debate in the cognitive sciences is ongoing still (Barrett and Kurzban 2006).

Now, whilst Fodor (1983, 2000) made the case for a sort of minimal (or indeed modest) lowlevel peripheral-systems modularity of mind theory according to which one's mental structure can be divided into modular information processing modules in the input and output systems that reside around the mind's periphery (for example for vision, audition, face-recognition, various motor-control systems and, most relevantly to this thesis, language-processing, and an amodular modular faculty of central cognition (in which concepts are deployed, beliefs formed, inferences drawn and decisions made (Browne 1996)), evolutionary psychology researchers (e.g. Symons 1987; Cosmides and Tooby 1992, 1994; Tooby and Cosmides 1992; Sperber 1994, 1996; Pinker 1997) have argued that a broader notion of modularity is possible. For these researchers, the central systems are modular as well and aren't as different from the peripheral systems as Fodor (1983, 2000) thought (Symons 1987; Tooby and Cosmides 1992; Cosmides and Tooby 1994; Sperber 1994; Pinker 1997).

Many attempts have been made to disprove this proposal (formally known as the massive modularity thesis (Sperber 1994; Samuels 1998; Carruthers 2005)) (e.g. Elman et al. 1996; Deacon 1998; Ramachandran and Blakeslee 1998; Buller and Hardcastle 2000; Fodor 2000; Panksepp and Panksepp 2000, 2001; Quartz and Sejnowski 2002; Buller 2005), and many researchers position their theses somewhere in between those two poles (e.g. Carey 1985; Carey and Spelke 1994; Spelke 1994; Smith and Tsimpli 1996; Hauser and Carey 1998;
Hermer-Vazquez et al. 1999; Cosmides and Tooby 2001). These in-between theses go beyond the scope of this one, though. Thus, this is all I will say about them.

The first section of this chapter provides an overview of Fodor's (1983) modestly modular view and explicates the nine properties he attributes to peripheral modular systems in section 2.1.1 before describing the differences he draws between peripheral and central cognitive systems and explaining why central systems are decidedly nonmodular to his mind in 2.2. In section 3, meanwhile, I map out Carruthers' (2006a) three main arguments for massive modularity (the argument from design in section 3.2.1, from animals in section 3.2.2 and from computational tractability in section 3.2.3), before describing the definition of a 'module' that would follow should those arguments have grounds. The result is a thesis of modularity that is some distance from Fodor's (1983, 2000) (in particular in its definition of modularity which refers to a more restrictive list of features (Prinz 2006) and in its claim that cognitive systems needn't be informationally encapsulated to be modular as Fodor (2000) so vehemently suggests (Barrett and Kurzban 2006)). I explore this in detail in section 3.3, before making some concluding remarks in section 4.

#### 2. Modest modularity

#### 2.1 Fodorian modules

Recall also from Chapter 2 that Fodor (1983) characterises modules by appeal to nine properties as domain specific, innately specified processing systems that deliver 'shallow' (non-conceptual) outputs (Marr 1983), are mandatory in their operations, swift in their processing and isolated from and inaccessible to the rest of cognition, are associated with specific neural structures and so are liable to characteristic patterns of breakdown and develop according to a paced and distinctively arranged sequence of growth (Carruthers 2006a).

Each of these features call for clarification before I describe why the mind is modular around its edges and but not at its centre for Fodor (1983), and so I will comment briefly on the various elements of this account (i.e. on their (i) characteristic breakdown patterns, (ii) fixed association with fixed neural architecture, (iii) mandatory way of operating, (iv) fast processing, (v) 'shallow' outputs, (vi) characteristic pace and sequencing, (vii) domain specificity, (viii) informationally encapsulation and (ix) limited central access) here. To make the exposition as streamlined as possible, the features will be thematically clustered and examined on a cluster by cluster basis along the lines of what Prinz (2006) did in *Is the Mind Really Modular*?. Dissociability and localisability will be explored first in section 2.1.1.1,

mandatoriness, speed and superficiality in 2.1.1.2, ontogenetic determinism in 2.1.1.3, domain specificity in 2.1.1.4 and inaccessibility and encapsulation in section 2.1.1.5.

## 2.1.1 Features of Fodorian modules

## 2.1.1.1 Dissociability and localisability

A functionally dissociable system is one that's operations can be selectively impaired (i.e. damaged independently of the operations of other systems (Robbins 2015)). These selective deficits are often observed as a consequence of focal brain injury and are said to strongly evidence the claim that mental faculties are localised in biological tissue (Prinz 2006). Agnosia (a cognitive disturbance caused by neurological damage affecting a single, sensory modality) is a prime example of dissociability and localisability at work. Its first description was provided by Lissauer in 1890 who made the claim that focal brain lesions can impair either visual or auditory perception leaving the other sensory modalities intact (Lissauer 1890). This view is one that is shared by Mesulam (2000) too, who claims that the visual modality regions adjacent to V1 in the brain are responsible for the processing of the colour, form and motion of visual stimuli while the primary auditory cortex in the left hemisphere of the brain (A1) is responsible for auditory processing and that damage to each of these areas can result in selective impairment.

See Figure 3.1 for an illustration of the brain's functional zones (where 'AA' represents the auditory association cortex, 'ag' the angular gyrus, 'A1' the primary audition cortex, 'B' Broca's area, 'cg' the cingulate cortex, 'f' the fusiform gyrus, 'FEF' the frontal eye fields, 'ins' the insula, 'ipl' the inferior parietal lobule, 'it' the inferior temporal gyrus, 'MA' the motor association cortex, 'mpo' the medical parietooccopital area, 'mt' the middle temporal gyrus, 'M1' the primary motor area, 'of' the orbitofrontal region, 'pc' the prefrontal cortex, 'rs' the retrospenial area, 'SA' the somatosensory association cortex, 'gy the superior parietal lobule, 'st' the superior temporal gyrus, 'S1' the primary somatosensory area, 'tp' the temporopolar cortex, 'VA' the visual association cortex, 'V1' the primary visual cortex and 'W' Wernicke's area):



Figure 3.1 Functional zones in the brain (adapted from Daroff et al. 2012: 95)

Adjacent and anterior to the primary visual cortex in the posterior occipital lobe (V1) is the visual association cortex (VA). This region contains several areas including V4, which is specialised for colour recognition and can cause contralateral hemiachromatopsia (a loss of colour perception (or colour blindness)) when damaged (Mesulam 2000). Bilateral damage to the primary auditory cortex (A1) (which is the part of the cerebral cortex that's responsible for processing auditory information and plays a pivotal role in the auditory system (Pickles 2012)) can cause a type of hearing loss called central deafness, meanwhile (Mesulam 2000).

Clinical linguistic research has brought us equivalent cases of selective impairments from focal injury such as agrammatism (the loss of complex syntax), dyslexia (disordered reading and writing ability) and (something that's especially relevant to Chapter 5) anomia (a deficit

in word retrieval). These three language disorders have been found in individuals that are otherwise cognitively normal (Damasio et al. 1996), which would suggest that the capacities lost in them are subserved by functionally dissociable mechanisms. Clinical linguistics has also brought us examples of localisable systems – those responsible for auditory production and comprehension most markedly. Broca's aphasia, for example, which affects auditory production but not auditory comprehension (Darley et al. 1975; Marshall et al. 1990; Friedmann 2006; Fogle 2008) is said to be caused by damage to Broca's area (see 'B' in Figure 3.1), the posterior-inferior frontal gyrus of the left cerebral hemisphere or the surrounding vicinity (Naeser and Hayward 1978; Kertesz et al. 1979; Damasio 1989; Kearns 2004; Keller et al. 2009). Wernicke's aphasia, on the other hand (which causes problems in the opposite direction in that it affects auditory comprehension but not auditory production (Albert et al. 1981)) is said to result from damage to Wernicke's area (see Figure 3.1's 'W'), the posterior-superior temporal gyrus of the left cerebral hemisphere (Naeser and Hayward 1978; Fitzgerald 1996; Kolb and Whishaw 2008). This is, of course, suggestive that the mental faculty responsible for language production lies in Broca's region and the one responsible for language comprehension in Wernicke's region, and that damage to those different domains can result in quite different deficits (although one should recall from Chapter 2 that functional and structural modules don't *always* map onto one another as neatly as this).

### 2.1.1.2 Mandatoriness, speed and superficiality

Something else that merits an explanation here is the mandatory quality that's assigned to Fodor's (1983) modules. At its core, mandatory operation has to do with whether a cognitive system is in any way controlled by consciousness. If a mechanism is uncontrolled by consciousness, which is to say that it is capable of both starting and stopping its operating without any mindful effort to do so, then it is what we call mandatory in its processing (Bargh and Chartrand 1999). Fodor's (1983) central point about mandatoriness, essentially, is that modules automatically and obligatorily process information; that the processing operations observable in input and output systems 'are mediated by automatic processes which are obligatorily applied' (Marslen-Wilson and Tyler 1981: 327). To illustrate, he gives an example from spoken word recognition pointing out that 'you can't help hearing an utterance of a sentence as an utterance of a sentence' (Fodor 1983: 52) and that 'you can't hear speech as noise even if you would prefer to' (Fodor 1983: 53). This, he says, is because the linguistic system's computations take place independently of will; they involuntarily respond to

relevant stimuli and so (can be, and indeed have been) likened to reflexes, specified to apply whenever they can (Schwartz 1986).

Further evidence for the mandatoriness of input systems has been found in Stroop-type tasks (Stroop 1935) that require their subjects to name the colours in which written words are presented to them. When the written words are themselves the names of colours, response latencies and accuracies are affected by whether they are the same as or different from the colours in which they presented. Latencies are shorter, and accuracies are higher when these two things are one and the same, for example, participants more quickly and accurately identify a font colour as being pink when it used to scribe the word *pink* than the word *blue* and vice versa. This suggests that visual-word recognition accesses word meaning mandatorily (Coltheart et al. 1999). The same is true for visual-word recognition's access to phonologically unrelated to the name of its font's colour than when it is (e.g. if the colour of a target word's font is blue, participants would more quickly identify the word *clue* than its synonym *hint* as there is more of a phonological similarity (compare [b l u:] vs. [k l u:] and [b l u:] vs. [h i n t])) (Coltheart et al. 1999).

Mandatoriness and speed of processing are actually said, by Fodor (1983), to be positively correlated qualities. This is because in shadowing experiments (experiments where subjects repeat what they hear as quickly as they can) he observed that there was only a 250 msec lag on average between the stimuli the subjects were presented with and their responses to them. He related the speed of the systems' analyses to their mandatory means of operating, which he said suggested that the latter was the cause of the former. For Fodor (1983), then, if a system's operations are automatic, the natural consequence is that they must also be fast.

The third feature Fodor (1983) attributes to modular systems is that they produce, what he calls, 'shallow' outputs. The depth of an output is a function of two properties: how much computation is required to produce it (i.e. shallow means computationally cheap (Robbins 2015)) and how specific its informational content is (i.e. shallow means informationally general (Robbins 2015)) he says (Fodor 1983). These two properties are correlated too in that outputs with more specific content are typically expensive (i.e. take much effort) for a system to produce, whereas outputs with more general content can be produced inexpensively (i.e. don't take much effort). For example, it doesn't take the perceptual system much to process shallow concrete words (i.e. words that can be proven by appealing to the physical senses)

26

such as *table* or *chair*, but it does to produce abstract terms (which do not have any physical referents because they aren't tangible) or highly theoretical concepts (from theoretical quantum physics, for example), which are too semantically dense to meet the shallowness criterion (Fodor 1983).

#### 2.1.1.3 Innateness and ontogenetic determinism

Fodor (1980) also argues that modules are ontogenetically determined in that they develop predictably according to a universal maturational timeline in all healthy individuals (Cowie 1999). In a word, they're *innate*, he says. The most obvious example of an innate module of the mind for me to give here is language since extensive research into us humans' capacities for language has provided strong support for the nativist view that humans are born with a genetic predisposition to learn it. Even before the age of five, children can, without having had any formal instruction, comprehend and produce sentences that they have never heard before, and it was, in fact, these very capacities children have for language comprehension and production that led Chomsky (1965) to formulate the 'poverty of the stimulus' argument that served as the foundations for the nativist view he proposed in the 1960s.

In Chomsky's (1965, 1980) eyes, the reason that children can so easily master the complex operations of language is that they have an innate knowledge of the rules and principles that guide them in developing a grammar. In other words, language learning is facilitated by a tendency humans have towards certain structures of language. There exist for us a set of constraints including both formal universals (e.g. principles and parameters) and substantive universals (e.g. lexical categories and features) (Dabrowska 2015), that are hardwired into the brain at birth and manifest without being taught (Crain and Lillo-Martin 1999).

Just as babies naturally develop arms and not wings while they are in the womb, they learn to speak and not chirp when they are out of it, and language emerges on a fixed schedule (children reliable learn single words at around 12 months of age, come to combine them into telegraphic speech at around 18 months, acquire complex grammar at around 24 months and so on (Stromswold 1994)), Fodor (1983) argues. Please note, though, that although Fodor (1983) sees modules as being innate, he doesn't necessarily think that they evolved. I touched on this point in Chapter 2.

27

#### 2.1.1.4 Domain specificity

A cognitive processor is said to be domain specific if it is only able to compute solutions to a restricted subset of the problems that the mind has to solve as a whole (Cam 1988). In other words, domain specificity restricts the class of information the processor is able to accept as inputs (Cam 1988). As Fodor (1983: 103) puts it, 'domain specificity has to do with the range of questions for which a device provides answers' (i.e. the range of inputs for which it computes analyses). The narrower the range of inputs a system can compute, the narrow the range of questions the system can answer and the narrower the range of those questions, the more domain specific the device (Carruthers 2006a; Samuels 2006).

There are results owing to researchers at Haskins Laboratories that strongly suggest that the perceptual systems responsible for the phonetic analysis of speech are domain specific. Mattingly and Liberman (1988), for example, claim that they are different from the systems responsible for the analysis of nonspeech, as experiments have shown that how a signal sounds to a listener depends on whether the acoustic context indicates that the signal is linguistic or not. The very same signal that is heard as the onset of a syllable when the context specifies that the stimulus is speech may be heard as a chirp when it is isolated from the speech stream. This certainly seemed to be the case in Mattingly and Liberman's (1988) study: when a brief synthesised resonance of changing centre frequency (like one of the two in 3.2 (a)) was presented to participants in isolation, a nonspeech chirp was heard, but when the resonance was the transition (i.e. first formant) of a third-formant trajectory in a three speechlike formant-pattern (like one of two in 3.2 (b)), listeners heard either the consonant-vowel (CV) syllable [da] or the CV syllable [ga] depending on the slope of the transition. The pattern with the falling transition was heard as [da] and the one with rising transition as [ga]:



Figure 3.2 A schematic illustrating the domain specificity of speech perception where (a) illustrates isolated resonances and (b) those same resonances producing [d] and [g] in a speechlike context (adapted from Mattingly and Liberman 1988: 69)

Mattingly and Liberman (1990: 502) argue that 'when the resonance is presented in isolation, the auditory module for timbre interprets its center-frequency slope as a chirp, but when this same resonance is in the appropriate context, the phonetic module interprets its slope, together with other parts of the pattern, as a phonetic event'. The rather strong implication of this is that the computational systems that come into play in the perceptual analysis of speech are domain specific in that they are only able to operate on acoustic signals that appear to be part of a speech stream (Liberman et al. 1967; Fodor et al. 1974).

## 2.1.1.5 Inaccessibility and encapsulation

The final two properties on Fodor's (1983) list, limited central accessibility and informational encapsulation, are closely linked in that they both pertain to the character of information flow across systems, albeit in different directions: while limited central accessibility restricts the flow of information out of a system, informational encapsulation restricts the flow of information into it (Robbins 2015).

Let us look at limited central accessibility first. For Fodor (1983), a system is inaccessible if its inner workings are opaque to introspection – that is, if the intermediate-level representations it computes are inaccessible to consciousness. This, Prinz (2006) points out, is a characteristic that's seen in the peripheral systems, given that despite our ability to make use of the five (traditionally recognised) senses of seeing (opthalmoception), hearing (audioception), taste (gustaception), smell (olfaception), and touch (tactioception) every day, we have no introspective access to how our sensory systems operate. Likewise, although we have tacit knowledge of our native language (i.e. native speaker competence) and the ability to produce it, we do not understand the operations of each and every part of our linguistic system (Davenport and Hannahs 2010), note.

Informational encapsulation, on the other hand, concerns the information a cognitive mechanism has access to. It has been defined by Fodor (1983: 69) as 'the claim that the data that can bear on the confirmation of perceptual hypotheses includes, in the general case, considerably less than the organism may know'. An informationally encapsulated system is one that 'ha[s] access to only some of the information available to the mind as a whole' (Chien 1996: 1), that operates in isolation from any information that's stored beyond it, sensitive only to information stored within the mechanism itself, (in a proprietary database, say) and any input it receives (Fodor 1983; Prinz 2006; Robbins 2015). Being informationally encapsulated, modules neither consult one another, nor are they 'susceptible to influence from information from higher levels, levels at or above that at which they deliver outputs' (Currie and Sterenly 2000: 147) for example general memory in central cognition (Collins 2005). They simply operate quickly and automatically on what they take as input, and can do so precisely because their processes' databases are so well defined (Collins 2005) – this is a point I will come back to in section 2.2.

The encapsulation of perceptual systems 'is evidenced directly by the fact that perception does not change when it conflicts with belief, as in cases of visual illusion', Currie and Sterenly (2000: 148), argue. And that certainly seems to be the case. Consider this example from Fodor (1983: 66-67):

When you move your head, or your eyes, the flow of images across the retina may be identical to what it would be were the head and the eyes to remain stationary while the scene moves. So: why don't we experience apparent motion when we move our eyes? Most psychologists now accept one or another version of the "corollary discharge" answer to this problem. According to this story, the neural centers which initiate head and eye motions communicate with the input analyser in charge of interpreting visual stimulations. Because the latter system knows what the former is up to, it is able to discount alterations in the retinal flow that are due to the motions of the receptive organs.

Well, the point of interest for us is that the visual-motor system is informationally encapsulated. Witness the fact that, if you (gently) push your eyeball with your finger (as opposed to moving it in the usual way: by an exercise of the will), you *do* get apparent motion. Consider the moral: when you voluntarily move your eyeball with your finger, you certainly are possessed of the information that it's your eye (and not the visual scene) that is moving[, b]ut this explicit information, available to you for (e.g.) report, is *not* available to the analyzer in charge of the perceptual integration of your retinal stimulations. That system has access to corollary discharges from the motor center *and to no other information that you possess*.

Since a cognitive processor is one that has access to nothing more than the information stored within the local structures that subserve it, it is, of course, the *lack* of access to knowledge about what the finger is doing that demonstrates the encapsulation of the visual-motor system in this example. The visual mechanisms involved in this exercise are simply not designed to take any externally caused movement of the eyeball or any knowledge of that into account (Barrett 2005).

The Müller-Lyer (Day 1989) illusion (which also has to do with visual perception) (see Figure 3.3) is also said to support Fodor's (1983) claim that input systems are modular. In this illusion, two equal-lengthed parallel lines are flanked by arrows pointing inwardly in one case and outwardly in the other, but although the lines are of equal length, the one with outward-pointing arrows consistently appears shorter to those that perceive them. Even when people are made aware that the two lines are the same length, the lines still look as though they are different lengths to them; they cannot use their knowledge that they are not to alter their visual perception that they are. This, Fodor (1983) says, suggests that visual perceptive processes are encapsulated from module-external information (such as dimensions, in this case):

31



Figure 3.3 A Müller-Lyer illusion with typical central shaft and inward and outward fins

A similar effect might be observed if somebody asked one of their friends to remove something that was stuck in their eye, for it's likely that they would blink as their friend's fingers approached it in spite of trusting that their friend wouldn't hurt them (Chien 1996). If this effect was observed, it could count as strong evidence to suggest that the eyelid reflex is encapsulated, sensitive only to motions around the eye and not one's beliefs about other people (Chien 1996). It is for this reason that 'informational encapsulation' is sometimes used interchangeably with 'cognitive impenetrability': what one believes does not (and cannot) make a difference to how modules work.

Now although Fodor didn't distinguish the nine properties described above with respect to weight or priority in 1983, in later essays (e.g. Fodor 2000) he emphasised informational encapsulation at the exclusion of others, as the defining feature of modularity (Garfield 1987; Applebaum 1998).

#### 2.2 What is and isn't modular for Fodor

We obviously know now from Chapter 2 (and the first section of this one) that Fodor (1983) wrote a bold, two-part thesis about the structure of the mind (Wilson 2005). His first claim is a positive one, that part of it (i.e. its input systems (those responsible for lower-level cognitive functions such as perception and language)) is modular while the second is negative, that central cognition (whose systems are responsible for higher-order cognitive functions such as belief fixation and practical reasoning) is not (Robbins 2015).

32

Fodor (1983) characterises some of the components of psychological systems by analogy to the organisation of computers (or, more specifically, Turing machines), which are, for Fodor (1983: 39), 'as general [an example of a central processing unit] as any kind of computer can be'. Turing machines are, according to Fodor (1983: 39), informationally encapsulated devices in that 'the sole determinants of their computations are the current machine state, the tape configuration, and the program [with] the rest of the world being quite irrelevant to the character of their performance'. Organisms 'are forever exchanging information with their environments, and much of their psychological structure[s are] constituted of mechanisms which function to mediate such exchanges' (Fodor 1983: 39), meanwhile, so if we are to model anything in cognitive psychology on Turing machines, it must not be the mind as a whole, but the input systems that are 'embedded in a matrix of subsidiary systems [(i.e. transducers)] which affect their computations in ways that are responsive to the flow of environmental events [and function] to provide the [input systems] with information about the world' (Fodor 1983: 39).

These subsidiary systems (or transducers) Fodor (1983) speaks of on page 39 are helpfully distinguished from the input systems and central processors in the trichotomous taxonomy he provides on page 42:

Input systems function to get information into the central processors; specifically, they mediate between transducer outputs and central cognitive mechanisms by encoding the mental representations which provide domains for the operations of the latter. [...] Whereas transducer outputs are most naturally interpreted as specifying the distribution of stimulations at the 'surfaces' (as it were) of the organism, the input systems deliver representations that are most naturally interpreted as characterizing the arrangement of *things in the world*. Input analyzers are thus inference-performance systems within the usual limitations of that metaphor. Specifically, the inferences at issue have as their 'premises' transduced representations of proximal stimulus configurations, and as their 'conclusions' representations of the character and distribution of distal objects

In other words, there are three different types of mental mechanism for Fodor (1983): transducers, input systems, and central systems. Transducers lie at the interface between the mind and the world producing symbolic mental representations as output from the physical, non-symbolic input they find externally (Cain 2013, 2016); that is, they convert the energy that impinges on the body's sensory surfaces into something that's computable by the input systems (Robbins 2015). The retina (i.e. the light-sensitive tissue that lines the back of the

eye), for example, is a transducer that in response to being sensorily stimulated by light, produces symbolic output that represents the light's properties (e.g. its intensity, propagation direction, frequency or wavelength, to name a few (Buser and Imbert 1992)) (Cain 2013).

Input systems, on the other hand, are mechanisms that function to 'present[...] the world to thought' (Fodor 1983: 40) by taking the outputs of the sensory transducers as input and transforming them into representations of their distal causes as output by means of computation (Cain 2013). These output representations are then passed on to the central system which decides on these bases what decisions are to be made about the external world (Cain 2016).

As an example, let's consider what happens when one smells something sweet. First of all, sensory information is conveyed to the central nervous system (CNS) and perceived in four steps (as exemplified in Figure 3.4): (1) stimulation (where 'a physical stimulus impinges on a sensory neuron or an accessory structure' (Raven 2006: 1104)); (2) transduction (where 'the stimulus energy is used to produce electrochemical nerve impulses in the dendrites of the sensory neuron' (Raven 2006: 1004)); (3) transmission (where 'the axon of the sensory neuron conducts action potentials along an afferent pathway to the CNS' (Raven 2006: 1104)) and (4) interpretation (where 'the brain creates a sensory perception from the electrochemical events produced by afferent stimulation' (Raven 2006: 1104)). It is fair to say that we actually smell (as well as see, hear, touch and taste) with our brains as opposed to with our sensory organs then.



Figure 3.4 The path of sensory information (Raven 2006: 1104)

When somebody smells something sweet, the chemoreceptors or ciliated neurons located in the linings of their nasal passages transduce (i.e. respond to) chemical substances present in their environment (in particular Glycophore d = 2.6 compounds which are made up of two oxygen atoms that are separated by a diagonal distance of d = 2.6 Angstrom and attached to two carbon atoms in a cis- configuration (Fulton 2005)) in order to generate electrochemical nerve impulses as output. These are then taken as input by the neurons in the olfactory input system which transmit impulses through their axons to the brain for interpretation via the olfactory nerve:



Figure 3.5 Smell (Raven 2006: 1110)

The central system might then use the information that's been made available to it about something smelling sweet together with one's memory of past events that something that smelled sweet tasted sweet too to fixate the belief that the same correlation will be found in this instance; that whatever they have in front of them will taste as sweet as it smells.

The olfactory system is only one of six input systems, however, for there are input systems that correspond to each of the other traditionally conceived senses as well (i.e. the visual system for seeing, the auditory system for hearing, the gustatory system for tasting and the somatosensory system for touching) as well as the one for language (Cain 2013) which functions to generate from the auditory and visual information associated with the spoken and signed modalities of language, respectively, their underlying semantic, syntactic and phonological structures. I will have more to say about this, peripheral modularity, that is, in section 2.2.1.

# 2.2.1 Peripheral modularity

Like transducers, input systems are reflexive and automatic, and like cognitive mechanisms, they are inferential and computational, Fodor (1983) claims (Applebaum 1998). They are also informationally encapsulated, he says, in that they operate to map characteristic inputs onto characteristic outputs using the informational resources stored in their proprietary databases only. This makes them modular by definition (Fodor 2000).

Because of the arbitrary constraints on input systems' access to information, they are said to be 'irrational', and because they operate blindly in the sense that they do so without being able to access all that the organism knows they are said to be 'dumb'. Irrational and dumb aren't actually bad things for modules to be though, it turns out, for it is precisely because of the restrictions of the data that's available to modules that they are able to process things as quickly and mandatorily as they do (Carruthers 2005). On pages 64 through 71 Fodor (1983) emphasises these points:

[T]he point of the informational encapsulation of input systems [...] is to restrict the number of *confirmation relations* that need to be estimated as to make perceptual identifications fast. (Fodor 1983: 71)

The informational encapsulation of the input systems is [...] the essence of the analogy between the input systems and the reflexes. (Fodor 1983: 71)

Because [the input systems'] processes are automatic, you save computation (hence time) that would otherwise have to be devoted to deciding whether, and how, they ought to be performed. (Fodor 1983: 64)

Central systems, on the other hand, 'operate without antecedently established constraints on the information they are able to recruit in the course of their operations' (Applebaum 1998: 319) and are therefore rational, smart, but most importantly amodular (due to their lack of informational encapsulation), in Fodor's (1983, 2000) eyes.

# 2.2.2 Central nonmodularity

Fodor's (1983, 2003) argument for the nonmodularity of central systems has been broken down by Robbins (2015: 13) as follows:

- 1. Central systems are responsible for belief fixation.
- 2. Belief fixation is isotropic and Quinean.
- Isotropic and Quinean processes cannot be carried out by informationally encapsulated systems.

# *Hence* (from 2 and 3):

4. Belief fixation cannot be carried out by an informationally encapsulated system.

But:

5. Modular systems are informationally encapsulated.

*Hence* (from 4 and 5):

6. Belief fixation cannot be carried out be a modular system.

Hence (from 1 and 6):

7. Central systems are not modular.

Two terms call for explication in this argument (*isotropy* and *Quineanness*), and so I will do that here. Both have to do with the confirmation of scientific hypotheses (Wilson 2005). Isotropy, as defined by Fodor (1983: 105), refers to the epistemic interconnectedness of beliefs in that 'everything the scientist knows is, in principle, relevant to determining what else he ought to believe'. Carruthers (2003a: 76) gives the following example:

As an illustration of the supposed holism of belief, consider an episode from the history of science. Shortly after the publication of *The Origin of Species* a leading physicist, Sir William Thompson, pointed out that Darwin couldn't just assume the long time-scale required for gradual evolution from small differences between individual organisms, because the rate of cooling of the sun meant that the Earth would have been too hot for life to survive at such early dates. Now we realize that the Victorian physicists had too high a value for the rate at which the sun is cooling down because they were unaware of radioactive effects. But at the time this was taken as a serious problem for Darwinian theory – and rightly so, in the scientific context of the day.

To say that scientific confirmation is Quinean, on the other hand, is to say that 'the degree of confirmation assigned to any given hypothesis is sensitive to properties of the entire belief system' (Fodor 1983: 107), that 'the shape of our whole science bears on the epistemic status of each scientific hypothesis' (Fodor 1983: 107). Both isotropy and Quineanness preclude the encapsulation (and therefore modularity) of central systems since a system's possession of those two features requires it to have 'potentially unlimited access to the contents of central memory' (Prinz 2006). Put in slightly different terms, isotropy and Quineanness are global

properties, and since globality rules out encapsulation, so too do central processes (which means they cannot be modular) (Prinz 2006).

For many, Fodor's (1983) argument is a difficult one to resist, which is unsurprising, really, given its logicality. His main points are these: first, that there is a strong negative correlation between globality and encapsulation, second that there is a strong positive correlation between encapsulation and modularity, and from those two points a third, that there is, therefore, a strong negative correlation between *globality* and modularity (Robbins 2015). For Fodor (1983), the more global a process is, the less modular the system is that performs it (Robbins 2015).

There are three ways, then, that Fodor's (1983) conclusion can be countered: the first way is to deny that central processes are global, the second is to deny globality and encapsulation are strongly negatively correlated and the third to deny that encapsulation and modularity are strongly positively correlated (Prinz 2006). The third option (denying that modularity requires encapsulation) is the approach Carruthers (2006a) takes. In *The Architecture of the Mind* Carruthers (2006a) draws a distinction between two kinds of encapsulation: 'narrow-scope' encapsulation (which is seen in systems that cannot draw on information beyond their system boundaries during the course of their processing and corresponds to Fodor's (1983) use of the term encapsulation) and 'wide-scope' encapsulation (which is seen in systems that indeed do have access to exogenous information during the course of their processing, but cannot access everything all at once). This, wide-scope encapsulation, that is, is a much weaker sense of encapsulation than the one Fodor's (1983) proposed. As such, it allows for the central systems to be seen as modular enterprises after all.

#### 3. Massive modularity

Researchers in the cognitive science community have gone a step further than Fodor (1983, 2000) by claiming that the mind is wholly, or at the very least massively, composed of modules (Cosmides and Tooby 1992, 1994; Tooby and Cosmides 1992; Sperber 1994, 1996, 2002; Pinker 1997; Gallistel 2000; Barrett 2006; Barrett and Kurzban 2006). There are a number of good reasons, then, for thinking that this is so, and so I will begin this section by shedding light on some of these for they will bring to bear on later discussion. In sections 3.1.1 through 3.1.3 I will look briefly at some of the developmental, pathological and experimental evidence for the mind having such an architecture before honing in on evolutionary psychologists' principal theoretical argument for massive modularity in section

39

3.1.4, which has to do with evolution. Section 3.2 will be devoted to the discussion of Carruthers (2006a) arguments for massive modularity (the argument from design in section 3.2.1, the argument from animals in section 3.2.2 and the argument from computational tractability in section 3.2.3) and section 3.3 his definition of a module, for it differs from Fodor's (1983, 2000) in quite a number of ways.

#### 3.1 Evidence for modularity

## 3.1.1 Developmental evidence

A variety of arguments (that are themselves of varying strengths) have been put forward to support some form of (or at least, some facets of) a massively modular view of the mind (Carruthers 2003a). The first set (and indeed the set to be studied in this subsection) have to do with development.

Developmental psychologists now generally agree that cognitive development is a process that proceeds at different speeds in different domains (such as naïve physics, naïve psychology, and, for Atran (2002), naïve biology, for example) (Carruthers 2004). Children's competence in at least some of these domains is observable very early on in infancy (Carruthers 2003a; Caruthers 2004), in some cases as young as four months old (Carruthers 2004). Evidence for early competence in contact mechanics (and therefore naïve physics) has been found too, and so has evidence for early competence in social understanding (ergo naïve psychology) (Spelke 1994; Baillargeon 1995; Woodward 1998; Phillips et al. 2002).

Cognitive scientists have argued that children's abilities to know so much, so fast, with so little information available to them cannot be explained without deploying some sort of 'poverty of the stimulus' argument like Chomsky's that was put forward in support of the innateness of linguistic knowledge. According to Carruthers (1992, 2003a, 2004), Leslie (1994), Spelke (1994), Baron-Cohen (1995) and Dwyer (1999), among others, it is hard to see how children can acquire so much knowledge at such young ages with only general-learning mechanisms to help them. There must, therefore, be an innately channelled learning module for knowledge acquisition in each of the different domains they say, much like there is a learning module for the acquisition of linguistic knowledge.

#### 3.1.2 Pathological evidence

Related arguments for some domains being dissociable from others come from psychopathological damage studies (Shallice 1998; Tager-Flusberg 1999). During development, theory of mind can be damaged while physical/spatial thinking is not, for example, and the reverse is possible too, Carruthers (2004) points out.

The first case, actually, is what we see in autism (a developmental condition in which people show a selective impairment in the naïve psychological domain in that they can find it difficult to understand the mental states of themselves and other people but can be of normal intelligence otherwise (Baron-Cohen 1995)). The second case, on the other hand, is something seen in children with Williams' syndrome (a developmental disorder in which people have difficulties in the domain of practical problem solving (which of course, implicates naïve physics) but not necessarily anywhere else (Karmiloff-Smith et al. 1995; Mervis et al. 1999)).

There has also been evidence to suggest that the naïve biological domain can be subject to dissociable damage, particularly so from stroke victims who have category-specific semantic impairments in the realm of animate vs. inanimate things (Warrington and McCarthy 1983; Warrington and Shallice 1984; Sartori and Job 1988; Job and Surian 1998; Atran 2002). Warrington and McCarthy (1983) and Warrington and Shallice (1984) were among the first people to report this: Warrington and McCarthy (1983) described a patient who had suffered a left hemisphere stroke and had impaired production and comprehension performance for non-living objects but not living ones while Warrington and Shallice (1984) described the opposite pattern of breakdown in four patients with the herpes simplex encephalitis virus.

#### 3.1.3 Experimental evidence

There isn't as much experimental evidence for modularity of mind as there could be, for most of its proponents are theoretical, as opposed to empirical, researchers. What little evidence there is, however, is strongly suggestive that the mind is indeed modular (Carruthers 2004).

The first piece of evidence concerns the existence of a geometric module in rats. In 1986, Cheng trained rats to search for food in rectangular enclosures. On each trial, food was put in a particular corner, and 75 seconds after being shown the food, the disorientated rats were allowed to search for it. When there were no other cues, rats searched the correct corner and its geometrically equivalent corner (the one diagonally opposite). What was most interesting, however, is that the rats continued to confuse geometrically equivalent locations even in the presence of highly salient clues that could disambiguate the geometric information and that they would well recognise in other circumstances (such as different odours and colours). The rats' inability to integrate geometric information with information of other kinds led Cheng (1986) to suggest that the processing of local spatial geometry is modular, according to Fodor (1983)'s definition of the word, as to him the study suggested that geometric processing in rats was cognitively impenetrable (or informationally encapsulated, if you prefer).

Since then, similar paradigms of research have been conducted on humans (Cheng and Newcombe 2005). Hermer and Spelke (1994, 1996), for example, extended Cheng's (1986) work to human children and found that between the ages of two and four children behaved identically to rats when they were disorientated, splitting their choices evenly between the correct corner and the diagonally opposite corner of a rectangular room regardless of a nongeometric landmark (a blue wall) to relocate an object hidden in one of its four corners. Using the data from Hermer and Spelke (1994, 1996) (and indeed their own) as evidence, Hermer-Vazquez et al. (1999) went on to sketch out how they thought children might reorient. Until children acquire spatial language (at around six years of age), they will, like rats, reorient themselves on the basis of spatial representations within a geometric module of mind, they explain.

There is also evidence to suggest that Pavlovian (1955) associationist models of simple conditioning are 'subserved by a special-purpose computational system that is designed to predict varying temporal contingencies' (Carruthers 2004: 306). In these sorts of model, a behaviourally neutral stimulus called the conditioned stimulus is repeatedly paired with a motivating stimulus called the unconditioned stimulus until the subject responds to the former in a manner that demonstrates their anticipation of the latter (Gallistel and Gibbon 2001). In Pavlov's experiments which investigated the conditioning and learning by association, recall, a dog that was exposed to the ring of a bell (the conditioned stimulus) whenever it was fed (the unconditioned stimulus) over and over again, learnt to associate the sound of the bell with the expectation that food would come and would salivate upon hearing it.

Gallistel and Gibbon's (2001) argument for there being an information processing computational module that subserves associationist conditioning is as follows: neither the delay between stimuli and reinforcements nor the ratio of reinforced to unreinforced presentations of conditioned stimuli have been found to affect rates of acquisition. And, while

42

neither of these things can be understood from a perspective that sees learning as the mere building of associative connections (i.e. Hebbian synapses), they can be easily explained within a computational model which assumes that what animals really do is estimate likelihoods and calculate rates of return (Gallistel and Gibbon 2001).

## 3.1.4 Evidence from evolutionary psychologists

The fourth and final set of arguments (to be discussed in section 3.1, at least) for massive modularity are evolutionary in nature. According to evolutionary psychologists such as Tooby and Cosmides (1995), Fiddick et al. (2000) and Cosmides and Tooby (2001), biological systems (which they take to include the mind as well) characteristically evolve by becoming more modular in response to evolutionary pressures from their environment, by 'bolting on' additional structures that are specialised to perform particular tasks and solve certain problems (Carruthers 2004; Carruthers 2006a). The anti-modular 'general-purpose computer' model of the mind cannot be correct, in their eyes, for no such computer could evolve in the way that the mind seems to.

There is indeed evidence to suggest that this is the case: computational simulations by Kashtan and Alon (2005) and Kashtan et al. (2007), for example, have shown that networks evolve modularity and evolvability in environments that are subject to change, while evolution in environments that are unchanging produces nonmodular networks that are comparatively slower to adapt (Kashtan and Alon 2005; Kashtan et al. 2007). Follow up studies undertaken by Parter et al. (2007) further support this thesis that it is an organism's changing environment that generates its modular structure. They found that the modularity of bacterial metabolic networks is positively correlated with the frequency and degree to which their environments change.

The argument is summarised by Tooby and Cosmides (1995: xiii-xv) as follows:

just as the body contains distinct morphological adaptations that evolved to perform distinct physiological functions, the mind must contain distinct "mental organs," which evolved to solve a Vast Number of adaptive problems that required distinct behavioral solutions (Symons 1992, p. 142). Since the behavioral solution to any one of these problems wouldn't have transferred to any of the other problems [...] each adaptive problem would have selected for its own specialized cognitive adaptation. Thus, "our cognitive architecture resembles a confederation of hundreds of functionally dedicated computers (often called modules In their eyes, when we need a new skill, a new system simply evolves to subserve that. I will come back to this point in sections 3.2.1 and 3.2.2, as this has to do with Carruthers' arguments for massive modularity.

## 3.2 Carruthers' arguments for massive modularity

In his 2006 publication *The Architecture of the Mind*, Carruthers offers a clear and comprehensive defence of the idea that the human mind is massively modular. He begins by providing a definition of modularity that's appropriate to the thesis that central cognition is modular too, for unlike Fodorian (1983) modules that can be characterised by nine features, Carrutherian (2006a) modules are more simply than that function-specific automatic processing systems which are independent of one another and whose internal operations are largely inaccessible to the rest of cognition. The result is a notion of modularity that is dramatically different from Fodor's (1983) – it is much closer to the use of notion in the biological research domain and even closer to its use by researchers in computing (a point I will return to in section 3.3). In the book's first three chapters Carruthers (2006a) put forward three arguments for massive modularity (the argument from design, the argument from animals and the argument from computational tractability). Each of those will be addressed in turn here.

# 3.2.1 The argument from design

## 3.2.1.1 Biological modules

The first (and arguably the most developed) of Carruthers' (2006a) arguments for massive modularity (the argument from design) follows the lead of Simon (1962), who said that systems are optimally designed if they are modular, for the modular organisation of complex systems means that they are:

constructed hierarchically out of dissociable sub-systems (each of which is made up of yet further sub-systems), in such a way that the whole assembly can be built up gradually, adding sub-system to subsystem; where the properties of sub-systems can be varied independently of one another; and in such a way that the functionality of the whole is buffered, to some extent, from changes of damage occurring to the parts (Carruthers 2006a: 12)

He told the tale of two watchmakers, Tempus and Hora, to illustrate this point. Tempus and Hora both made watches from myriad parts and were interrupted frequently by phone calls as they worked, Simon (1962) said. Hora's designs were no less complex than Tempus' were, but while Hora designed his watches as decomposable systems made up of modules, Tempus did not, and so each and every time Tempus was interrupted by the telephone and had to set aside his work to answer it, he had to come back and reassemble the whole watch from scratch (Simon 1962). By contrast, Hora designed small stable subassemblies (or microassemblies) that could be fashioned into larger stable subassemblies, and so when he was interrupted by the telephone, only the last unfinished subassembly had to be reassembled; his earlier work, meanwhile, was of course preserved (Simon 1962).

What made Tempus' watch design so unstable in the end was not the number of parts it had but the interdependencies among them (Langlois 1999). This is because in a nondecomposable system (like Tempus'), the successful operation of one part depends on the successful operations of all others, meaning that when a system is missing parts, it is no longer functional (Langlois 1999). In a decomposable system (like Hora's), on the other hand, the proper working of a part depends more on the parts within its subassembly than the parts outside of it, which means that it may well be able to operate as it should even if some of the system's other parts are missing (Langlois 1999).

The modularity of technological systems then, by limiting the scope of interactions between elements of tasks, can shorten their designs' developmental times, as it eliminates the need for pre-assembly operations Wilhem (1997) points out. It can also reduce their cost of production (Garud and Kumaraswamy 1995; Sanchez and Mahoney 1996; Muffatto 1999; Cusumano and Nobeoka 1998), economise their scale and scope (Pine 1993; Friedland 1994; Garud and Kumaraswamy 1995; Sanchez 1999) and increase their number of compatible suppliers (Langlois 1992, 2000; Langlois and Robertson 1992; Garud and Kumaraswamy 1993; Morris and Ferguson 1993; Reed 1996; Baldwin and Clark 1997; Sanderson and Uzumeri 1997). The main thing that artificial systems (like technological ones) have in common with natural systems (like biological ones), though, is that their modularity allows their structures to evolve more quickly and easily than they would have been able to had their architecture not been modular. Just like technological systems have to be modularised in order that they can evolve through new parts being added and old ones subtracted, debugged, improved or updated without running the risk of errors being introduced elsewhere (Carruthers 2006a), modularity in complex biological systems is essential for their biological evolution through natural selection (Wagner and Altenberg 1996; Bolker 2000; Raff and Raff 2000; Wimsatt

45

and Schank 2004; Carruthers 2006a; Futuyma 2009; Epinosa-Soto and Wagner 2010 (see also, Carruthers (2006a: 21-22)):

The basic reason why biological systems are organized hierarchically in modular fashion is a constraint of evolvability. Evolution needs to be able to add new functions without disrupting those that already exist; and it needs to be able to tinker with the operations of a given functional sub-system – either debugging it, or altering its processing in responses to changes in external circumstances – without affecting the functionality of the remainder.

Simon's (1962) argument is really a one from *design*, then, whether the designer is a human engineer (as in the case of technological systems) or natural selection (as in the case of biological ones) (Carruthers 2006a).

## 3.2.1.2 Is the mind a biological module?

Carruthers (2006a: 13) argues that a belief in massive modularity is 'well nigh ubiquitous across the biological sciences' and that while not everyone uses the word 'modularity', others are used to denote the same concept (Nijhout (1991) uses 'autonomy', Maynard-Smith and Szathmáry (1995) and Kirschner and Gerhart (1998) use 'compartmentalisation', Larson and Losos (1996) use 'individualisation' and West-Eberhard (1996) uses 'sub-unit organisation'). There is indeed a great deal of evidence from the field of biology, suggesting that complex biological systems are modular. This has found to be true for many levels of the organism such as genes, cells, cellular assemblies, whole organs, organ assemblies, the organism itself and even multi-organism units (Seeley 1995; West-Eberhard 2003; Schlosser and Wagner 2004; Callebaut and Rasskin-Gutman 2005). Consider the evolution of orchids, for example, who have adapted through natural selection to promote reproduction (Schiestl et al. 2000).

Orchids reproduce through pseudocopulatory pollination (so-called because the behaviours of the individuals involved mimic copulation even though there is no actual sexual activity involved (Schiestl et al. 2000)). The pollinator sees the flower as a potential mate and attempts to engage in coitus. In doing so, its body makes contact with the viscidium of the flower, and pollinium is pulled from the anther, which is connected to the viscidium by something called the caudicle (Schiestl et al. 2000). The pollinator then leaves and approaches another flower of the same species to copulate, depositing the pollinium to the stigma of the second flower and thereby completing the pollination process (Schiestl et al. 2000).

In order to increase its chances of pollination, the orchid may mimic a potential female mate for the pollinator visually (van der Pijl and Dodson 1966). A prime example of this can be seen in the hammer orchid (*Drakaea*), which is highly modified in that its labellum resembles a female *Thynnid* wasp in both shape and colour to lure in potential partners to pick up and deposit its pollen to another *Drakaea* (Groom and Lamont 2015).

Orchids are also known to exploit the attraction of male insects to female sex pheromones by emitting a scent that mimics that of a potential mate (Tan and Nishida 2000). The early spider orchid (*Ophrys sphegodes*), for example, which is pollinated by *Adrena nigroaenaea* males, secretes chemicals that smell similar to the sex pheromones of a female *Andrena nigroaenaea* from glands called osmophores in order to attract them (Shiestl et al. 2000).

Note that in each of the cases presented here, only particular parts of organisms undergo adaptations: in *Artemia salina*, it is the osmoregulation system, in *Drakaea* it is the labellum and in *Ophrys sphegodes*, it is the osmophores. This is because of the organisms' modular organisations: the modularity of organisms allows evolution to occur as it allows the functions of units to be altered without altering the functioning of other units (Riedl 1978; Gilbert et al. 1996; Raff 1996).

This is also the case in animals as well as plans. Nobody supposes that there could be a single general-purpose sensory organ for fulfilling all of the five sensory functions. On the contrary what we find are five distinct organs that have been shaped by natural selection to fulfil each of the five sensory functions – the eyes for seeing, the ears for hearing, the nose for smelling, the tongue for tasting and the skin for touching (Carruthers 2004). By the same token, nobody expects to find a single general-purpose organ that fulfils both the functions of the heart and lungs or the cardiovascular system and the respiratory system. Rather, the cardiovascular system is made up of the heart and blood vessels (arteries, capillaries and veins) and the respiratory system the nose, mouth, pharynx, larynx, trachea, bronchi and lungs (Carruthers 2004). The same, then, should surely be true of the mind, Carruthers argues, if that is a biological system too: one 'should expect there to be one distinct subsystem for each reliably recurring function that human minds are called upon to perform' (Carruthers 2006b: 10) – especially seeing as the senses, which like language are subserved by the input systems, are thought of as being modular.

## 3.2.1.3 Carruthers' argument from design summarised

In short, Carruthers (2006a) argues that the mind is an evolved biological system and as such, must, therefore, obey the same principles of design that other biological systems do. His argument is neatly summarised on page 25 as follows, and is closely related to his second argument from animals, which will be the focus of section 3.2.2):

- (1) Biological systems are optimally designed systems, constructed incrementally.
- (2) Such systems, when complex, need to have massively modular organization.
- (3) The human mind is a biological system, and is complex.
- (4) So the human mind will be massively modular in its organization.

#### 3.2.2 The argument from animals

The computational theory of mind has been the principal position in the field of philosophy on how the mind works for the past five plus decades. The first suggestion that something like the Turing machine might model the mind well was made by McCulloch and Pitts in 1943, and by the 1960s Turing computation came to be central to cognitive science (Rescorla 2017). The classical computational theory of mind has two basic tenets. The first is that the mind is a representational system and the second is that mental representations are processed computationally (Sterenly 1990). I consider these both here.

Now, the problem most people have with the computational theory of mind is that, because it has the word 'computational' in its title, they are inclined to think of the mind as a computer – but that's not what the theory intends (Rescorla 2017). Rather, what the computational theory of mind proposes, is for the mind to be thought of as a computational *system*. Describing the mind as a computer strongly suggests that it is susceptible to programming which neither minds (nor most Turing-style computational systems, for that matter) are (Rescorla 2017). As a result, a number of critics (e.g. Churchland et al. 1990) have falsely objected to the computational theory of mind by virtue of their thinking that the mind is not like a programmable, general-purpose computer. In reality, though, nobody is saying it is.

With that being said, some of the most widely accepted articulations of the computational theory of mind are due to Fodor (1975, 1987, 2000) (see also Fodor and Pylyshyn 1988). One of the main contributions he made was that the mind has an inner medium that carries out computations, and that compositionality, productivity and systematicity of thought could only be explained if that inner medium has a structure of language which has those three

properties. According to the language of thought hypothesis, thought and thinking are done in a mental language where words express concepts and sentences propositional attitudes (Murat 2015).

The language of thought hypothesis as it is understood today can be characterised as a combination of the following three theses (Murat 2015: 2-3):

- A. Representational Theory of Mind (RTM) (cf. Field 1978:37, Fodor 1987:17):
  - Representational Theory of Thought: For each propositional attitude *A*, there is a unique and distinct (i.e. dedicated)-psychological relation *R*, and for all propositions *P* and subjects *S*, *S As* that *P* if and only if there is a mental representation #*P*# such that
    - a. *S* bears *R* to #P#, and
    - b. #P# means that *P*.
  - 2. Representational Theory of Thinking: Mental processes, thinking in particular, consists of causal sequences of tokenings of mental representations.
- B. Mental representations, which, as per (A1), constitute the direct "objects" of propositional attitudes, belong to a representational or symbolic *system* which is such that (cf. Fodor and Pylyshyn 1988:12–3)
  - representations of the system have a combinatorial syntax and semantics: structurally complex (molecular) representations are systematically built up out of structurally simple (atomic) constituents, and the semantic content of a molecular representation is a function of the semantic content of its atomic constituents together with its syntactic/formal structure, *and*
  - the operations on representations (constituting, as per (A2), the domain of mental processes, thinking) are causally sensitive to the syntactic/formal structure of representations defined by this combinatorial syntax.
- C. Functionalist Materialism. Mental representations so characterized are, at some suitable level, functionally characterizable entities that are (possibly, multiply) realized by the physical properties of the subject having propositional attitudes (if the subject is an organism, then the realizing properties are presumably the neurophysiological properties of the brain).

What is most interesting about the language of thought hypothesis however, as Carruthers (2006a) points out, is that it has important implications for animal cognition. According to Fodor (1975, 1987, 2000), language of thought is not dependent on any natural language, which leaves open the possibility that non-linguistic creatures (i.e. animals) can think as well (Carruthers 2006a).

Recall from 3.2.1 (i.e. Carruthers' (2006a) argument from design) that the modularity of cognitive capacities constitutes an evolutionary feature than enhances the adaptability of organisms so that they can cope more efficiently with their environments. In his second argument for massive modularity (the one which comes from animals), Carruthers (2006a) develops this point further, claiming that non-human animal minds are also massively modular and that the characteristic was conserved during the evolutionary transition from the animal mind to the human mind as valuable characteristics often are.

Unlike the argument from design, the argument from animals isn't summarised in Carruthers (2006a), though it has been helpfully broken down by Wilson (2008: 278) like so:

- (1) Animal minds are massively modular.
- (2) Human minds are incremental extensions of animal minds.
- (3) So human minds are massively modular.

## 3.2.2.1 Types of module

## 3.2.2.1.1 Learning modules

Following Gallistel (1990, 2000) and Tooby and Cosmides (1992; 1995), Carruthers begins his argument from animals with a 'reflection on the differing task demands of the very different learning challenges that people and animals must face, as well as the demands of generating appropriate fitness-enhancing intrinsic desires' (Carruthers 2006a: 29). It is one task to work out the sun's azimuth angle (which, with other measures such as its zenith angle) defines its position in the sky at any given time of day (Seinfield and Spyros 2006; Sukhatme 2008; Duffie and Beckman 2013) and use it to find your way from point A to point B, but it is quite another to follow the Polaris star north, say, and turn 90 degrees clockwise to go east, another 90 to go south, and a further 90 to head west, Carruthers (2006a) says. These are both learning problems animals can solve, but they require quite different learning mechanisms to do so (Carruthers 2006a). To reinforce that argument, Carruthers (2006a) widens his focus on navigation to a number of other learning problems such as vision, speech recognition, mind-reading, cheater detection and complex skill acquisition. Each of these learning problems pose computational challenges that are distinct from the computational challenges of other learning problems, he points out, and '[s]o for each problem, we should postulate the existence of a distinct learning mechanism, whose internal processes are computationally specialized in the way required to solve the task', he says (Carruthers 2006a: 29). Carruthers (2006a: 29) finds it 'very hard to believe that there could be any sort of *general* learning mechanism that could perform all of these different roles', and he is not alone in this view. Gallistel (2000), Gallistel and Gibbon (2001) and Gallistel et al. (2001) have all argued forcefully, for example, that there is *no such thing*.

## 3.2.2.1.2 Desire modules

Next, Carruthers (2006a: 30) defines learning as a process that 'issues in true beliefs, or beliefs that are close enough to the truth to support [the] inclusive fitness' of an individual. Desires, he says, therefore aren't learned in *that* sense of the word, but need to be formed in ways that will support an individual's inclusive fitness still. Carruthers points out that while some desires are 'instrumental' in that they are 'derived from ultimate goals together with beliefs about the means that would be sufficient for realizing [them]' (Carruthers 2006a: 30), not all acquired desires are formed that way. Rather, 'evolutionary psychology postulates a rich network of systems for generating new desires in the light of input from the environment and background beliefs' (Carruthers 2006a: 30).

Much of these desires, Carruthers (2006a) says, are, therefore, 'ultimate' in the sense that they are 'produced by inferences taking place in systems dedicated to creating desires of that sort' as opposed to being 'produced by reasoning backwards from the [need] to fulfil some other desire' (Carruthers 2006a: 30). A desire to have sex with someone, for example, isn't always produced by reasoning that such an act is likely to fulfil an evolutionary goal of reproduction, he argues. Rather, it can be generated by some sort of system (module) that has evolved for that purpose, he says (Carruthers 2006a).

## 3.2.2.1.3 Other modules

What emerges from Carruthers' (2006a) thoughts in Chapter 1 then is a picture of the mind being made up of a host of systems specialised for learning, and another host of systems that have been designed to generate fitness-enhancing desires, but although this looks like *much* of the mind is modular, it does not yet follow that the mind is composed *exclusively* of modules. Being well aware of this, Carruthers (2006a) goes on to argue in Chapter 2 that not only do animal minds consist of sets of learning modules and desire-generating modules, the systems charged with belief-generation and emotion-generation, and the selection, organisation and control of action are decomposable as well. As such, there is nothing left of the mind that could be considered nonmodular, for all of its primary functions (perception, belief, desire, practical reasoning and motor-control) are subserved by modules.

Carruthers (2006a) devotes all but one paragraph of Chapter 2 to defending the first premise of his argument from animals (that animal minds are massively modular) (Wilson 2008).<sup>1</sup> The final remaining one, however, begins as follows, and is taken by Wilson (2008) to be the reasoning behind premises (2) and (3) (that human minds are incremental extensions of animal minds and that human minds are massively modular, respectively):

Given that animal minds are organized along massively modular lines, then normal biological reasoning should lead us to expect that massively modular architecture will be preserved in the minds of members of *Homo sapiens*, too. (Carruthers 2006a: 149)

For Carruthers (2006a) it makes sense from an evolutionary perspective that if animal minds are massively modular, human minds will be as well. Fitness-enhancing biological structures are generally preserved in the evolutionary transition from one species to another, so provided it is appropriate to extend biological and evolutionary principles to minds, he says, we should expect the mind to be made up of the same modular systems that are present in apes.

## 3.2.3 The argument from computational tractability

Carruthers' (2006a) third and final argument in support of massive modularity is the one from computational tractability, which is derived in part from the earlier Fodorian (1983, 2000) tractability argument that Carruthers (2006a: 44) has helpfully summarised as follows:

- (1) The mind is computationally realized.
- (2) All computational mental processes must be suitably *tractable*.
- (3) Only processes that are informationally encapsulated are suitably tractable.

<sup>&</sup>lt;sup>1</sup> The chapter also locates those modules within a basic perception/belief/desire/practical reason/motor-control architecture, but I will say no more about that for it goes well beyond the scope of this thesis.

#### (4) So the mind must consist entirely of encapsulated computational systems.

Carruthers' (2006a) argument begins like Fodor's (1983, 2000) with the relatively uncontroversial claim, in the cognitive sciences, that cognitive systems are realised computationally (Gottschling 2009). If this is correct, then their computations must be suitably tractable, he (again, like Fodor (1983, 2000)) claims, which means they must be (at least in principle) able to be carried out within finite time (Gottschling 2009). The tractability of computations requires them, therefore, to be frugal in two respects: first in that the algorithms used in their computation must be suitably simple; and second in that they use in their computation only a limited amount of information (Gottschling 2009).

<sup>(</sup>[I]n order to be tractable, computations need to be encapsulated' (Carruthers 2006a: 51) says, <sup>(</sup>for only encapsulated processes can be appropriately frugal in the informational and computational resources that they require' (Carruthers 2006a: 51). This looks a lot like something Fodor (1983, 2000) would have said at first glance, but while that may be, his definition of encapsulation here is quite different from Fodor's (1983, 2000) it turns out, as section 3.2.3.1 explores.

#### 3.2.3.1 Narrow-scope vs. wide-scope encapsulation

On pages 57 through 59 Carruthers (2006a) draws a distinction between narrow-scope and wide-scope encapsulation, using a sketch from Atran (2002) of a practical-reasoning systems which takes as its initial input whatever is currently the strongest desire for P and conducts a search through long-term memory for beliefs of the form  $Q \rightarrow P$ . If it finds one, a belief of the form  $Q \rightarrow P$ , that is, it checks its database to see whether Q is something for which a motor-scheme exists (i.e. whether Q is something that can be done in the here and now). If so, it goes ahead and does that, but if not, it conducts another search, this time for beliefs of the form  $R \rightarrow Q$  and so on. The system also has a couple of simple stopping rules, Carruthers (2006a: 58) claims, e.g.: '[i]f it [...] go[es] more than n conditionals deep without success, or if it [...] search[es] for the right sort of conditional belief without finding one for more than some specified time t, then it stops and moves on to the next strongest desire', making it frugal 'both in the information that it uses, and in the complexity of its algorithms' – but does that count as it being encapsulated?

Well, not in the way that informational encapsulation is traditionally understood, Carruthers (2006a) stresses, for here [...] the practical-reasoning system can search within the total set of

the organism's beliefs, using structure-sensitive search rules' (Carruthers 2006a: 58), and those beliefs would not affect a cognitively impenetrable system. There does seem to be *a* sense in which the system's encapsulated though, he points out. And so he draws a distinction between narrow-scope and wide-scope encapsulation, where narrowly-scoped encapsulated systems are those that cannot be affected by *all* that's beyond them, and widely-scoped encapsulated systems those that can be affected by *some* but can't be by most. See below for an explanation:

Put as neutrally as possible, it can be said that the idea of an encapsulated system is the notion of a system whose internal operations *can't* be affected by *most or all* of the information held elsewhere in the mind. But there is a scope ambiguity here.<sup>[2]</sup> We can have the modal term 'can't' take narrow scope with respect to the quantifier, or we can have it take wide scope. In its narrow-scope form, an encapsulated system would be this: concerning most of the information held in the mind, the system in question *can't* be affected by *that* information in the course of its processing. Call this 'narrow-scope encapsulated system would be this: the system is such that it *can't* be affected by *most* of the information held in the mind, no the other hand, an encapsulated system would be this: the system is such that it *can't* be affected by *most* of the information held in the mind in the course of its processing. Call this 'such that it *can't* be affected by *most* of the information held in the mind in the course of its processing. Call this 'such that it *can't* be affected by *most* of the information held in the mind in the course of its processing. Call this 'such that it *can't* be affected by *most* of the information held in the mind in the course of its processing. Call this 'such that its 'such that it *can't* be affected by *most* of the information held in the mind in the course of its processing. Call this 'such that its 'such that 'such

Since wide-scope encapsulation of information is enough for a system to be frugal (and therefore tractable) as demonstrated by the above example, Carruthers (2006a) argues, premises (3) and (4) must be revised for the argument from computational tractability to be an accurate one. Narrow-scope encapsulation (i.e. encapsulation as it is traditionally understood) is not necessary, he says, and so the following is proposed instead (Robbins 2015: 20):

- (1) The mind is computationally realized.
- (2) All computational mental processes must be tractable.
- (3) Only processes that are at least weakly (i.e., wide-scope) encapsulated are tractable.
- (4) So the mind must consist entirely of at least weakly encapsulated systems.
- (5) So the mind is massively modular.

<sup>&</sup>lt;sup>2</sup> 'Modal terms like 'can' and 'can't' have wide scope if they govern the whole sentence in which they occur; they have narrow scope if they govern only a part. Compare: 'I can't kill everyone' (wide scope; equivalent to, 'It is impossible that I kill everyone') with; 'Everyone is such that I can't kill them' (narrow scope). The latter is equivalent to, 'I can't kill anyone.'' (Carruthers 2006a: 59)

The claim that modules must be encapsulated in Fodor's narrow-scoped sense was criticised by a number of researchers (e.g. Marlen-Wilson and Tyler 1987; Churchland 1988) before Carruthers drew a distinction between narrow- and wide-scope encapsulation in 2006a, but one of its least complementary critiques is actually due to Prinz (2006) who argues that the linguistic system rarely (if ever, in fact) satisfies this requirement. Prinz (2006) argues that top-down effects (effects of the high-level central systems) on linguistic processing could count as strong evidence that the input systems aren't encapsulated. He draws special attention to the phoneme restoration effect in this argument (which is a perceptual phenomenon in which, under certain conditions, sounds that are missing from the speech signal can be auditorily hallucinated by the brains of and therefore 'heard' by listeners). This can be so strong that listeners don't even pick up that there were actually any phonemes missing to begin with, they simply 'fill-in' the acoustic gap with a phoneme determined by the sentence's semantic interpretation (Warren and Warren 1970; Samuel 1987; Kashino 2006). If subjects hear *The eel is on the axel*, for example, they experience a /w/ sound in the gap, and if they hear *The eel is on the orange*, they experience a /p/ sound, Prinz (2006) points out.

Something similar is done for the perceptual systems by Churchland (1989), who argues that although Fodor (1983) found evidence to suggest the lack of power knowledge has over perception in the Müller-Lyer illusion (Day 1989), the well-known duck-rabbit figure (see Figure 3.6) shows the opposite, that there are 'a wide range of elements central to visual perception [...] which are cognitively penetrable' (Churchland 1989: 261):



Figure 3.6 The duck-rabbit figure (Jastrow 1899: 312)

Unlike in the Müller-Lyer illusion (Day 1989) where a person's knowledge that the two lines in it are of the same lengths cannot change the fact that they are seen to be of different ones, knowledge that the duck-rabbit figure is ambiguous and can be seen in one way or the other (i.e. either as a duck or as a rabbit) *can indeed* alter peoples' perceptions of the figure – which would suggest that the visual system has access to knowledge after all.

#### 3.3 Carrutherian modules

There are two cross-cutting spectra of opinion about how much of the mind is modular, then – Fodor's (1983, 2000) modest opinion and evolutionary psychologists' massive one (e.g. Smith and Tsimpli 1995; Tooby and Cosmides 1995; Segal 1996; Fiddick et al. 2000; Carruthers and Chamberlain 2000; Cosmides and Tooby 2001; Carruthers 2006a) – and the two camps' definitions of modularity clearly cannot be the same. According to Fodor (1983), whose argument was designed to apply to the input and output systems only, modules are domain specific and innately specified processing systems that have their own proprietary transducers and that deliver shallow (i.e. non-conceptual) outputs (Carruthers 2003a). They are also held to be 'mandatory in their operation, swift in their processing, isolated and inaccessible to the rest of cognition, associated with particular neural structures, liable to specific and characteristic patterns of breakdown, and develop according to a paced and distinctively-arranged sequence of growth' (Carruthers 2003a: 68).

Those that argue for a more modular account of cognition, however, have had to weaken Fodor's (1983) definition somewhat so that it can be applied to central systems as well (Smith and Tsimpli 1995; Segal 1996; Carruthers and Chamberlain 2000; Carruthers 2006a). In *The Architecture of the Mind,* for example, Carruthers (2006a: 12) lists which of Fodor's features would most likely have to go for his thesis of massive modularity to stand up:

[I]f a thesis of mental modularity is to be remotely plausible, then by 'module' we cannot mean 'Fodor-module'. In particular, the properties of having proprietary transducers, shallow outputs, fast processing, significant innateness or innate channeling, and encapsulation will very likely have to be struck out. This leaves us with the idea that modules might be isolable function-specific processing systems [...] which are domain specific (in the content [i.e. Fodorian] sense), whose operations aren't subject to the will, which are associated with specific neural structures (albeit sometimes spatially dispersed ones), and whose internal operations may be inaccessible to the remainder of cognition. Of the original set of descriptors for Fodorian modules, then, Carruthers strikes out four, leaving him with only domain specificity, dissociability, automaticity, neural localisability, and central inaccessibility. Informational encapsulation, notice, is absent from this list.

#### 4. Concluding remarks

In what follows, I consider the extent to which the phonological processor can be considered modular in Fodor's modest (1983, 2000) sense of the word (in Chapter 4) and modest in Carruthers (2006a) massive sense of it (in Chapter 5). For Fodor (2000) informational encapsulation is the defining feature of modularity, but for Carruthers (2006a) it is not as necessary as another feature – that feature being domain specificity. Carruthers (2006a) places greater emphasis on modules of mind having to be domain specific than he does informationally encapsulated because, to his mind (recall from the argument from design in section 3.2.1 and from animals in 3.2.2), modules only exist to subserve a certain function, and so for frugality purposes (recall from his argument from computational tractability in 3.2.3), they should only be able to operate on information that's relevant to their specialism. A module with a function specific to a domain is *ipso facto* only concerned with computing the kinds of input that have to do with that domain in Carruthers (2006a) eyes – and I am inclined to agree.
# **Chapter 4: A Modestly Modular Approach to Phonology**

### 1. Introduction

Formalism and functionalism are two diametrically opposed approaches to linguistics (Newmeyer 2010). This functionalist/formalist distinction is particularly apparent in phonology given its position within the architecture of the language faculty between the mental and the physical (Hannahs and Bosch 2018). Phonetics and phonology are traditionally thought of as being two different things. According to formal linguists, phonology is concerned with the connections between the internal grammar and the external sound system (that is, phonetics) (Hannahs and Bosch 2018)<sup>3</sup>. In recent years, however, some phonologists have come to conflate the two. This has led to a number of unconventional conclusions about phonology (and indeed the language faculty in which it resides) (Hannahs and Bosch 2018).

As Hannahs and Bosch (2018) rightly point out, usage-based phonological analyses owing to the likes of Bybee (2003, 2010) refer to facts about speech as it is used in the real world to inform their thinking about language, and corpus-based work (as in that of Durand et al. (2014), for example), does the same. Articulatory phonologists (e.g. Browman and Goldstein 1986) meanwhile, argue that the phonological system is organised in accordance with the ways in which speech is articulated, and even optimality theorists (such as Hayes et al. (2004) – who are more formal in their approach than Bybee (2003, 2010), Durand et al. (2014) and Browman and Goldstein (1986) put together) argue for the phonetic grounding of optimality constraints (Hannahs and Bosch 2018).

This blurring of the distinction between phonetics and phonology has gone as far as to lead some scholars to assert that phonetics and phonology are actually one and the same thing (e.g. Johnson 2007) and others still further that phonology plays no part in the grammar at all (Burton-Roberts 2011). Even those who maintain that there's a distinction between phonetics and phonology are in disagreement about where and how the distinction should be drawn. For

<sup>&</sup>lt;sup>3</sup> I should probably point out that when I say phonology here, I'm talking about spoken language phonology in particular (hence my reference to the sound system) despite the fact that it's my belief that the only difference between spoken language and signed language is that the former is produced by the mouth and perceived by the ears, and the latter is produced by the hands and perceived by the eyes, and so underlying, their phonological structures should be the same.

some (e.g. Archangeli and Pulleyblank 1994; Hayes et al. 2004) phonology is distinct from phonetics but is 'phonetically grounded', and for others (e.g. Blaho 2008; Hale and Reiss 2008; Iosad 2013, 2017) the symbols the phonological processor computes are actually abstract because phonology is independent of phonetics or 'substance-free', as it is so-called.

It is against the phonetically grounded/substance-free debate that this chapter has been written, as the substance-free approach to phonology has interesting things to say about Fodor's (1983) informationally encapsulated definition of modularity. A Fodorian modular phonology must be substance-free, and the same is true in the opposite direction: if phonology is phonetically-grounded, then it can't be modestly modular. It's no wonder that we haven't reached a clear conclusion about whether phonology is modular or not then, for many theories of phonology are still (incorrectly, I argue) grounded in phonetics.

Personally, I take a position on the substance-free side of the fence. This is, importantly, for reasons that *don't have to* do with modularity though, which will be discussed in due course along (roughly) the lines of Blaho (2008) and Iosad (2013, 2017). Spontaneous phonologisation (a phonological process described by Fruehwald in 2013 and 2016) also leads me to this line of thinking, and so this is explored also at the end of the chapter.

### 2. Substance-free phonology

Phonetically grounded phonologists (e.g. Archangeli and Pulleyblank (1994), and those who published papers in Hayes et al. (2004): i.e. Crosswhite (2001), Hayes and Steriade (2004), Wright (2004), Jun (2004), Kaun (2004), Blevins and Garrett (2004), Dresher and Zhang (2004), Flemming (2004) and Gordon (2004)), blur the distinction between phonetics and phonology in their claim that phonological processes are the result of articulatory and acoustic factors (Blaho 2008). They argue that articulatory and acoustic knowledge is encoded in phonology in the form of constraints, requiring phonological representations to be characterised in accordance with information important to production and perception (Blaho 2004).

It's easy to see why people may adopt an approach to phonology such as this one given the Jakobonian legacy of substantive markedness and the universal features described in classic generative phonology of course (Jakobson et al. 1952; Chomsky and Halle 1968). Even OT, which is a relatively recent phonological framework, assumes the phonetic grounding of some of its phonological constraints (Prince and Smolensky 1993). A number of other researchers,

however, are of the opposing position that phonology is actually autonomous of phonetics; that both phonological representations and the processes that compute them are devoid of phonetic influence (Dresher et al. 1994; Hale and Reiss 2000a, b; Hume 2003; Avery and Rice 2004; Blevins 2004; Blevins and Garrett 2004; Mielke 2004, 2005; Hale et al. 2007; Morén 2007a, b, c; Blaho 2008; Hale and Reiss 2008; Iosad 2013, 2017). This argument appears to have good grounds.

There are five schools of thought when it comes to substance-free phonology, and they're listed here as follows, from the most substance-free to the least substance-free:

- The Concordia school of thought (Hale and Reiss 2000a, b; 2003, 2008; Hale et al. 2007)
- The Toronto school of thought (Avery and Rice 1989; Rice and Avery 1991; Piggott 1992; Rice 1993; Dresher et al. 1994; Avery 1996; Dresher 2001, 2003)
- Element Theory (Harris 1990, 1994, 2005, 2006; Harris and Lindsey 1995)
- The Parallel Structures Model (Morén 2003a, b; 2006, Morén 2007a, b, c)
- Radically Substance-free Phonology (Odden 2006, 2013; Blaho 2008, Youssef 2010)

For a detailed overview of these many varying schools of thought, one could consult Blaho (2008), but for this thesis, it should suffice to say that there are indeed five, and that, although they differ in their degrees of substance-freeness, they are all alike in that they reject a one-to-one correspondence between phonetics and phonology.

In the sub-section that follows (section 2.1), I will offer some assumptions about substancefree phonology, before summarising some arguments that were offered in the theses of Blaho (2008) and Iosad (2013) to support the substance-free position (including gradual phonologisation) in sub-section 2.2. I'll then present a couple of analyses of /ay/ raising in Philadelphian English which I believe to evidence a substance-free phonology: Fruehwald's (2013) (in 2.3.1) who assumes that /ay/ raising in Philadelphia is due to spontaneous phonologisation, and the phonological analysis that I myself subscribe to, that of Bermudez-Otero (2017), who claims pre-fortis clipping is what causes /ay/ to raise (in 2.3.2). In section 3 some concluding remarks are made, mainly that there is plenty pointing to phonology being substance-free, and therefore an informationally encapsulated module of the mind.

### 2.1 Some assumptions about substance-free phonology

According to Blaho (2008: 2), the three major tenets of substance-free phonology are as follows:

- (1) Phonology refers to the *symbolic computational system* governing the *significant*, i.e., the non-meaningful level of linguistic competence. Phonology is taken to be *universal* common to all (natural human) languages and all modalities –, and *innate*.
  Phonological knowledge is part of [Universal Grammar(]UG[)], but phonetics is not.
- (2) Phonological primes are substance-free, in that their phonetic interpretation is invisible to phonology, and thus does not play a role in phonological computation.
- (3) Markedness and typological tendencies (in the sense of Greenberg (1957, 1978)) are not part of phonological competence, but rather an epiphenomenon of how extraphonological systems such as perception and articulation work.

A number of arguments have been made in favour of the above, and so I consider those next in section 2.2.

### 2.2 Some existing arguments for a substance-free phonology

## 2.2.1 Explanatory adequacy

An existing argument from Hale et al. (2007) (that was described by Blaho (2008)) for a substance-free phonology has to do with explanatory adequacy. As Hale et al. (2007: 662) point out, the set of attested languages (that is, the number of languages (living or dead) we have evidence to prove existed) is not equal to the set of statable languages (that is, the number of languages (that is, the number of languages (that is, the set of statable languages (that is, the number of languages (that is, the number of languages (that is, the number of languages that a model of phonology can predict). Rather, the relationship looks a little something like this (Blaho 2008: 7):

(1) attested  $\subset$  attestable  $\subset$  humanly computable  $\subset$  statable

Hale et al. (2007: 663) describe the pattern in the following way:

First, the set of attested languages is a subset of the set of attestable languages (where 'attestable' includes all linguistic systems which could develop diachronically from existing conditions-e.g., all dialects of English or Chinese or any other language in 400 years, or 4000 years, etc.). In addition, the set of attestable languages is a subset (those which can evolve from current conditions) of the set of humanly computable languages. (In our opinion, the human phonological computation system can compute a featural change operation such as  $p/ \rightarrow [a] / d$  but it is of vanishingly small probability that such a rule could arise from any plausible chain of diachronic changes.) Finally, the set of humanly computable languages is itself a subset of formally statable systems (which could include what we take to be humanly impossible linguistic processes such as  $/V \rightarrow [V:]$  in prime numbered syllables). The key point here is that the set of diachronically impossible human languages is not equivalent to the set of computationally impossible human languages.

The argument that follows from this view of language, Blaho (2008) reasons, is that 'it is preferable for a model of phonology to have as few assumptions as possible, *even at the expense of overgenerating*' (Blaho 2008: 8). For example, given three groups of sounds A, B and C, if there was evidence of languages with A and B, B and C, and A, B and C, and yet there wasn't any of languages with A and C, then most phonologists would be inclined to argue that UG has a rule prohibiting systems consisting of A and C, right? Well, according to Blaho (2008), not necessarily. We can't always assume that what we observe, has a phonological reason behind it, she stresses.

It might be that a pattern observed has an extra-phonological explanation having to do with production or perception, for a start. To take Blaho's (2008) trivial example, the fact that there have been no sounds produced with the larynx and lips as the articulators, for instance, needn't be encoded in the phonology as some sort of feature co-occurrence restriction, because it can already be explained by the anatomy of speech organs, in that the larynx and lips are simply too far apart for this to be a possible pattern.

Principles of UG of the type 'features can combine freely, except for A and B' or even worse 'all sounds are possible except for this one and that one because the articulators are too far apart' are therefore not necessary in Blaho's (2008) opinion, and nor are they in mine. As Blaho (2008) puts it: explanatory adequacy shouldn't be sacrificed for the sake of observational or descriptive adequacy. And so anything that can be explained outside of phonology should be.

### 2.2.2 Multiple modalities

One of the most obvious arguments for the autonomy of representations, though, to my mind, comes from the existence of languages that aren't spoken. A key property of grounded phonology, (Blaho 2008) points out, is that constraints refer to aspects of spoken language only, even though there are many more modalities than that. Linguistics has long established that signed languages are natural human languages in the way that spoken ones are (Stokoe 1960; Klima and Bellugi 1979; Senghas et al. 2004); there are a number of autonomous and mutually unintelligible systems of communication used in deaf communities around the world (MacSweeney et al. 2008). Now, to my mind, the only way in which spoken languages and signed languages differ is in their methods of production and perception surely (in that spoken languages are produced by the mouth and perceived by the ears, whereas signed languages are produced by the mouth and perceived by the ears, whereas signed languages are produced by the hands and perceived by the eyes) and so underlyingly, if phonological knowledge is as universal to humans as Chomsky and Halle (1968) say it is, then their linguistic structures should be the same.

As Blaho (2008) says, it is not easy to see how a phonetically grounded model of phonology could account for signed language, for this isn't a modality in which (oral/nasal) articulatory production and acoustic perception plays a part. If it is the case that the constraints of phonology *are* phonetically grounded, then UG must have at least two sets, surely: one for spoken language phonology, one for signed language phonology (and possibly others if it turns out that those modalities have different phonetics than do the spoken and signed ones). Per Blaho's (2008) (and indeed my own) understanding, the simplest explanation is that innate phonology is free of phonetic information concerning production and perception, and the mapping from underlying forms to their surface realisations is something that's acquired during language learning instead at an interface between phonology and phonetics. In other words, a substance-free phonology would be less of a cognitive load than a phonetically grounded one would, and therefore the one most likely to play a part in a human mind that is bound by the constraints of memory.

#### 2.2.3 Emergence

## 2.2.3.1 Emergent features

A fundamental question that's been asked by generative phonologists concerns how phonological patterns of the world's languages are characterised. In generative phonological theory, it's argued that universal (and therefore innate) building blocks of speech sounds called distinctive features are responsible for this – and these characterisations have to do with

63

phonetics (e.g. [voice] for which segments can be + or - depending on whether they're produced with the vocal chords vibrating or not, respectively, and [nasal] for which segments can be + if they are produced with the velum lowered which allows air to flow through both the nose and the mouth or - if there is velic closure, that is if they are produced with the velum raised allowing air to flow through the mouth only).

Segments (and natural classes of them) are typically interpreted in the generative phonology literature as bundles of these phonetically defined feature sets that match the ways in which they are articulated but, although the central idea behind distinctive feature theory (that contrasts between phonemes can more elegantly be described in terms of properties of segments as opposed to treating segments as alphabetic atoms) makes sense, in many of the features being defined by auditory and articulatory terms distinctive feature theory is clearly not supportive of the suggestion that phonology is substance-free.

Distinctive feature theory, which draws on the work of Saussure (1966), Trubetzkoy (1969) and Jakobson et al. (1952) dates back as far as Jakobson (1942) but received its most sophisticated defence at the hands of Chomsky and Halle in 1968 in *The Sound Pattern of English* (SPE). Chomsky and Halle (1968) recast the acoustically defined features as articulatory ones so that all distinctive features had to do with articulation. In more recent years weaker versions of distinctive feature theory (owing to Kiparsky (1985, 1995), Steriade (1987, 1995), Archangeli (1988), Archangeli and Pulleyblank (1994), Hale et al. (2007) and Hale and Reiss (2008)) have come about too, but in spite of their weakening of the others' theses, they wholeheartedly maintain that there exists a set of universal and innate features that specify segments (Iosad 2013).

More recently still, scholars have shown, however, that there isn't necessarily a *need* for features to be innate, for they are inducted over the input by the language learner during acquisition – that is, they are emergent from the ways in which the productive and perceptive systems work (Blaho 2008). In *The Emergence of Distinctive Features*, for example, Mielke (2008) set out to evaluate how well distinctive feature theories could account for the range of phonologically active classes in a cross-linguistic database. He compared the success of three theories (Jakobson et al.'s (1952) Preliminaries to Speech Analysis (PSA), Chomsky and Halle's (1968) SPE and Clements and Hume's (1995) Unified Feature Theory (UFT)) against a database consisting of 6,007 classes of sounds described in 628 language varieties, to see how phonologically active classes (that is, sets of segments that act as targets or triggers in

alternations) lined up with the classes that those featural theories predict. It turned out that 'no single theory [was] able to characterize more than 71 percent of the classes, and over 24 percent [were] not characterizable in any of the theories' (Mielke 2008: 3); the theory that fared the best was that of Chomsky and Halle (1968) – but even that was only able to cover 4,313 of the 6,077 classes, Mielke (2008) said.

Off the back of these results, Mielke (2008) argued that the segment classes predicted by phonological features in the literature aren't actually representative of those that are active extra-grammatically. Thus, a theory in which language learners observe categorical distributions in surface forms and map them onto underlying phonological categories is one that would explain the natural and unnatural class patterns found in the data.

To his mind, then, humans are not born with a limited set of universal features, they are born with the ability to create them using the data they're exposed to during the course of language acquisition. The same is said by Hale and Reiss (2000) about a phonological pattern that was originally depicted in terms of teleological constraints by Beckman (1997), and so I'll consider this in subsection 2.2.3.2.

# 2.2.3.2 Emergent patterns

According to Hale and Reiss (2000b), there is no need for UG to prescribe that which can be procured from the physiological. They argue that that pattern described by Beckman (1997) isn't one that is innately imposed but in actuality arises through emergence, and so I summarise what they (Beckman (1997) and Hale and Reiss (2000b), that is) had to say about that in what follows.<sup>4</sup>

In *Positional Faithfulness, Positional Neutralisation and Shona Vowel Harmony*, Beckman (1997: 1) argues that:

The distribution of the feature [high] in Shona verbs is a prototypical example of positional neutralisation accompanied by vowel harmony.[...] In languages which exhibit positional neutralisation of vowel contrasts, one or more vowels (generally, the most marked

<sup>&</sup>lt;sup>4</sup> Hale and Reiss (2000a, b) also reject that *r*-insertion in Massachusetts English (McCarthy 1993) and opacity effects in Hebrew spirantisation (1996) should appeal to teleological constraints, but given the limitations on space in this thesis, I've elected to explain the most illustrative of the examples only. For an overview of the other two, please feel free to refer back to the original text.

members of the vowel inventory) may occur distinctively in only a small subset of the structural positions available in the language. Outside of these positions, the marked vowels may surface only if they harmonise with a similar vowel in the privileged position.

The mid vowels /e/ and /o/ in Shona verbs then, she says, 'are contrastive only in root-initial syllables [and may appear] in subsequent syllables only when preceded by a mid vowel in root-initial position' (Beckman 1997: 1). 'Low vowels neither trigger nor propagate height harmony; only [+high] *i* and *u* may follow a low vowel, even if the root-initial vowel is mid' (Beckman 1997: 1). 'Harmony fails to apply between a root-initial mid front *e* and a subsequent round vowel'[, and t]hus, rather than a height-harmonic string of *e* ... *o*, we find disharmonic *e* ... *u*. Harmony applies regularly when the initial vowel is round *o*', meanwhile (Beckman 1997: 2).

Beckman (1997) goes on to present a full analysis of Shona height harmony, proposing that the phonological properties of initial and non-initial syllables (described above) arise due to a dispersion of a couple of IDENT constraints:

(1) IDENT- $\sigma_1(hi)$ 

A segment in the root-initial syllable in the output and its correspondent in the input must have identical values for the feature [high].

(2) IDENT(hi)

Correspondent segments in output and input have identical values for the feature [high].

(1) and (2) together allow faithfulness to the feature [high] to be maintained in some contexts but not others, she says, as the context-sensitive constraint IDENT- $\sigma_1$ (hi) in (1) ranks above \*HIGH (a markedness constraint that is violated by the presence of high vowels), which itself ranks above the general constraint IDENT(hi) in (2):

(3) IDENT- $\sigma_1(hi) >> *HIGH >> IDENT(hi)$ 

More specifically, Beckman (1997: 8) argues that in Shona:

the ranking of *IDENT*- $\sigma_1(hi)$  above the vowel-height markedness constraints (see below) permits the full range of height contrasts to

occur in initial syllables, and further renders these syllables impervious to height harmony. By contrast, the ranking of the contextfree constraint *IDENT(hi)* below the markedness constraints renders noninitial syllables incapable of licensing marked vowels and further, susceptible to height harmony.

Hale and Reiss (2000b), on the other hand, argue that many markedness patterns are actually emergent – including the one explored by Beckman (1997). Consider the alternative account they offer (Hale and Reiss 2000b: 159-160):

Imagine a child exposed to a language  $L_1$  that allows high vowels in all syllables—initial, medial, and final. Imagine further that  $L_1$  has initial stress and that stress is realized as relatively increased duration and intensity. Given this scenario, it would not be surprising to find that a child constructing  $L_2$  on the basis of output from  $L_1$  consistently fails to acquire a contrast between mid and high vowels in relatively short, quiet syllables (those that are noninitial and thus unstressed), but succeeds in acquiring it in initial syllables, which are stressed and thus longer and louder. This mapping from  $L_1$  to  $L_2$  is an example of "sound change"—in particular, [...] a "conditioned merger"[.]

On the other hand, it is highly implausible that a child would consistently fail to correctly analyze the mid/high contrast in longer, louder (stressed) syllables, yet successfully analyze the contrast in relatively short, quiet syllables.

We see therefore that the existence of positional faithfulness phenomena can be understood as merely reflecting the nature of the learning situation[ and is therefore not a reflection of any grammatical principle].

If the acoustic cues of a given contrast in the target language are correctly analysed by the acquirer in a context where they are relatively weak, they will also be analysed correctly in a context where they are relatively strong.

They suggest instead that the markedness pattern is an emergent property, rather than an innate one. "Positional faithfulness" is due' Hale and Reiss (2000b: 160) claim, 'not to the nature of phonology [itself], but to the "sifting effect" of acquisition on the incidental, arbitrary nature of the phonetic substance associated with phonological symbols'.

Beckman's claim that positional faithfulness constraints should be encoded in the grammar is therefore rejected by Hale and Reiss (2000b) as, for them, if something can be explained by something external to phonology (as is the case here, because 'those [effects] observed by Beckman already have a coherent extragrammatical account within acquisition theory' (Hale and Reiss 200b: 160)), then it need not and should not be fixed into it in the form of teleological constraints. There is no reason why UG should stipulate that faithfulness in less prominent positions is outranked by faithfulness in more prominent positions if the same pattern emerges through perception, they say. '[B]uilding positional faithfulness into a theory of universal phonology is a misuse, or abuse, of phonetic substance in theory construction' according to Hale and Reiss (200b: 160), and Blaho (2008: 4) agrees. As she puts it: 'it would be superfluous to duplicate this 'knowledge' and build it into our model of phonology'.

The conclusion Hale and Reiss (2000b: 162) draw from the above example (and their reanalyses of *r*-insertion in Massachusetts English (McCarthy 1993) and opacity effects in Hebrew spirantisation (McCarthy 1996), for that matter) is that:

the best way to gain an understanding of the computational system of phonology is to assume that the substance of phonological entities is *never* relevant to how they are treated by the computational system, except in *arbitrary*, *stipulative ways*. What this means is that many of the so-called *phonological universals* (often discussed over the rubric of markedness) are in fact epiphenomena deriving from the interaction of extragrammatical factors like acoustic salience and the nature of language change. [...] We propose extending the Saussurean notion of the arbitrary nature of linguistic signs to the treatment of phonological representations by the phonological computational system. Phonology is not and should not be grounded in phonetics since the facts that phonetic grounding is meant to explain can be derived without reference to *phonology*.

I think it's safe to say that Hale and Reiss' (2000b) argument for a phonology that is substance-free (due to their demonstration that the markedness tendency seen in Shona verbs can be explained on phonetic grounds) is a convincing one – duplicating what's going on outside of the grammar inside it is a waste of time and space, surely, and so would be avoided by a computational system at all costs.

Recall Carruthers' (2006a) argument from computational tractability in Chapter 3. The mind, including the grammar and the component parts of it, are computationally realised, he says, and all computational processes must be suitably tractable so that they can be carried out within finite time (Carruthers 2006a: 44):

- (1) The mind is computationally realized.
- (2) All computational processes must be suitably tractable.

In computational phonology, the field that approaches the study of sound patterns from a computational perspective, problems are classified into two types: tractable problems and intractable problems (Heinz 2011a, b). According to Heinz (2011a: 9):

Problems are tractable iff there is a algorithm which implements this function in fewer than f(n) steps where f is a polynomial function and n is the length of the input (this input is the problem instance in some string-based representation). Decidable problems which cannot be so bounded are intractable.

He also stresses, though, that the variables of *restrictiveness* and *complexity* affect computational tractability as well. The more restrictive a computational system, he says, the less complex it is. And the less complex a computational system, the more tractable it is.

Now, OT has been criticised by some (e.g. Idsardi 2006; Vaux 2008) for its computational complexity as it is, Iosad (2013: 47) points out. Idsardi (2006) and Vaux (2008) have each criticised OT for being too complex, and therefore not tractable enough. Of course, the smaller the set of constraints that exist in phonology, the more restrictive the phonological computational system would be. The more restrictive it would be, the less complex it would be, and the less complex it would be, the more tractable would be.

I'm inclined to argue, then, that if it is the case that phonology is a computational system, and it's imperative that computational systems are tractable, then it wouldn't express patterns with phonological constraints that are already expressed by phonetics, because that isn't being very restrictive at all. A computational phonology would not make more work for itself by handling something grammatically that's already being handled extra-grammatically if it's aim was to be tractable, for that would be the opposite of restrictive (or as Carruthers 2006a puts it, frugal) – that would be extravagant, lavish, wasteful. The fewer constraints a phonological computational system has the better. For the more constraints it has, the less restrictive  $\rightarrow$  more complex  $\rightarrow$  less tractable it is.

John Ohala has similarly argued that markedness tendencies (for similar reasons) should be explained phonetically as opposed to phonologically. For an extensive exploration, one should see Ohala (1998). Various work on learnability owing to Boersma et al. (2003), Escudero and Boersma (2003), Boersma (2006, 2007), Apoussidou (2006) and Boersma and Hamann (2007) strongly supports Hale and Reiss' (2000) claim too; their studies showed that rankings can be derived from phonetic data during language learning as well (Blaho 2008).

69

### 2.2.4 Diachrony

A fourth and final existing argument for substance-free phonology to be considered in this thesis concerns diachrony. Although diachronic change is not (usually) assumed to play a part in synchronic linguistic competence, there is no denying the fact that it is related to the synchronic system's functioning in some way, as Iosad (2013) says, and so I consider it here. Classical generative phonologists (Chomsky and Halle 1968), diachronic generative phonologists (Kiparsky 1965, 1968; King 1969), natural phonologists (Stampe 1972; Donegan and Stampe 1979), natural generative phonologists (Vennemann 1972a, b, 1974; Hooper 1976), phonologists interested in variation and sound change in progress (Labov 1971; Labov et al. 1972), those concerned with phonetic explanations of phonetic patterning and sound change (Ohala 1974, 1981; Thurgood and Javkin 1975; Hombert et al. 1979), and researchers working on intrinsic and extrinsic variations in speech (Wang and Fillmore 1961; Chen 1970; Mohr 1971) have been at great debate about the relation between synchrony/diachrony and phonetics/phonology, and for many (dating as far back as Baudouin de Courtenay (1972), actually) 'the original cause of the emergence of alternants is always purely anthropophonics' Baudouin de Courtenay (1972: 184); they believe that phonological structure is derived from phonetic substance.

Hyman (1972, 1976), a phonologist who considers himself to sit somewhere in between generative phonology and the structural phonology of the Prague school of thought (e.g. Trubetzkoy 1939; Martinet 1960) is concerned, in particular, with a process he calls phonologisation, which refers to the change of a phonetic property into a phonological one. It is thought by some phonologists (e.g. Iosad 2013) that the theory of phonologisation counts as evidence for a substance-free phonology, and I can certainly see why.

Spanning the years of 1972 to 1976, Hyman gave a number of definitions for phonologisation, including, but not limited to, these ones:

A universal phonetic tendency is said to become 'phonologized' when language-specific reference must be made to it, as in a phonological rule (Hyman 1972: 170).

*phonologization*, whereby a phonetic process becomes phonological [...] (Hyman 1975: 171).

[...] what begins as an intrinsic byproduct of something, predicted by universal phonetic principles, ends up unpredictable, and hence, extrinsic. (Hyman 1976: 408).

But what *exactly* would phonologisation look like, one might wonder? In *Phonological Involvement in Phonetic Change*, Fruehwald (2013) paints a picture.

## 2.2.5 Gradual phonologisation

According to Fruehwald (2013), two things have to do with phonologisation: a phonetic process called coarticulation and a phonological one called differentiation. To exemplify these processes (and how they play a part in phonologisation), he brings evidence from a hypothetical sound change to a vowel /V/, which appears in two totally different segmental contexts /\_\_\_ x/ and /\_\_ y/. I summarise his explanation in the exposition that follows.

Fruehwald (2013) presents the distributions of [Vx] (in red) and [Vy] (in blue) in F1 x F2 vowel space as in Figure 4.1, where F1 stands for the first formant of vowel space (which is inversely related to vowel height) and F2 the second formant (which has to do with the degree of backness of a vowel), according to Ladefoged (2006). A formant, Ladefoged (2006) says, is the concentration of acoustic energy that surrounds a particular frequency in a speech wave. Each formant corresponds to a certain resonance in the vocal tract, and we can distinguish one vowel from another by the differences they have between these overtones (e.g. we could distinguish /i/ and /u/ by their F2 values (backness) seeing as /i/ is a high front vowel and /u/ a high back one, and /i/ from /æ/ in terms of their F1 values (height) given that /i/ is a high front, and /æ/ a low front, vowel).



Figure 4.1 Distribution of contextual variants ([Vx] and [Vy]) of a hypothetical vowel (Freuhwald 2013: 61)

As you can see, Figure 4.1 clearly demonstrates the difference in [Vx] and [Vy]'s articulation – [Vy] is produced further back than [Vx]. However, there are a couple of ways in which [Vx] and [Vy] could have been generated, Fruehwald explains. On the one hand, it could have been through phonological differentiation in that /y/ could have spread some feature *f* (say, [+ back]) onto /V/, creating a featurally distinct and therefore phonologically different allophone of /V/:

(1) 
$$V \rightarrow V_f / \_ y$$

In this case, [V] (in red) and  $[V_f]$  (in blue) would have independent targets for phonetic implementation (as is depicted in Figure 4.2):



Figure 4.2 Independent targets of phonetic implementation produced by phonological differentiation (Fruehwald 2013: 62)

On the other hand, however, it might have been the case that phonology wasn't involved at all and rather, that there was only one mapping to the phonetic target [Vx] (from its underlying representation /V/), and [Vy] was produced due to a coarticulatory pressure being placed on [Vx] by [y], pulling its production back. In this case, [Vx] and [Vy] don't have independent targets for phonetic implementation:



Figure 4.3 The effect of coarticulation on shifting productions from intended targets (Fruehwald 2013: 63)

Now, it should be clear from Fruehwald's (2013) example, at this point, why phonetic coarticulation could be reanalysed as phonological differentiation – and that's what phonologisation is. Before they undergo diachronic change, you see, the ways in which [V] and [Vx], and [V<sub>f</sub>] and [Vy] are realised in F1 x F2 vowel space look more or less the same. When [V] and [V<sub>f</sub>] and [Vx] and [Vy] change diachronically, however, a difference can be observed.

Firstly, in the case of phonological differentiation [V] and  $[V_f]$  have independent targets of phonetic implementation, so it is possible for them to have separate diachronic trajectories as in Figure 4.4. In the case of phonetic coarticulation, however, the realisation of [Vy] is yoked to that of [Vx]. What this means, is that the *diachronic trajectory* of [Vy] is yoked to that of [Vx] too, and so the same thing has to happen to each of them (as in Figure 4.5):

Compare the two:



Figure 4.4 The interaction of phonological feature spreading and diachronic phonetic change (Fruehwald 2013: 64)



Figure 4.5 The interaction of phonetic coarticulation and diachronic phonetic change (Fruehwald 2013: 64)

Figure 4.4, Fruehwald (2013) explains, illustrates the interaction between phonological feature spreading and diachronic change. The data represents an independent shift of [V] frontwards along F2;  $[V_f]$  does not move. In Figure 4.5, which illustrates phonetic coarticulation's interaction with diachronic phonetic change, [Vx] shifts frontwards along F2, and so too does [Vy] given that it's yoked to [Vx], it is explained by Fruehwald (2013). This is interesting because phonologisation is, in fact, the reanalysis of a difference between two segments being due to phonetic coarticulation as being due to a phonological process with featurally distinct objects, Fruehwald (2013) says and, through diachronic change data, we can pinpoint precisely when that phonologisation takes place.

The effects of the reanalysis of phonetic coarticulation as phonological differentiation are illustrated by Fruehwald (2013: 68) with the following figure:



Figure 4.6 The reanalysis of phonetic coarticulation as phonological feature spreading, and its effect on the rate of phonetic change

In the top graph in Figure 4.6, you can see that [Vx] and [Vy] undergo the same change until roughly 1930, which means that up until that point the difference between the two must have been due to phonetic coarticulation (the trajectory of [Vy] was yoked to [Vx], which meant that they each underwent the same change). At around the year of 1930, however, you can see that the former ([Vx], now [V]) continues on along its previous path being affected by the change, whereas the latter ([Vy], now  $[V_f]$  does its own thing because affected it is no longer. This is because, at this point, the phonetic effect was reanalysed as being phonological, and so the process entered the grammar creating two featurally distinct allophones [V] and  $[V_f]$ which had independent targets of phonetic implementation, and therefore independent trajectories of change.

For representative examples of real phonologisation at play one might want to read Barnes' (2006) work on vowel neutralisation or Kingston's (2007) on tonogenesis – but for now, it should suffice to say that phonologisation could also count as evidence for a substance-free phonology, for the difference between phonological and phonetic patterns establishes the existence of two different domains of grammar (Iosad 2013).

#### 2.3 Alternative arguments for a substance-free phonology

## 2.3.1 Spontaneous phonologisation

Now, with all that about gradual phonologisation being said in section 2.2.5, I should probably point out that if gradual phonologisation could count as evidence for a substancefree phonology, then Fruehwald's (2013, 2016) argument *against* gradual phonologisation in favour of one *for* spontaneous phonologisation in Philadelphian English could count as evidence that's even better. If his analysis of the phonological pattern is proven to be correct, then it is even more challenging to the view that there isn't a qualitative difference between phonetic and phonologisation are. In his dataset, Fruehwald (2013, 2016) found evidence to suggest that the phonological processes which differentiate allophones enter the grammar at the very onset of sound changes, rather than partway through the change as late-stage reanalyses. This means that they cannot simply be the codifications of phonetic effects, he says, because, at the outset of the changes, there were no phonetic effects to be codified.

Canadian /ay/ raising – wherein the diphthong /ay/ surfaces as /Ay/ before voiceless consonants has been a topic of linguistic interest since the 1940s. A number of researchers have written about raising thereafter, in Canadian varieties of English and other ones. It was this that was the focus of Chapter 5 of Fruehwald's doctoral dissertation which was published in 2013, and also a later paper *The Early Influence of Phonology on a Phonetic Change* which appeared in *Language* three years later in 2016. He analyses /ay/ raising in longitudinal data extracted from the Philadelphia Neighborhood Corpus (Labov and Rosenfelder 2011) which spans the speech of people born between 1889 and 1998, and demonstrates that in the data all /ay/ diphthongs before underlyingly voiceless consonants pattern together, regardless of whether or not their trigger (the proceeding plosive) is voiced on the surface or not.

Fruehwald (2013) found, more specifically, that pre-voiceless /ay/ raising applied before flapped /t/ and /d/ in the same way it did before un-flapped /t/ and /d/, which suggests, he said, that the neogrammarian phonetic change targeted an underlying *phonological* category as opposed to a surface phonetic one. Fruehwald's (2013) findings are clearly really relevant to questions about the contested relationship between phonetics and phonology, then, and so it's important that they're looked at.

Fruehwald (2013: 111) begins his argument by providing the following ordered rule analysis, to show where and how /ay/ raising occurs in Philadelphian English:

	Writer	rider
Input	ıaitəı	ı.ebiaı
RAISING	Ivitəl	-
FLAPPING	LEJIAL	Levier
Output	IeJIVI	ıenia

Table 4.1 Opaque interaction between /ay/ raising and flapping

As you'll see from the table above, /ay/ raising applies opaquely with respect to flapping, Fruehwald (2013, 2016) argues. Whilst words like *write* which have voiceless codas are produced with the raised variant of the vowel /Ay/, words like *ride* which have voiced ones are produced with the variant that is not raised (i.e. [ay]). /ay/ raising also occurs before flapped /t/, but not flapped /d/. So a word like *writer* in which the /t/ is flapped will surface with the higher diphthong, whereas for a word like *rider* with a flapped /d/, /ay/ raising will not apply.

It therefore looks like raising is phonological, he says. That it is triggered before a flapped /t/ which is voiced on the surface, only because the /t/ is voiceless underlyingly. He argues that it is not how the trigger is realised phonetically that is relevant to this sound change, but rather, the voicing of the underlying representation.

Some phonetic arguments have been made for /ay/ raising, however, (e.g. those of Joos (1942) and Chambers (1973) and Moreton and Thomas (2007)), and so Fruehwald (2013) carefully considers those explanations, in order to see whether the way in which /ay/ patterns in the Philadelphian data set can be attributed to phonetic pressures after all. Pre-voiceless vowel shortening is the phonetic precursor for pre-voiceless /ay/ raising that's most commonly argued for in the literature, and so he began with that one.

According to Joos (1942) and Chambers (1973), /ay/ diphthongs are long gestures that are shortened before voiceless segments and so in order to shorten the diphthong, they argue, a speaker may well raise the nucleus of it to reduce gesture length. To test Joos' (1942) and Chambers' (1973) hypotheses that /ay/ raising is due to duration, Fruehwald (2013) assessed the median duration (in msecs) of /ay/ before flapped /t/ and surface /t/ and flapped /d/ and surface /d/ after putting them in order of shortest to longest (as in Table 4.2):

Segment	Context	Median duration (msecs)
/t/	Flapping	111
/t/	Surface	144
/d/	Flapping	156
/d/	Surface	237

Table 4.2 Median /ay/ durations by context (adapted from Fruehwald 2013: 117)

As the table shows, /ay/ before flapped /t/, surface /t/ and flapped /d/ form a set of shorter distributions, and /ay/ before surface /d/ forms another, longer one. If Joos (1942) and Chambers (1973) were right in what they were saying, that /ay/ raising is phonetically conditioned by diphthong duration, then this would predict flapped /t/, surface /t/ and flapped /d/ all participate in raising, Fruehwald (2013) points out. However, since /ay/ raising only occurred before flapped and surface /t/s in the Philadelphian data, it is evident that their thesis is incorrect, he says.

A second phonetic conditioning hypothesis comes from Moreton and Thomas (2007). They, like Fruehwald (2013), make an argument against Joos (1942) and Chambers' (1973) prevoiceless shortening account, pointing out that in dialects in which /ay/ is monophthongised, the monophthongisation is actually *least* advanced before voiceless segments, meaning it's at its longest in this context and therefore at its least likely to raise as a result of being shortened. They hypothesise alternatively that the glide is peripheralised in pre-voiceless contexts, a hypothesis which cleverly captures, Fruehwald (2013: 116) says, 'both the coarticulatory pressure to raise the nucleus towards the glide, and the resistance to monophthongization'.

Fruehwald (2013) considers this phonetic conditioning explanation for /ay/ raising too, but since /ay/ doesn't raise before /r/ in Philadelphia, he has to do so with data derived from Rosenfelder (2005), who studied /ay/ raising in Victoria, British Columbia. The ways in which surface /t/ and /d/ and flapped /t/ in Rosenfelder's (2005) data are given in Figure 4.7 below:



Figure 4.7 Nucleus to glide trajectories in Victoria, British Columbia (Fruehwald 2013: 199)

Now, whilst it's quite clear from the figure that glide targets are more peripheralised before the voiceless obstruents than the voiceless one to a degree that isn't in proportion to the heights of the nuclei they proceed (which would seem to support the glide peripheralisation hypothesis) in Victoria, British Columbia, it also predicts something quite different from what we see in the Philadelphian data, Fruehwald (2013) figures.

If flapped /d/ was less peripheralised than surface /d/ (in the same way that flapped /t/ is less peripheralised than surface /t/ and as the peripheralisation hypothesis would predict) and flapped /d/, therefore, had a more similar glide target to surface /d/ and flapped /t/ than surface /t/, Fruehwald (2013) points out, then the glide peripheralisation hypothesis would predict that only surface /t/ (the most peripheralised of all) would undergo raising. We can see from Figure 4.8 though that this is not the case – raising applies to both variants of /t/ (Fruehwald 2013: 121):



Figure 4.8 /ay/ height by date of birth and context

The trajectories in Figure 4.8 (which plots /ay/ height by date of birth and contexts (before surface and flapped /t/ and surface and flapped /d/), as you can see, match neither the predictions of the duration precursor hypothesis (which predicts that flapped /t/, surface /t/ and flapped /d/ all participate in raising) nor the glide peripheralisation hypothesis (that only surface /t/ does), then, for n this dataset 'the height of /ay/ appears to pattern according to the underlying phonological voicing of the following segment, with {surface /t/, flapped /t/} patterning together, and {surface /d/, flapped /d/} patterning together', with surface and flapped /t/ undergoing raising, and surface and flapped /d/ not undergoing it (Fruehwald 2013: 120).

The fact that /ay/ raising applies to underlyingly voiceless /t/ and /c/, despite their surface realisations, Fruehwald (2013) argues, points to this being a phonologically conditioned change, not a phonetically conditioned one. This is, of course, a qualitative conclusion, however, and so he ran his data through Stan (a probabilistic programming software) to see whether this argument was one that could be supported by statistics.

Fruehwald (2013) plotted the estimated F1 trajectories for /ay/ in each context faceted by from (surface or flapped) together with 95% highest density posterior (HPD) intervals (i.e. the coloured bands in the graph) which specified a probability of 95% that the true value for each context lay within the coloured band associated with it. The potential scale reduction factor  $(\hat{R})$  is less than 1 for this dataset Fruehwald (2013) says, which suggests that the model adequately depicts the data being analysed:



Figure 4.9 Model estimates of /ay/ height, faceted by surface vs. flapped realisations

Unlike the hypothetical example of gradual phonologisation in which [Vx] and [Vy] underwent a change together due to a phonetic effect until around the time of 1930, at which point the change was reanalysed as being phonological, and two featurally distinct allophones [V] and  $[V_f]$  were formed and underwent change separately, phonologisation was observed at the outset with ay-raising, Fruehwald (2013) argues.

You can indeed see quite clearly from Figure 4.9 that surface /t/ and flapped /t/, and surface /d/ and flapped /d/ have always had independent trajectories. And, in Figure 4.10, which plots the same trajectories but faceted by the underlying stop to see whether it is this, specifically, that affects naming, you can see why Fruehwald came to this conclusion even more clearly. /ay/ before /t/ (as both a surface form, and as the underlying form of flap)'s estimated normalised F1 value lowers over time (indicating that it raises in height), whereas /d/ (in both its surface and flapped underlying forms) does not:



Figure 4.10 Model estimates of /ay/ height, faceted by /t/ vs. /d/

The fact that /ay/ before /t/ and /ay/ before /d/ follow different trajectories from the get-go, suggests, Fruehwald (2013) says, that the change didn't bring to bear on phonetics during the change at all, and so phonological factors have greater explanatory power than phonetic ones in the case of /ay/ raising. He suggests that the phonological differentiation of /ay/ as high or low is due to a grammatical process that entered the phonology before the change came about. These results, of course, pose a challenge for the commonly accepted view that phonologisation (the process by which a phonetic pattern becomes a phonological one, recall) occurs gradually and, as such, has important implications for theories of phonology – including the one of interest to this thesis that is the substance-free framework.

Fruehwald's (2013) account of /ay/ raising is clearly incompatible with theses in which the boundary between phonetics and phonology is blurred like phonetically grounded phonology ones, because the change, to his mind at least, had nothing to do with phonetics. It is not incompatible with the substance-free phonology thesis though of course, as according to this one phonological knowledge is not simply the codification of phonetic patterns, but rather something more abstract. As such, I think it's safe to say that Fruehwald's (2013) work can count as strong evidence against the former, and for the latter.

Note though that most of this discussion thus far has been about what Fruehwald (2013) has said *didn't* happen to /ay/. Evidence from his 2013 dissertation demonstrated that /ay/ raising was likely neither (a) phonetically conditioned by diphthong duration as Joos (1942) and Chambers (1973) hypothesise, and nor was it (b) phonetically conditioned by glide peripheralization (which was what Moreton and Thomas's (2007)) offered as an explanation. Rather, Fruehwald (2013) argues, /ay/ raising is a *phonologically* conditioned change.

He explains that if two variants of a speech sound are to diverge in their phonetics over time, then that must be because they are treated as quantitatively different categories (that is, as different allophones) by speakers from the moment that they begin to diverge. He argues that the split of /ay/ into two allophonic categories is *not* the reanalysis of a phonetic change (as would be the case if this was phonetic coarticulation being reanalysed as phonological differentiation). Rather, that the longer term phonetic change is only possible precisely *because* /ay/ split into two allophones before the phonetic change began. This split is what allowed for their phonetic targets to diverge phonetically according to the voicing specification of the following segment, flapped /t/ or flapped /d/, he says, with one

82

undergoing a gradual (phonetic) change in height (before flapped /t/) and the other remaining low (before flapped /d/).

If this is the case though, then the next question that arises is this one: how did those two categorically different variants of /ay/ come to exist in the first place to be conditioned by flapping? An answer is provided by Bermudez-Otero (2017) and so I discuss that here.

## 2.3.2 The enhancement-of-clipping hypothesis

An alternative (although related) phonological analysis of the pattern (that I'm more inclined to concur with because the analysis can explain /ay/ raising in a number of other varieties of English including Canadian and Scottish<sup>5</sup>) is the one offered by Bermudez-Otero (2017). Although different from Fruehwald's (2013, 2016), this argument is still consistent with a modular architecture of the mind in which phonology free of phonetic substance in that there are actually *two* different types of process involved in /ay/ raising – pre-fortis clipping and flapping which are phonological, and the raising of the vowel itself, which is phonetic.

Pre-fortis clipping, Bermudez-Otero (2017) argues, is a categorical phonological process that applies at the *stem-level* of phonology, and therefore underapplies when the voiceless segment that conditions it belongs to a suffix at the word-level. Since flapping occurs across word boundaries (e.g. 'sit in the park' 'si[r] in the park' (Kaisse and Shaw 1985)), this must be a post-word level process (Kiparsky 1985; Kaisse and Shaw 1985) he says, occurring at the *phrase-level* of phonology. He argues furthermore, that since phrase-level processes occur after stem-level ones, flapping applies after clipping.

<sup>&</sup>lt;sup>5</sup> The analysis offered by Bermudez-Otero explains not just /ay/ raising in Philadelphian English, but /ay/ raising in Ontario, Canada (see Chambers (1973, 1989, 2006); Chambers and Hardwick (1986); Thomas (1991) Rosenfelder (2007) and in Scotland (see Aitken (1981), Agutter (1988), McMahon (1991)) – for a more nuanced discussion of how prefortis clipping affects /ay/ raising in these varieties of English, one should consult Bermudez-Otero (2017).



Figure 4.11 The stem-level, word-level and phrase-levels of phonology (adapted from Bermudez-Otero and Trousdale 2012: 700)

Raising, Bermudez-Otero (2017) argues, is a phonetic process that applies opaquely with respect to *clipping* (not flapping as Fruehwald (2013, 2016) believes to be the case). It is the clipped allophone only which undergoes the /ay/ raising change under his proposal, and therefore the rule for raising is:

**not** as  $\rightarrow$   $\Lambda i / [-voice]$ **but**  $\breve{a}$   $\rightarrow$   $\breve{\Lambda}i$ 

As is exemplified by the following set of data:

		ri <b>d</b> er	wri <b>t</b> er	idle	ti <b>t</b> le
Stem level	(clipping)	Jaid	Jăit	aɪdəl	tăıtəl
Word level		reprer	ığıtər	aɪdəl	tăıtəl
Phrase level	(flapping)	lenier.	remer	aırəl	tăırəl
Phonetics	(raising)	lenier.	ιeniăr	aırəl	tăirəl

Table 4.3 Derivations in early 20th-century Philadelphia (adapted from Bermudez-Otero 2017: 8)

Now, whilst this weakens Fruehwald's (2013) argument for this sound change being 'phonologisation' that occurs spontaneously as opposed to gradually, it importantly *preserves* his core argument that a phonetic change operated over surface phonological representations. Bermudez-Othero (2017) proposes that a pre-existing phonological process (pre-fortis clipping) created two different allophones of /ay/, [ai] and [ăi], before /ay/ began to raise phonetically. The phonetic change that resulted in the raising of pre-voiceless /ay/, he says, targeted the clipped allophone ([ăi]) only.

This analysis makes more sense, to my mind, given that pre-fortis clipping and /ay/-raising are observed elsewhere in identical environments – consider, for example, the distribution of /ay/ raising in Scottish English (Bermudez-Otero 2017: 13-14):

In Scottish English, /ai/, among other vowels, is...

- clipped before all consonants other than voiced continuants,
   e.g. sign, side, life, sight
- unclipped elsewhere
  - e.g. sigh, dive

If it's the case that raising targets only clipped tokens of /aɪ/, it would follow that in Scottish English, raising would only target tokens of /aɪ/ that have been clipped by the Scottish Vowel Length Rule (SVLR) (and would do so even when those tokens are followed by a voiced consonant).

Evidence for this has indeed been found by (Scobbie et al. 1990: 241):

•	[Ăi]	clipped by the SVLR	and so raised:	sign, side, life, sight
•	[aɪ]	unclipped by the SVLR	and so unraised:	lie, alive

Bermudez-Otero (2017) then, like Fruehwald (2013, 2016), importantly draws a distinction between phonological processes (clipping and flapping, in this instance) and a phonetic process (that is, raising) in his account of the diachronic change in Philadelphian English. This analysis (unlike analyses of the change which attribute it to phonetic conditioning) is, of course, consistent with a modular feedforward architecture of the grammar in which phonology precedes phonetics. This analysis importantly demonstrates a difference between changes that occur in a phonological module (in which clipping and flapping apply) and a phonetic one (where raising applies). This is supportive of the substance-free framework of phonology, which separates phonology from phonetic substance.

### 3. Concluding remarks

As you can see, there is a wealth of evidence pointing towards phonology being substancefree. In section 2.2 I explored some existing arguments for a substance-free phonology, including explanatory adequacy in 2.2.1, multiple modalities in 2.2.2, emergence in 2.2.3, diachrony in 2.2.4 and gradual phonologisation in 2.2.5.

After that, I examined the extent to which Fruehwald's (2013) account of /ay/ raising as being due to spontaneous phonologisation, not gradual phonologisation, could count as even stronger evidence for a substance-free phonology and concluded that, if correct, it could. Since phonetic change for the variants within that vowel category were best explained in terms of phonological allophony than in phonetic disposition, it is easier to see how this could be the case if phonology wasn't grounded in phonetics than if it was. The Philadelphian data was suggestive that phonology can bring about change without making any reference at all to phonetics, and more specifically that the divergent pattern of change is best attributed to categorical allophones that were created by the phonology at the outset of it, which is of course what a substance-free phonology thesis would predict.

However, as I point out in the later subsections of this Chapter I myself actually subscribe to an alternative analysis of the pattern, the one owing to Bermudez-Otero (2017), which although different from Fruehwald's (2013) is still supportive of a substance-free view of phonology. Bermudez-Otero (2017) also argues for separate phonological (clipping and flapping) and phonetic (raising) processes in his analysis, which is very much in alignment with a substance-free framework of phonology which separates phonological processes from phonetic ones, with phonological processes occurring inside the phonology, and phonetic processes outside of it.

# **Chapter 5: A Massively Modular Approach to Phonology**

### **1. Introduction**

The chapter examines the extent to which phonology can be considered domain specific and therefore a module of the mind by massive modularists' (in particular Carruthers 2006a's) standards. To do so it considers the effects semantically- and phonologically-targeted treatments have on phonological anomia (a disorder of naming that has to do with damage to the phonological level of language processing) under the rationale that if it is the case that the phonological processor is a domain specific module of the mind, it must be able to operate only on input relevant to its function, that is, phonological information. If semantically-targeted treatments were found to facilitate word retrieval in people with phonological anomia, then this could count as evidence that the phonological processor is anomia, then this could count as evidence that the phonological processor is anomia, then this could count as evidence that the phonological processor is anomia, then this could count as evidence that the phonological processor is anomia, then this could count as evidence that the phonological processor is anomial (i.e. input that is *irrelevant* to its function), I argue.

An understanding of the complex cognitive system that's involved in spoken word production is a prerequisite for a proper understanding of the disorders of naming that are observed in aphasia, Raymer (2005) rightly points out, as they are actually caused by damage that disrupts that system's activity. Many models of spoken word productions' details have been subject to great debate, but in spite of this, there is a general agreement among researchers that there are two stages (semantic and phonological) involved in the word retrieval process (Caramazza et al. 2000; Herbert 2004), though, as I point out in section 2.4.1 (2.4.1.1 specifically) syntax is involved as well. Researchers also generally agree that subsequent to these stages exist postlexical processes in which speech production is planned (Raymer 2005).

The current cognitive approach to the assessment of and intervention into naming disorders in aphasia locates language breakdown in normal models of language processing at one, some or all of these levels, and so the chapter will begin with a detailed discussion of those models. I also present my preferred model (one that isn't currently referenced in the clinical literature, that of Jackendoff (2002)). This model, in my opinion, better explains how and where syntax plays a part, where the lexicon (and the lexical entries) that make it up is stored, and how modules are connected to and communicate with one another. There are number of problems with the models of word retrieval referenced by clinicians, and Jackendoff's (2002) solves them all, I point out. I also, importantly, explain how word retrieval can be handled by

Jackendoff's (2002) model of the language faculty. It is not just a model of competence, but of performance too, I point out. And in my eyes, is the most superior of all argued for so far.

I begin by outlining leading theories and areas of agreement and disagreement among the theorists each model is due to, before assessing the adequacy of their abilities to account for the word retrieval errors we see in normal (in section 2) and aphasic (in section 3) patients. In section 4 though (which is the largest one of all) I turn my attention to anomia therapy, and both qualitatively and quantitatively consider what the results of 43 studies conducted between the years of 1980 and 2002 have to say about the domain specificity and modularity of phonology, which is that it is neither domain specific nor modular. My results show that phonology can operate on semantic information, in that semantic therapy is able to treat phonological anomia. Section 5 concludes.

### 2. Models of spoken word production

Cognitive models of word retrieval paint a picture of how the processes involved in it are organised, and as a result, are clinically relevant in that they provide a framework for the formulation of testable hypotheses about language behaviour (Laine and Martin 2006). Data that has been used to test these hypotheses include the speech errors that are produced by normal speakers (Fromkin 1971) and people with aphasia (Dell et al. 1997), as well as analyses of tip of the tongue states in normals and aphasics (Brown and McNeill 1966), as section 2.2.1.1 and (briefly) 2.2.4.1 explore. The evidence from each of these sources has made a significant contribution to the development of cognitive models; without speech error and tip of the tongue data, we wouldn't know anywhere near as much about spoken word production as we do today.

Current cognitive models of word retrieval have their origins in the neuroanatomical models of language breakdown in aphasia that were proposed by 19<sup>th</sup> century neurologists (Laine and Martin 2006). In 1825, for example, Jean-Baptiste Bouillon distinguished between two types of language production impairments: one that affected ones' underlying knowledge of words, and one that affected the planning and execution of the speech movements that were required to produce them (Laine and Martin 2006). A similar distinction was drawn by Jacques Lordat in 1843, and by 1865 the third frontal convolution had been identified by Broca as being the site of expressive language (i.e. Broca's area). By 1874 Wernicke had linked language comprehension to the left superior temporal lobe's posterior two-thirds (i.e. Wernicke's area) (Laine and Martin 2006).

Broca's (1865) and Wernicke's (1874) investigations into neuroanatomical sites of language acted as the foundations on which the more detailed models of the mental operations that underlie language production were built. In fact, it was Wernicke's (1874) work together with that of Lichtheim (1885) that led to the first fully-fledged model of the language system: the Wernicke-Lichtheim model (which was, rather self-explanatorily, named after the pair). The Wernicke-Lichtheim model had interesting applications to aphasiology for two reasons: on the one hand, because it identified the component parts of the language processor (centres for auditory images (A), motor images (M) and concept elaboration (C) as well as auditory perceptual (a) and motor encoding (m)), and because it described different profiles of aphasia in terms of disruption (see the slashes with numbers) to those components or to the spaces between them, they said (Laine and Martin 2006).

A lesion at site (A) in Wernicke-Lichteim's model, which is Wernicke's region, would result in Wernicke's aphasia whereas if there were a lesion at site (M) (which is Broca's region), Broca's aphasia would result. This is because A, is where auditory word images are processed (Laine and Martin 2006) (the hence comprehension problems) and Wernicke's aphasia), and (M) where motor images are processed (Laine and Martin 2006) (hence Broca's aphasia which causes problems with production:



Figure 5.1 A depiction of the Wernicke-Lichtheim model of language processing and the different aphasia profiles accounted by the model (Laine and Martin 2006: 5)

Now, although the Wernicke-Lichtheim model isn't what we would consider particularly descriptive today it was compared to other models in its time, and it was enough so that its assumptions led to the creation of the classical taxonomies of aphasic syndromes that were used in the assessment of and intervention into aphasia in the 60s and 70s (Geschwind 1965; Benson 1979).

Wernicke and Lichtheim's brain-behaviour approach (that is, their belief that language components have neuroanatomical correlates) was being cared for less and less by the 1980s, however. At his point researchers were beginning to develop more functional, as opposed to structural, approaches to the modelling of the language system (Morton and Patterson 1980; Dell 1986; Kay and Ellis 1987; Howard and Franklin 1988). Spoken word retrieval was mostly modelled, by this point, by 'box and arrow diagrams' (e.g. Morton (1970, 1979), Butterworth (1989) (and even Levelt et al. 1999)) which owed their instantiations to analogies with computers. They made use of boxes to illustrate stores of representations (e.g. semantic/phonological/morphological) and arrows the processes between them (e.g. which took the output of one representational store's route as input to the next) (Herbert 2004; Laine and Martin 2006). These functional, box and arrow models have proved useful clinically

speaking as diagnostic frameworks in the field of aphasiology (as we'll soon see), for their detailed descriptions of the language processor's components provided clinicians with clear profiles of their patients' spared and impaired processes that enabled them to more precisely pinpoint the language impairments that were underlying the patients' language disorders, and what aspects of language in particular (e.g. semantic/phonological) should be targeted to treat them (Laine and Martin 2006). The functional approach to modelling also provided a vocabulary with which to describe disorders of language in terms of behaviourist phenomena. This is just as important an advance now as it was back then (even if interest in behavioural-neuroanatomical correlates has been brought back by innovations in neuroimaging (Laine and Martin 2006).

A renewed interest in brain imaging studies has led some to question how specified these 'box and arrow' models actually are, and *connectionist* computational models have been developed (through applications of the advances in the computer simulation technology of mental phenomena) to take their place (e.g. Shattuck-Hufnagel 1979; Stemberger 1985; Dell 1986, 1989). You see, whereas box and arrow models group representations together in one or more boxes, connectionist models specify even further than that, representing individual items as nodes (Herbert 2004). These nodes are connected both within *and between* hierarchical levels; and although some still assume that language is processed temporally in one level after the other, others argue that processing can occur in one level at the same time that processing is occurring in others (Herbert 2004). I explore each of those types of models in section 2.2.

However, the relationship between models of spoken word production and the diagnosis and treatment of language disorders is, of course, a co-dependent one. Whilst models of language production can be used to diagnose and treat disorders of language, systematic case studies of language disorders can be (and indeed have been) used to test how well models of language production can account from them. Aphasic speech error data has been used in particular for this – but so too has the speech error data of normal speakers and data from tip of the tongue states in both normal and aphasic speakers as I said earlier – and so this section and the next one will devote discussion to that.

I begin this section with a description of how words are thought to be represented in and retrieved from the mental lexicon by Morton (1970, 1979) in 2.1.1, before pointing out the problems with that approach. I draw on speech error and tip of the tongue data in order to do so, and I have drawn a distinction between Morton's (1970, 1979) model (which just has one

step) and those of others (which are made up of two steps) in section 2.2.

The first two-step model I consider (in 2.2.2) is that of Butterworth's (1989) semantic lexicon model (which is similar to Morton's (1970, 1979) model in many ways, except that it incorporates an explicit level of lexical-semantic processing (the semantic lexicon) which the logogen model does not). I then map out Levelt et al.'s (1999) WEAVER++ model (in 2.2.3) before contrasting the pair of feed-forward activation models (Butterworth (1989) and Levelt et al.'s (1999), that is) with Stemberger (1985) and Dell's (1986) interactive activation models in 2.2.5. In section 2.3 I discuss the process by which the phonological information that is retrieved from stored lexical representations is converted into a form that can be used by the articulators for speech, and in doing so cover two well-known slot-and-filler models Shattuck-Hufnagel (1979) and Dell's (1989) interactive accounts of phonological encoding which describe how segments are slotted into syllabic frames in sections 2.3.1 and 2.3.2, respectively.

Some concluding remarks are then made about the adequacy of these models. I point out the questions these models leave unanswered, and how a model like Jackendoff's (2002) is able provide what we're looking for. In particular, I criticise the current models for not explaining the part syntax plays in language processing, and how connections are made and communication is handled between the semantic, syntactic and phonological (what Jackendoff (2002) calls *integrative*) processors. I also highlight how one could and should view the phonology-phonetics/phonetics-phonology *interface* module(s), as well as where the lexicon is and what it does, in particular how it operates in tandem with the rest of the (*integrative and interface*) modules. A more detailed discussion of anomia begins in section 3.

### 2.1 One-step models

One of the most representative examples of one-step models of spoken word production (and therefore the only one I'll consider in this chapter) is that of Morton (1970, 1979).

### 2.1.1 Morton's (1970, 1979) logogen model

In Morton's (1970, 1979) logogen model lexical items are represented as logogens, which are said to be counting devices that are put in place 'whenever there is an input of an attribute that matches one of those attributes specified within the defining set for that logogen' (Nickels 1997: 11). In speech production these logogens accumulate conceptual-semantic information

92

from the cognitive system (which is depicted at the top of the diagram) and deliver it to be expressed by the phonological output lexicon (i.e. the first, rectangular box in Figure 5.2), he claims:



Figure 5.2 The logogen model of spoken word production (Nickels 1997: 11)

Each logogen is assigned a threshold, and when the count of the accumulated conceptualinformation exceeds that threshold the logogen that was assigned it 'fires', sending a phonological code to the response buffer (i.e. the second, rectangular box) which results in the production of speech.

The single step involved in this process (taking information from the cognitive system and turning it into speech) is what the term 'one-step' in one-step models refers to; despite there being two boxes (the phonological output lexicon and the response buffer) there is only one (phonological) process necessary to get from the cognitive system to speech. Now, when one-step models are used in clinical work, many researchers believe that the diagnosis and treatment of aphasia being offered by clinicians using them is unsatisfactory, for it relies on an inadequate interpretation of the problem at hand. To their minds, there are actually two levels of representation (and therefore, two steps that need to be taken in order to retrieve words). As such, many two-step models of spoken word production have been proposed in more recent years (as section 2.2 explores).
### 2.2 Two-step models

Two-step models postulate two levels of language processing: a semantic one and a phonological one (with semantic representations of lexical items being accessed at the semantic level of language processing and their respective phonological representations from the phonological level of language processing at which instructions for realisations phonetically are given). Evidence for the existence of these levels has been found in the speech of both normal and aphasic speakers. I'll cover evidence from normal speakers in section 2.2.1.1.1 of this chapter, and evidence from aphasic speakers in 2.2.1.1.2.

### 2.2.1 Semantic and phonological levels

### 2.2.1.1 Evidence for independent semantic and phonological levels

### 2.2.1.1.1 Evidence from normal speakers

Speech errors or 'slips of the tongue' are by no means an infrequent occurrence in the everyday speech of normal speakers (Nickels 1997). They are not thought to occur randomly, though. Rather, systematically so, and so over the years, they've been extensively exploited for the insights they can provide about the production of spoken words (Nickels 1997).

### 2.2.1.1.1.1 Speech errors

Shattuck-Hufnagel (1979, 1987), Stemberger (1985), (Dell 1986, 1989) and (Butterworth 1989), for example, use speech error data as support for the theoretical models of spoken word production they've proposed in recent years that argue for two levels of representation in lexical access as opposed to just the single level of representation that's argued for in Morton's (1970, 1979) one-step logogen theory. The argument for two levels of lexical representation (one organised by meaning and the other organised by sound), they say, is based on the independence of semantics and phonology in errors (as evidenced by targets that are substituted with semantically related but not phonologically related words and those that are substituted with phonologically related words but not those that are related semantically) such as those in the following examples that were put forward by Fromkin (1971: 46):

(a)	Semantically related:	I like to - hate to get up in the morning		
		the oral - written part of the exam		
(b)	Phonologically related:	bottle (target: bottom) of page five		
		while the present - pressure indicates		

Within a model that has just a single level of representation, semantic errors such as *like* for *hate* and *oral* for *written* can be easily accounted for by the logogen model, Nickels (1997: 16) argues:

As the lexicon (comprising output logogens) accumulates semantic information from the cognitive system, a range of semantically related items will be activated at one time and random noise within the system (e.g. temporary lowering of a threshold due to recent firing of a logogen) may result in a non target semantically related to the logogen firing.

But it is less of an easy feat trying to explain the occurrence of phonologically related real word errors (e.g. *bottle* for *bottom* and *present* for *pressure*) within Morton's (1970, 1979) model, for logogens are said to accumulate conceptual-*semantic* information, and so processing problems will result in errors that are semantically related only, she points out Nickels (1997). Once the logogen delivers the conceptual-semantic information from the cognitive system to the phonological output lexicon to be expressed errors might well occur in encoding, Nickels (1997) says, but this is what we call a phonemic process as opposed to a phonological one (I'll come back to this point properly in section 3) and so the result will be random phonemic errors that result in non-words (or real-words but just by chance) as opposed to errors which involve the switching out of the target word for something that is phonologically similar (Nickels 1997).

### 2.2.1.1.1.2 Tip of the tongue states

Models with single levels of representations and therefore 'all or nothing' retrieval of lexical items also have difficulty in accounting for tip of the tongue states in normal speakers. In a tip of the tongue state a speaker 'knows' the word that they want to produce, but can't access its form from memory. They might be able to evidence their knowledge of the word by saying something about it (e.g. that it begins with this or that sound or that it is similar in meaning to this word or that one), but when it comes to accessing its information in its entirety they're - for lack of a better word or phrase here – stumped. Tip of the tongue states can shed further light then on the inner workings of the speech production system in that they capture it in a state of interruption, as opposed to impairment, either because of something we call blocking or something else we call incomplete activation (Herbert 2004) (which is a point I'll come back to in due course).

Brown and McNeill (1966) were among the first to study the kind and quantity of information subjects have available to them in tip of the tongue states. They induced these states in subjects by reading them the definitions of low frequency English words (e.g. 'a navigational instrument used in measuring angular distances, especially the altitude of the sun, moon, and stars at sea' for

sextant) and asking them to identify what it was that they were describing.

Subjects who didn't recall the target word in this particular example immediately but felt that they knew the word was in there somewhere provided two kinds of information to Brown and McNeill (1966) while they searched for it: semantic and phonological. They were also asked to make Brown and McNeill aware of any other words that came to mind in their attempt to access the target one and these could be divided into two types as well: those that were similar in meaning (e.g. *astrolabe*, *compass*, *divider*, *protractor*) and those that were similar in sound (e.g. *secant*, *sextet* and *sexton*).

Brown and McNeill (1966) pointed out that the nearer subjects were to the successful recall of the target words, the more accurate the information was that they possessed. This insight led Brown (1991) to formulate two hypotheses about the cause of tip of the tongue states (recall): on the one hand, he said that they could be due to blocking, and on the other that they could be due to incomplete activation. The blocking hypothesis suggests that similar sounding words to the target are activated as well as the target and in turn obstruct its retrieval while the incomplete activation hypothesis suggests that tip of the tongue states arise when a target fails to reach the threshold level of activation that's necessary for its retrieval (Brown 1991).

A series of experiments designed to test the two hypotheses owing to Brown (1991) was performed by Jones (1989) and Jones and Langford (1987). The studies used the same methodology as did Brown and McNeill (1966), but modified it slightly to accompany each definition they provided to subjects with a phonologically related, semantically related, both phonologically and semantically related or both phonologically and semantically unrelated prime (e.g. when presented with the definition 'medieval forerunner of chemistry' to induce a tip of the tongue state for the target word *alchemy*, the phonologically related prime *axial* was presented as well). Jones (1989) found tip of the tongue states to be more common when the target and prime were phonologically related or both phonologically and semantically related than semantically related or both phonologically and semantically unrelated, which led them to conclude that the results were more consistent with Brown's (1991) blocking hypothesis than they were with his incomplete activation one.

There were a number of problems with this study, however, in particular in that the prime stimuli differed in difficulty (read: some were high frequency words and some were low ones) and subjects weren't provided with their definitions so if their meanings weren't already known the

efficacy of the semantic primes couldn't be properly measured. Due to this, Meyer and Bock (1992) carried out two further experiments following Jones (1989) and Jones and Langford's (1987) methodologies (but made sure to present primes' definitions as well to avoid differences in difficulty contributing to any faciliatory effects found).

Results revealed that words were retrieved more often when their definitions were followed by phonologically, semantically or phonologically and semantically related words than they were when they were followed by phonologically and semantically unrelated ones. They didn't find any evidence to suggest that the provision of semantically or phonologically related information reduced target words' accessibility either (which the blocking hypothesis would of course predict) and so Meyer and Bock (1992) suggested that the partial activation hypothesis looked more likely than the blocking one for this set of data.

When considering the kinds of responses produced by participants in tip of the tongue states the logogen model fails once again then, Nickels (1997) stresses. Just as was the case with speech error data, although the logogen model can easily explain the production of words that are semantically related to the target it can't account for phonologically related responses (Nickels 1997). Furthermore, the availability of partial phonological knowledge in tip of the tongue states is completely incompatible with a threshold model like the logogen one, for in the logogen model (remember) the logogen either reaches an activation threshold at which point the full phonological form, and no information is retrieved (Nickels 1997). There is no point at which, in the logogen model, partial information is made available, and so those circumstances described in tip of the tongue states cannot be adequately accounted for by it (Nickels 1997).

### 2.2.1.1.2 Evidence from aphasic speakers

Two cases from the aphasiology literature (JCU (from Howard and Orchard-Lisle 1984) and EST (from Kay and Ellis 1987)) have been cited over and over again to support the assertion that there are two independent semantic and phonological levels of language processing (e.g. Lesser 1989) as well. Whilst JCU made semantic errors but not phonological ones in their speech, EST made phonological errors but not semantic, which researchers have taken to be indicative that there are two processing levels – one semantic and one phonological – that can be impaired independently of each other. I come back to cover these case studies (among a number of others) in more detail in section 3 (sections 3.1 and 3.2, to be exact), but for now, in section 2, I think it should suffice to say that they are thought to have deficits in language processing at two different levels: JCU a

deficit at the semantic one (hence the semantic but not phonological errors) and EST a deficit at the phonological one (hence the phonological errors, but not semantic).

Models that incorporate two levels of lexical representation can easily account for speech errors and the availability of information in tip of the tongue states in normals and aphasics however, and so I provide a summary of some of the most widely discussed models made up of two levels of representation here: Butterworth's (1989) semantic lexicon model (in section 2.2.2) and Levelt et al.'s (1999) WEAVER ++ model (in 2.2.3), as well as Stemberger's (1985) and Dell's (1986) interactive activation models in section 2.2.5.1. In 2.3.1 and 2.3.2, respectively, I discuss Shattuck-Hufnagel (1979) and Dell's (1989) interactive activation accounts of phonological encoding, before arguing for a model that looks more along the lines of Jackendoff's (2002) one and turning my attention to anomia and anomia therapy in sections 3 and 4.

### 2.2.2 Butterworth's (1989) semantic lexicon model

In 1989, Butterworth developed a model with two levels of lexical representation (lexicalsemantics and a phonological lexicon) at which lexical access occurs temporally. First, lexicalsemantics (which is a transcoding device that takes as input a semantic code from the cognitive system and outputs a phonological address) is accessed. After that, the phonological output lexicon is (which is another transcoding device that takes the phonological address (recall, the output from lexical-semantics) and outputs the phonological form). Access is strictly top-down for Butterworth (1989), (which remember means information flows downward only and can't be fed back from the lower levels up to the higher ones (except if they're the results of checking procedures (hence the two-way arrows in Figure 5.3), but that's not really relevant to this thesis)). You can see this in the following figure:



Figure 5.3 The proposed stages of spoken word production: Butterworth's (1989) semantic lexicon model (from Nickels 1997: 21)

Butterworth's (1989) semantic lexicon is made up of pairs of feature sets that associate semantic properties (those that comprise the criterion for the semantic search for a word in lexical-semantics, that is) with phonological ones in the phonological output lexicon (e.g. such as the number of syllables a word has, which of these are stressed and the segments that make up the syllables' onsets, nuclei and codas). Within the phonological output lexicon there exists pairs of addresses (output from the semantic lexicon) and strings of phonemes for the formation of words, and so unlike Morton's (1970, 1979) model of spoken word production, Butterworth's (1989) can easily explain the independence of semantically related and phonologically related spontaneous speech errors, it's pointed out.

Semantically related errors may occur for one of two reasons, Butterworth (1989) argues: (a) because of an error in generating a semantic search criterion, or (b) due to an error in matching a correctly generated search criterion with its partner representation in lexical-semantics (Nickels 1997). Phonological errors, by contrast, occur for just one reason only, when there is a failure to match a correctly retrieved (from lexical-semantics) phonological address with its partner in the phonological output lexicon and a near neighbour, instead of the target, is activated by accident (Nickels 1997).

Once again, unlike Morton's (1970, 1979) model, Butterworth's (1989) can give us a straightforward explanation of how partial phonological information is available to subjects in tip of the tongue states (Nickels 1997). Having accurately accessed a phonological address from the semantic lexicon, speakers might well be unable to retrieve the item that's at that address in the phonological output lexicon (Nickels 1997). If this is the case, the speaker might at that point produce no response, or use the address itself (that is, the one they retrieved from the semantic lexicon) to provide information about the target (that's in the phonological output lexicon) (Nickels 1997).

Addresses bear a relationship to the things that reside there in that similar addresses point to similar sounding words. The speaker could, therefore, retrieve a phonological neighbour (a word that phonologically resembles the target) or they could make use of the information that's in the phonological address to make 'guesses' at the target (relating in phonologically related words and nonwords) (Nickels 1997), e.g. if looking for the target word *anemone* a speaker might well say *enemy* (if the phonological address points to a similar sounding word), *anometer* (an otherwise phonologically similar real word) or *anenome* (a nonword) if they make use of the phonological information initial phoneme [æ], and four syllables but switch out the placement of the phonemes [n] and [m]. They might also even just report the phonological information they're aware of itself instead of trying to do anything with it (e.g. the word I'm looking for begins with 'a' and has four 'beats'); this is something we see all the time, in aphasics especially.

### 2.2.3 Levelt et al.'s (1999) WEAVER++ model

Now, whilst researchers are in general in agreement that there exist independent semantic and phonological levels of representation in spoken word production, there has been disagreement as to whether there exists or doesn't exist an intermediate level of representation (known as the lemma level) between them. Kempen and Huijbers (1983) were the first to propose the lemma level in models of spoken word production. The lemma, they said, is a semantically and syntactically specified representation that mediates between conceptual-semantic and phonological representations, and so a level depicting its processing belongs between them. Levelt et al. (1999) argue, however, that the lemma mediates between not conceptual-semantic but *lexical*-semantic representations and phonological representations, and that it is at the lemma level that information about an item's syntax is made available. Now, whether the lemma mediates between conceptual-semantic representations or lexical-semantic representations and phonological representations to both Kempen and Huijbers (1983) and Levelt et al. (1999) this much, at least, is clear: there are two stages to lexicalisation

(one from the semantic to the syntactic, and one from the syntactic to the phonological).

According to Levelt et al. (1999), there are three major steps that underlie speaking (conceptualisation, formulation and articulation). After perceiving an object-form (for argument's sake, let's say an apple), the concept is identified by the person perceiving it, and a message (which tells the formulation processor what that is) is sent by the conceptualisation processor. The formulation processor would then take the messages and access the appropriate word (i.e. the noun *apple*) in the mental lexicon to express it. It'd then encode syntactic and morphophonological structures for the result, and the result would be a phonetic plan (i.e. an articulatory program) for the utterance (Roelofs 2000). Finally, the articulators would execute the articulatory program, and the result of that would be speech (Roelofs 2000), as is exemplified by Roelofs (2000) in Figure 5.4:





Lexical access for Levelt et al. (1999), then, consists of lemma retrieval and word-form encoding

processes which are part of the formulation stages of syntactic encoding and morphophonological encoding, respectively (Roelofs 2000). In lemma retrieval, lexical concepts are used to retrieve lemmas (which are, again, representations of the syntactic properties of words) from memory, and lemma retrieval makes these properties available for syntactic encoding processes that contain slots for the specification of abstract morpho-syntactic parameters (such as mood, tense, person and number) (Roelofs 2000). In word-form encoding, on the other hand, lemmas and their parameter values are used to recover the appropriate morpho-phonological properties from memory (Roelofs 2000). The purpose of this is to construct phonetic plans (Roelofs 2000).

### 2.2.4 Evidence that the lemma (syntax) is independent of phonology

A number of studies have sought to prove the independence of syntactic (i.e. lemma level) information from phonological (i.e. lexeme level) information to evidence that word production processes take two steps. If a lemma level exists, syntax will be made available before phonology, people say, and evidence from tip of the tongue studies (described in 2.2.4.1) and neuropsychological studies (2.2.4.2) have suggested that this might indeed be the case.

### 2.2.4.1 Tip of the tongue studies

Evidence to support the independence of syntax and phonology has been found in tip of the tongue studies involving normal subjects (Caramazza and Miozzo 1997; Vigliocco et al. 1997). In both Caramazza and Miozzo's (1997) and Vigliocco et al.'s (1997) studies subjects were provided with definitions of words which elicited tip of the tongue states and asked questions about the items that they weren't able to name to gauge their knowledge of the words' syntactic (e.g. grammatical gender) and phonological (e.g. the first and last sound and the number of syllables) properties. The studies showed that subjects in a tip of the tongue state were able to report the grammatical gender of words they were unable to name, whether their phonological properties were available or not. The results have of course been interpreted as support for the existence of an intermediate level of language processing between the semantic and the phonological at which syntactic information is available, for syntactic features were known to the speakers even when the full phonological form wasn't.

### 2.2.4.2 Neuropsychological studies

Further evidence to support the independence of syntax and phonology has been found in neuropsychological studies (e.g. Henaff Gonon et al. 1989; Badecker et al. 1995; Vigliocco et al. 1999). In the earliest (Henaff Gonon et al.'s (1989)) study the case of a French anomic subject, GM was reported. GM presented with fluent speech, but marked word finding problems were evident. While he was able to correctly identify the grammatical gender of the words of the 13/14 test items he was unable to find, though, identification of information relevant to the words' phonological forms was less accurate, which suggested to Hennaff Gonon et al. (1989) that syntactic and phonological information were made available to the patient at two different points in time.

Badecker et al. (1995), meanwhile, report the case of Dante, a patient who presented with anomia as a result of a meningoencephalitis. According to Badecker et al. (1995) when presented with pictures he'd been unable to name in a picture naming experiment, Dante was able to identify the grammatical gender of 106 of the 111 items, but was unable to provide any phonological information about the words at all (not their first sound, not their last sound and not their length). The authors interpreted the results of the study as evidence in support of a two-step level of spoken word production: Dante had access to the lemma level (at which syntactic information was made available), but access to phonology after that was absent altogether.

More recent research (owing to Vigliocco et al. 1999) came to the same conclusion. Vigliocco et al. (1999) described the case of MS who was able to identify whether items were count or mass nouns despite not being able to find the items' phonological forms to produce them during a picture naming task. They too attributed their data to syntactic and phonological information being represented at separate, syntactic and phonological levels.

### 2.2.5 Activation within two-step models

Not all models that have two levels of language processing have two levels of lexical access, however, and as Nickels (1997) says, it's important we maintain that distinction. Activation can be one of two things: feed-forward or interactive. As its name suggests, in the former activation moves level by level throughout the system, whereas in the latter activation can move forward and backwards. In models with discrete activation processing, there is, therefore, no temporal overlap between stages, but in models with interactive activation processing, there is.

Both one-step models (e.g. Morton's (1970, 1979) logogen model) and some two-step models (e.g. Butterworth 1989, 1992; Levelt 1992; Levelt et al. 1999) assume discrete activation processing, that is that processing has to finish at one level before it can start at the next. Activation in these models feeds forward, one stage at a time, with the output from one level being the input to the level that follows it (Morgart 2015). Dell's (1986) two-step model (and subsequent versions of it (e.g. Dell et al. 1997; Dell 1988, 1989)), on the other hand, differ

from Butterworth's (1989, 1992), Levelt's (1992) and Levelt et al.'s (1999) two-step models, for in Dell's (1986, 1988, 1989) and Dell et al.'s (1997) models temporal overlap in processing between adjacent stages is assumed. This, interactive activation, that is, is something that's also assumed by Stemberger (1985) and Shattuck-Hufnagel (1979), as we'll see in sections 2.2.5.1 and 2.3.1, respectively.

### 2.2.5.1 Stemberger (1985) and Dell's (1986) interactive activation models

Both Stemberger (1985) and Dell (1986) have developed models in which there are two levels of representation that can be likened to those in the models of Butterworth (1989) and Levelt et al. (1999), but, as I've said above, whereas in Butterworth's (1989) and Levelt et al.'s (1999) models activation occurs at each level one after the other, the mechanism by which lexical items are accessed in Stemberger (1985) and Dell's (1986) models involves interactive activation.

Stemberger (1985) and Dell's (1986) models have units at a lexical level (called that in Stemberger's (1985) but the word/morpheme model in Dell's (1986)) which are similar to lemma levels in that they correspond to spoken words and a phoneme level (called that in both Stemberger's and Dell's models) at which phonemes are supposedly stored. Dell's (1986) model has three other levels (a syllable level and a rime level above the phoneme level and a feature level below it), but he omits these from later descriptions and depictions of the model (such as Dell (1989)), and so I won't say too much about them here.







Figure 5.6 Dell's (1986) interactive activation model (adapted from Dell 1986). Word/morpheme level nodes are initially activated from semantic representations that aren't in the diagram. At the phoneme level on = onset, nu = nucleus and co = coda

In both Stemberger (1985) and Dell's (1986) models activated units (or nodes) send some of their activation to every other unit/node they are connected to. More highly activated nodes have larger effects on other nodes, of course, whereas less highly activated nodes have smaller effects. Differences in activation at one level are reflected at the other levels, and activation from lower levels feeds back to higher levels of the system (e.g. activated phoneme nodes will spread activation to words that are made up of those particular phonemes, meaning that non-target words can become partially activated if they are phonemically similar to (that is, are made up of similar phonemes to) the target).

These interactive two-step models can, therefore, account for the semantic and phonological errors produced in tip of the tongue states in a similar way to Butterworth's (1989) non-interactive two-step model, then. Conceptual-semantic information activates units at the lexical level (in Stemberger (1985)) and the word/morpheme level (in Dell (1986)) that represent that information, and errors arise because of the systematic spread of activation to non-target units and them being mistaken for targets as a result. This can also be the case at the phoneme level; consider how easy it would be for *feather* to be mistaken for *leather* in Figure 5.5, which phonologically differs from the target in its first phoneme only.

Now, although Stemberger (1985) and Dell's (1986) models have phoneme levels and Dell's (1986) has feature, rime and syllable levels), what they don't do is specify how phonological

encoding is accomplished. Separate models of phonological encoding have been proposed, however (by Shattuck-Hufnagel (1979) and Dell (1989)), and so I'll consider them next.

### 2.3 Phonological encoding

In the previous sections, I've discussed some of the mechanisms in models that can lead to the retrieval of a word's phonological form. After that, though, 'the phonological form has to be translated into an articulatory programme for controlling the speech musculature' (Nickels 1997: 45). This process is what we call phonological encoding.

There are a number of models of phonological encoding out there, and almost all of them incorporate some sort of 'slot-and-filler' mechanism by which the sounds of words (the fillers) have to be inserted into their position in the word frame (that is, their slot). It is this process that is the source of many phonological speech errors, Nickels (1997) points out, and so this section begins by classifying the speech errors we see that are phonologically related to their targets. It moves on, then, to a detailed description of the factors that have even found to constrain their occurrence (an account that closely follows that of Shattuck-Hufnagel (1979), which is the paper that motivated the making of her model). I go on to describe Shattuck-Hufnagel's (1979) model of phonological encoding after that and discuss, as she does, how the various error types might arise in it. Next, I make a move to describe the ways in which Dell's (1986, 1989) interactive activation model is similar to Shattuck-Hufnagel's (1979) slot-and-filler model, and how, like Shattuck-Hufnagel's (1979), it can account for phonological encoding errors.

## 2.3.1 Shattuck-Hufnagel's (1979) interactive activation account of phonological encoding 2.3.1.1 How speech errors are classified

In her 1979 publication *Speech Errors as Evidence for a Serial-order Mechanism in Sentence Production* Shattuck-Hufnagel made use of the MITCU (Massachusetts Institute of Technology Cornell University) corpus of speech errors to develop a scan-copier model of phonological encoding. Out of the 6000 speech errors collected over six years in the corpus Shattuck-Hufnagel (1979) found five types (substitution, exchange, shift, addition and omission). She exemplifies these on page 299 as follows (Shattuck-Hufnagel 1979: 299):

- 1. Substitution: A target segment is replaced by an intrusion segment, which may or may not have an apparent source within the utterance, e.g.
  - (a) It's a shallower <u>test</u> chest, but broad
  - (b) Anymay, I think (anyway)

- 2. Exchange: Two target segments change places in the target sequence, each serving as the other's intrusion segment, e.g.
  - (a) e<u>m</u>e<u>n</u>y (enemy)
  - (b) It's <u>past fassing</u> fast passing by
- 3. Shift: A target segment disappears from its appropriate location and appears at another location in the target sequence, e.g.
  - (a) State-<u>l</u>owned-<u>and</u> owned-land (state-owned-land)
  - (b) in a b\_ack blo black box
- 4. Addition: An extra segment is added to the target sequence; this intrusion may or may not have an apparent source within the utterance, e.g.
  - (a) either the plublicity would be bad (publicity)
  - (b) they bring ab<u>r</u>out about a
- 5. Omission: A target segment is dropped from the target sequence' there may or may not be a similar sequence elsewhere in the utterance, e.g.
  - (a) the d<u>ug</u> -the drugs
  - (b) piano sonata \_umber ten (number)

Shattuck-Hufnagel (1979) also went on to categorise these five types of speech error along the dimension of something she called the direction of influence (i.e. wherein the sentence the source of error was). When the source was later in the sentence, Shattuck-Hufnagel (1979) classified the error as anticipatory (e.g. the addition in 4(a)), and when the source was earlier she classified the error as being perseveratory (e.g. the addition in 4(b)). Like the additions in 4(a) and (b) omissions and substitutions can be either anticipatory or perseveratory, Shattuck-Hufnagel (1979) said, though it isn't always the case that additions, omissions or substitutions have to have an identifiable source, she stresses (and then exemplifies in 1(a) and (b) and 5 (a) and (b)). Shifts can likewise be anticipatory or perseveratory, she says, but unlike additions, omissions and substitutions, these errors do indeed need to have an identifiable source. Exchanges, lastly, were classified as being neither anticipatory nor perseveratory. This is of course because in exchanges there is more than one source and so the direction of influence goes both ways. Take the *emeny* for *enemy* example from (2a), for instance. In this, both the /n/ and the /m/ have a part to play and can be described as the source, in that they switch places.

### 2.3.1.2 How speech errors arise

In light of these errors, Shattuck-Hufnagel (1979) developed a model of three parts which has been described, super succinctly, by Nickels (1997: 53):

- a dual representation consisting of serially ordered slots and an equal number of independently represented target segments, at least at the word level and at the sound level; [and]
- a scan-copier that selects the appropriate segment from the set of two monitors: a checkoff monitor, which marks or deletes segments as they are copied into their target slots, and an error monitor, which detects and deletes or otherwise edits error-like sequences in a planned utterance.

At the word level, ordered frameworks of word slots are generated and corresponding sets of target items retrieved from the lexicon to be stored (short-term) in a buffer. The scan-copier scans the target sets in the buffer for the ones that belong to the first of the ordered slots, and once those morphemes are found, they're copied into their positions. At this point, there is a check-off monitor that either marks the used morphemes as used (or deletes them from their sets if they're no longer needed) and the scan-copier moves onto the next slot, to repeat the process until all slots have been filled.

At the phoneme level, the same process is found but this time with phonemes as opposed to morphemes ones and ordered frameworks of, of course, phoneme-sized slots rather than ones that are morpheme-sized. Words are entered into a short-term storage buffer as they are retrieved, once again, with basic sets (such as initial, medial and final phonemes) lined up to be slot into their positions in the syllables of the target lexical items that are derived from rules of syllable structure and stress patterns. The scan-copier scans the buffer for what it needs each time (e.g. it scans the initial/final consonantal sets for the onsets/codas and medial sets for vowel phonemes for nuclei (or indeed for liquid or glide phonemes which can also act as nuclei)).

According to Shattuck-Hufnagel (1979), each of the five errors described above can be accounted for as being due to some sort of problem with this segment selection process. These too have been helpfully summarised by Nickels (1997: 54) as follows:

1. Exchanges arise from misselection by the scan-copier (but intact check-off, as described earlier).

- Anticipatory substitutions arise from miss election and then failed check-off, perseveratory substitutions from failed check-off and then misselection. "No source" substitutions are explained as incompletely or incorrectly transferred to the slot location, or an extra word found its way erroneously into the set and was miss elected (e.g. as in 'Freudian'' slips.
- 3. Additions involve miss election and failed check-off, where the miss election os of a segment for a slot that should have been empty.
- 4. Omissions are argued to arise from miss election of a null element followed by failed check-off.
- 5. Shifts are explained by two full misselections, one of a target for a null slot and the other of a null segment for a target slot.

In 1987, however, Shattuck-Hufnagel reexamined the evidence for the existence of processing units between word and phoneme levels and, in light of speech error data involving the movement or replacement of onsets, suggested that syllabic structure plays a more significant role in phonological encoding than she had initially thought. She went on to instead propose a two-stage model of phonological encoding that allowed for the serial encoding of syllabic structure. For a more thorough explanation than space permits me to give here, one should really refer to the original text.

### 2.3.2 Dell's (1989) interactive activation account of phonological encoding

Dell's (1989) interactive activation model is like Shattuck-Hufnagel's (1987) slot-and-filler model in a couple of ways, both in that it can account for phonological encoding and in that Dell (1989), like Shattuck-Hufnagel (1987), revised his earlier (1986) model to allow syllables to encoded serially (previously he had assumed simultaneous encoding of onset, nucleus and coda within syllables).

According to Dell (1989), there exists in the mind two networks (a lexical network and a wordshape network) as in Figure 5.7. This, of course, captures the conception that the phonological structure of a word has two components: a sequence of slots that specifies the shape of the word and the kinds and quantities of syllables and phonemes it contains, and a representation of the sounds that fill them:



Figure 5.7 Dell's lexical and word shape networks (adapted from Dell 1989). The intended word here is 'cat' as indicated by the flag on the listing of 'cat' at the word level in the lexical network. The dotted lines in the diagram represent connections between the lexical network and the word shape network, and the arrows between phoneme categories in the word shape network their sequence of activation. As in Dell's (1986) model, On = onset, Nu = nucleus and Co = coda

To Dell's (1989) mind, the lexical network is where words in the word level are connected with the phonemes that make them up in the phoneme level. At the phoneme level, phonemes are specified for their position in the syllable (i.e. whether they're syllable initial (lie in the onset), syllable medial (lie in the nucleus) or syllable final (lie in the coda)).

Word nodes in the lexical network are also connected to word shape header nodes that represent the pattern of phoneme categories for them, he says (e.g. the word node for 'cat' in Figure 5.7 is connected to the word shape header node 'CVC' which tells the speaker the word to be produced is a single closed syllable). The word shape header node then connects to a series of phoneme categories (e.g. the word shape header node 'CVC' for the word node 'cat' connects to three phoneme categories On (onset), Nu (nucleus) and Co (Coda) where /k/, /æ/ and /t/ will go, respectively).

The word that's currently being phonologically encoded in Dell's (1989) model is flagged (quite literally so with a flag in his drawing of the model but with an arbitrary amount of activation (100

units) in the mind) as being the 'current' word. Upcoming words in the phrase are primed for retrieval with a lesser amount of activation (50 units), and during each time step, nodes send some fraction of their activation to nodes they are connected to above and below them.

After a certain number (that's determined by the speech rate) of time steps has passed the most highly activated phonemes (in the category nodes of the word shape network) are selected for speech production. What order they're articulated in depends on the word shape network's word shape header nodes: these are sent activation from word nodes in the lexical network, and the header that's the most highly activated gets to decide what phonemes appear where.

Phonological errors occur, Dell (1989) says, if an incorrect phoneme is more active than the correct one of the same category (as was the case in his 1986 model). Say the target, like in Figure 5.7, was 'cat' for example. If the phoneme /t/ in onset position (which was meant for 'tack', not 'cat') was more activated than was /k/ in onset position, it might well be mistakenly selected for speech production resulting in the real word error 'tat'. Likewise, if the null coda (which was meant for 'car') or /k/ coda (which was meant for 'tack') were more activated than the /t/ coda, a speaker might well produce 'ca' or 'cack' instead of 'cat'.

### 2.4 Concluding remarks

In this (rather large) section, some background research to current methods in the remediation of spoken word production after brain damage has been outlined. Some of the key assumptions of cognitive neuropsychology have been discussed, as have some of the sources of data that have been called on to develop spoken word production theories (such as speech error data, tip of the tongue data and data from people with aphasia). The degree to which each can adequately account for speech error data has been analysed to some degree, but whilst word finding difficulties can indeed be explained by these models, the phonological mechanisms in the models aren't actually substance-free (and, as I made clear in Chapter 4, this is the theory of phonology that I myself subscribe to).

I am, of course, supportive of the idea that there are multiple components of language processing – a semantic one, and a phonological one – the aphasiology data evidences that much. I also see no reason why semantics and phonology couldn't be active at the same time. However, whilst it wasn't (and still isn't) my intention to propose a new model of word retrieval in this thesis, what I will say is this: If phonology is substance-free (and it certainly seems to be so), then the phonological component of word retrieval should be devoid of phonetic content. Rather, phonology and phonetics should be instantiated separately, with an interface (or interfaces) amid them that bridge(s) the gap between the two. Importantly, deficits in the interface(s) (or beyond it/them) shouldn't be described as phonological ones of word retrieval (even if the interface(s) is/are connected to phonology).

Another shortcoming of what's so far been sketched out in the clinical literature is that the models don't make it clear what the role of the *lexicon* is, where lexical entries are *stored*, where syntax is situated and how connections between the semantic, syntactic and phonological modules are made and communication handled. Thankfully, in *Foundations of Language: Brain, Meaning, Grammar, Evolution*, Jackendoff (2002) offers a solution to each of these problems.

# 2.4.1 An alternative account owing to Jackendoff (2002)2.4.1.1 The role of syntax

Traditional generative grammar only sees syntax as being generative. Neither semantics nor phonology are, in the eyes of generativists. There is 'a fundamental assumption embedded deep in the core of generative theory[...] that the free combinatoriality of language is due to a single source, localized in syntactic structure' (Jackendoff 2002: 107). Jackendoff (2002) takes issue with this, and develops 'the alternative assumption that language has multiple sources of combinatoriality, each of which creates its own characteristic type of structure'. The outcome is a theory of grammar which has a has a tripartite organisation and in which phonology, syntax and semantics are equally as generative as each other. In Jackendoff's (2002) model, syntax is all but one of the three parallel generative components. It is among the combinatorial systems, but it is far from the only one (Jackendoff 2002: 126):

'[The figure reveals] the role of syntax in the parallel architecture. [...] Syntactic structure serves as a "way-station" between [the semantic and phonological] structures, making the mapping between them more articulate and precise. Thus, although syntax is in the center [...], the grammar is no longer syntactocentric[.] Rather, syntax is simply one of the three major generative components in the grammar'

This model, as well as being different from models we see in traditional generative grammar, is different from the models of word retrieval we viewed earlier (e.g. the logogen model

(Morton 1970, 1979), the semantic lexicon model (Butterworth 1989), the WEAVER++ model (Levelt et al. 1999), the interactive activation models (Stemberger 1985 and Dell 1986), and models which gave interactive activation accounts of phonological encoding (Shattuck-Hufnagel 1979 and Dell 1989).

The first difference between aphasiological models of word retrieval and Jackendoff's (2002) is that under Jackendoff's (2002) view, there aren't just two components of word retrieval (semantics and phonology) there are three: one having to do with phonology, another having to do with syntax (which was absent, as its own module at least, from all of the models discussed thus far) and one still more having to do with phonology.

### 2.4.1.2 The integrative and interface modules

Jackendoff (2002) argues for phonological, syntactic and conceptual processors (which are similar to Fodor's (1983, 2000) modules, but it's necessary to point out here that his use of the word *module* is a little different than Fodor's (1983, 2000), as is some other terminology he uses. What Fodor calls *modules* Jackendoff calls *processors*, and for Jackendoff, there are three types:

- inferential processors (which are the exact same thing as Fodor's central systems),
- integrative processors (Fodor's modules) and
- interface processors (which make the integrative processors able to communicate)



Figure 5.8 Jackendoff (2002)'s model of the language faculty (Jackendoff 2002: 199)

According to Jackendoff's (2002) model of the language faculty, there are three integrative processors independent of one another: the phonological integrative processer (which processes phonological structures), the syntactic integrative processor (which processes syntactic ones), and the conceptual integrative processor (which of course, processes structures that are conceptual). The conceptual integrative processor is sometimes called the semantic integrative processor, and I myself use the terms *conceptual* and *semantic* interchangeably throughout.

They communicate with one another via interfaces: the phonological and syntactic integrative processors are interfaced by a phonological structures-semantics structures (PS-SS) processor, the syntactic and conceptual integrative processors being interfaced by a syntactic structures-conceptual structures (SS-CS) processor and the phonological and conceptual integrative processors take the phonological structures-conceptual structures (PS-CS) as an interface.

Under his view, phonology, syntax and semantics are all informationally encapsulated modules, which communicate with one another via interfaces that take their outputs and turn them into inputs the other modules are able to interpret – the PS-SS processor turns phonological output into syntactic input and vice versa, the SS-CS processor syntactic output into semantic input and vice versa, and the PS-CS processor phonological output into semantic output into phonological input.

There are also a couple of phonetics-phonology interfaces specialised for perception and production that a) tally well with a substance-free view of phonology and b) importantly account for anomias that are phonological and post-phonological in nature. It's because of these interfaces, first and foremost, that I subscribe to a model of the mind like Jackendoff's (2002), and so I'll describe them here.

### 2.4.1.2.1 The interfaces

Some have said that the phonetics-phonology interface is mediated by a transduction process that converts substance-free phonological forms into substance-laden phonetic ones. This approach to the interface (that is, that it is a transducer) is indeed the one that's argued for by Fodor (1983) (recall from Chapter 2, section 2.2), but it can be traced back even further, to Chomsky and Halle (1968) and Jakobson et al. (1952). In more recent years, Hale et al. (2007) and Hale and Reiss (2008) have also assumed this position.

What's interesting about theories like these (well, to this thesis at least) is that in differentiating between phonology and phonetics (and separating them with a substance infusing transducer), they can more adequately account for the difference between phonological and post-phonological deficits than current aphasiological models can (see 3.2). Phonological deficits occur due to damage of the phonological module, and post-phonological deficits due to damage of the phonetics-phonology interface(s), I imagine.

In the *Foundations of language: Brain, memory, grammar, evolution,* Jackendoff (2002) offers an alternative view of the interface (or interfaces, I should say), which is the one that I myself am inclined to concur with. Jackendoff (2002) doesn't just propose that there is one interface between phonology and phonetics, you see, he proposes that there are two: one from audition, which translates phonetic data having to do with hearing into phonological data, and one to vocalisation, which translates phonological data, into data having to do with speaking.

These interfaces are therefore different from the others (the PS-SS, SS-CS and PS-CS processors), as whilst the interface processor from audition is designed to take only phonetic information as input and the interface processor from vocalisation is designed to take only phonological information as input (meaning that they're specialised for one domain each), the modules between phonology and syntax and syntax and phonology are specialised for two; the interfaces between phonology and syntax, syntax and semantics and phonology and semantics in Jackendoff's (2002) model are required to have access to both the vocabulary of

both the sending and the receiving module (the phonology/syntax for both phonology and syntax, the syntax/semantics for syntax and semantics, and so on) (Scheer 2014). Jackendoff (2002) calls the phonetics-phonology interfaces domain specific, and the PS-SS, SS-CS and PS-CS interfaces *bi*-domain specific.

### 2.4.1.2.1.1 A side note on the phonetics-phonology interface

Whilst what's going on in the interfaces looks a little like transduction as Jakobson et al. (1952), Chomsky and Halle (1968), Fodor (1983, 2000), Hale et al. (2007) and Hale and Reiss (2008) etc., say in that they each serve to transform the output of one module (that's only interpretable by the module to which it belongs) into an input interpretable by another module, as you delve deeper into Jackendoff's (2002) discourse it's apparent that there is an important difference.

As Iosad (2013, 2017) so rightly points out about the phonetics-phonology interface, Jackendoff's (2002) processors are a lot richer compared to conceptions of the interface owing to Fodor (1983) in which the translation of information interpretable by one system to information interpretable by another is undertaken by transduction (e.g. Pylyshyn 1984). Transducers, for one, are innate and invariant (Hale et al. 2007: 647) and, in being so, they, of course, cannot adequately account for language variation in the same way that Jackendoff's non-innate (and therefore not invariant) interface phonetics-phonology processors can:

> these two transducers [perception  $\rightarrow$  phonology and phonology  $\rightarrow$  articulation] are innate and invariant – they are identical in all humans (barring some specific neurological impairment) and do not change over time or experience (i.e., they do not "learn")

In the 'poor', innate interface model (that's akin to the conception of Fodor (1983)) the mapping between phonological units and phonetic ones (having to do with production and perception) is consistent cross-linguistically for the most part.<sup>6</sup> This, of course, inadequately accounts for the variation we see between languages. In Jackendoff's (2002) 'rich', interface model, on the other hand, Iosad (2013, 2017) helpfully highlights, knowledge is not innate but may be learned, and so there is no expectation of universality.

<sup>&</sup>lt;sup>6</sup> Although Hale et al. (2007) do have a place for language specific mapping mechanisms (Iosad 2013, 2017).

Like Jackendoff (2002) (and indeed Iosad (2013, 2017), I hold the belief that the phoneticsphonology interface, or interfaces, are best viewed not as deterministic transducers, but as something akin to a module (or modules) that translate phonetic information into phonological information, and/or phonological into phonetic. As Iosad (2013, 2017) says, this view of the interface is necessary for the variability of phonetic phenomena. Models of the mind which paint a picture of interfaces as being nothing more than transducers cannot adequately account for the wide range of variability we see between surface representations, and so they surely can't be correct.

Jackendoff's (2002) parallel architecture model solves two problems that the models of word retrieval in the clinical literature cannot, then: a) it describes the part syntax plays in the grammar, b) it describes how syntax is instantiated (as an integrative module like the phonological and semantic integrative modules), and c) how those integrative modules, if informationally encapsulated, communicate with one another (via interface modules). It also, as I said, tallies well with the substance-free phonology thesis, in that it explains where phonetics, if not grounded in phonology, is instantiated and how phonology and phonetics communicate (via yet two more interfaces, one for perception and one for production).

Jackendoff (2002) doesn't stop there, though. He also, in the parallel architecture model, makes a convincing case for the role of the lexicon, and how that, together with everything else, plays its part in language processing.

### 2.4.1.3 The lexicon and lexical entries

Jackendoff's (2002) parallel architecture of the language faculty once again challenges some of the assumptions made in traditional generative grammar. The first assumption it challenges is that lexical items are stored in long-term memory and enter the grammar by being inserted into syntactic structures. The second is that those lexical items are always words. In his parallel architecture, lexical items emerge as *parts of the modules that interface integrative modules*. Lexical items aren't always just words either; they are of heterogeneous sizes. Some are smaller than words, e.g., affixes, and some are larger than words, e.g. idioms (Jackendoff 2002).

For most generativists, the lexicon is the store of words in long-term memory, which are used, by the grammar, to construct phrases and sentences (Jackendoff 2002). 'It is widely agreed

that a word is to be regarded as a long-term memory association of phonological, syntactic, and semantic features', Jackendoff (2002: 130) says, including by himself.

However, 'whilst mainstream generative grammar, following Chomsky (1965), inserts lexical items as a whole into syntactic structure; their phonological and semantic features [aren't] interpreted [until] later in the derivation by the appropriate components' (Jackendoff 2002: 130). But whilst this approach isn't impossible, Jackendoff (2002) points out, it *does* raise the question of why syntax should carry around all of those phonological and semantic features that it can't itself see. Other researchers over the years, having had a problem with this too, have suggested as a solution late lexical insertion, but whilst these approaches are an improvement in that syntax doesn't have to carry around the phonological and semantic features for as long, it still has to carry them, which doesn't make much sense, he stresses (Jackendoff 2002).

Something quite different is suggested in his sketch of a parallel architecture. 'A word, by virtue of it having features in each of the components of grammar, serves as part of the linkage between the multiple structures', he says (Jackendoff 2002: 130). The proper way to regard [a word] is as a small-scale three-way interface rule[ which] lists a small chunk of phonology, a small chunk of syntax, and a small chunk of semantics' (Jackendoff 2002: 130).

ten Hacken (2019) provides a most helpful example from Dutch. The lexical entry for the word *cheese* (which, in Dutch is *kas*), will look a little something like this, he says:

a. /kas/b. noun, non-neuterc. [Thing CHEESE]

This explains why people with semantic anomia sometimes mistakenly say *ham* when they mean to say *cheese*, because although syntactically they are the same, they are different semantically and phonologically. An intact phonological processor likely wouldn't pose a problem. But an impaired semantic one might, especially given that *cheese* and *ham* are often associated with one another, in that they, apparently, make a perfect food pairing. The lexical entry for *ham* in Dutch (*ham*) would look something along the lines of this:

(2) a. /ham/b. noun, non-neuterc. [Thing CHEESE]

If the function of lexical items, Jackendoff (2002) argues, is to serve as interface rules, then, the lexicon as a whole has to be regarded as part of the interface components. '[T]he formal role of lexical items is not that they are "inserted" into syntactic derivations, [he says,] but rather that they establish the correspondence of certain syntactic constituents with phonological and [semantic] structures' (Jackendoff 2002: 31).

Now, in addition to the usual lexical items that list all three structures (phonology, syntax and semantics), there exist lexical items that have phonology and semantics but no syntax, phonology and syntax but no semantics, syntax and semantics with no phonology and stored pieces of phonology that lack both syntax and semantics.

Examples of lexical items that have phonology and semantics but no syntax can appear alone as utterances, but cannot be combined into sentences with other words. These include (Jackendoff 2002: 131-132):

a. yes, no
b. hello, goodbye, thanks
c. ouch, oops, wow, phooey, hooray, gadzooks, oboy, oy vey, dammit, shit, yuck, upsey-daisy
d. hey, fiddlesticks, pshaw, humph, oo-la-la

e. shh, psst, tsk-tsk

- f. abracadabra, hocus-pocus
- g. bow-wow, cockadoodledoo

Words that have phonology and syntax but no semantics include the *it* in *it's hot in here* which is present to carry tense and tense only, Jackendoff (2002: 132) points out, and an example of a lexical item that has syntax and semantics but no phonology is mainstream generative theory's PRO which serves as the subject of infinitives (*Bill tried [PRO to talk]*).

Stored pieces of phonology that lack both syntax and semantics, meanwhile, include the nonsense refrains that are used to take up metrical structure in songs: fiddle-de-dee, hey-

diddle-diddle, hickory-dickory-dock, eenie-meenieminie-moe, she-bop-she-bop, rikiti-tikititin [and] ink-a-dink-a-doo', he says (Jackendoff 2002: 132).

So, to recap, lexical entries for Jackendoff (2002) are stored in long-term memory, and combined to form linguistic structures in short-term memory during language production. Jackendoff's (2002) model of competence is even more relevant to word retrieval than that, though. He uses it to sketch a model of performance.

### 2.4.1.4 Lexical processing

According to Jackendoff, there should be a clear connection between theories of competence (linguistic structure) and theories of performance (language processing). As such, he makes sure to show how his parallel architecture, and in particular, his treatment of the lexicon in the model, fits nicely into analyses of lexical access in production.



Figure 5.9 Chomsky's model of language (ten Hacken 2019: 207)

The standard architecture for generative grammar from *Aspects* looks a lot like what's sketched out in the model of language above. Generativists, recall, see grammar as being both syntactocentric and derivational as so, in Figure 5.9 above, with the generative capacity of language coming from syntax and syntax only (Jackendoff 2002). Linguistic structure is built by an initial stage of syntactic derivation in traditional generative grammar, Jackendoff (2002) explains. These derivational rules produce levels of syntactic structure that are subjected to

semantic and phonological interpretation. Semantic and phonological structures, therefore, do not have any generative capacities of their own (Jackendoff 2002).

The logical directionality of this syntactocentric architecture, however, is at odds with the logical directionality of language processing, it's pointed out by Jackendoff (2002), for in perception, one has to get from sounds to meanings (phonology to semantics) and in production, from meanings to sounds (semantics to phonology). Jackendoff's (2002) parallel, constraint-based architecture, on the other hand, is not. In fact, it bears a striking resemblance to theories of processing, particularly Levelt et al.'s (1999).

'A crucial role in this processor is played by linguistic working-memory', Jackendoff (2002: 200) emphasises, 'which is to be understood not just as a static storage space for linguistic material, but as a dynamic "workbench" or "blackboard" on which processors can cooperate in assembling linguistic structures. It has three divisions or "departments" or "buffers," corresponding to the three levels of linguistic structure' (Jackendoff 2002: 200).

In language production, the processor starts with figuring out what concept it wants to retrieve the word for in the semantic integrative module. The module then sends a call to the lexicon (in long-term memory), asking what saved structures (from as small as affixes, remember) could potentially express parts of that, and various candidates are activated. The lexical items activated then compete for integration into semantic working-memory. At some point, one wins – this is called lexical selection in the clinical literature.

Once the lexical item's semantic structure is bound from long-term to working-memory, its syntactic and phonological structures follow suit and bind to their blackboards. Mind, this is all being done incrementally, Jackendoff (2002) emphasises – although this description of the model's processing *sounds* sequential, it is not necessary for one level of structure to be fully formed in order for its interfaces to start passing information up and down the line Jackendoff (2002).

Once the other (syntactic and phonological) structures are all activated in their respective domains, the semantic, syntactic, and phonological structures combine together to create the intended message. Phonology, of course, speaks via an interface to phonetics, and the word is pronounced.

This clearly tallies well with Levelt et al.'s (1999) model WEAVER++ model (recall from section 2.2.3) which makes mention of lemmas and word forms (and of course, all others that do). Levelt et al.'s (1999) lemmas and word forms have clear correspondences in Jackendoff's (2002) model, even though there aren't 'lemma retrieval' or 'word-form encoding' areas argued for by Jackendoff (2002) as they are Levelt et al. (1999):

At the moment in time when a lexical item is initially activated, only its semantics is of interest. The conceptual integrative processor, which binds the lexical item to part of the thought being expressed sees nothing but meaning. The next step is to use the semantics-syntax interface to put some material on the syntax blackboard. For this process, it is crucial to activate and bind the item's syntax. Then the syntactic integrative processor can begin to work this item into the syntax of the utterance being built. None of the processors invoked so far can see the item's phonology-so for all intents and purposes they are working with the item's lemma. And whether or not the item's phonology is activated in long-term memory at this point is irrelevant to these processors. Similarly, the later step of phonological integration is carried out by the syntax-phonology interface processor and the phonology interface processor. These can see only the relation of the item's syntax to its phonology, and are more oblivious to its meaning. Hence they are in effect working with the item's word form.

Now, given the constraints on time and space, I will leave the discussion of Jackendoff's (2002) model here for now, so that my musings don't venture *too* far beyond the scope of this thesis, which was originally set to answering the question of whether or not phonology can or cannot be considered modular according to the definitions of modest (Fodor 1983, 2000) and massive (Carruthers 2006a) modularists. However, whilst it was not my intention in this thesis to pitch one model of the mind against another, or necessarily make an argument for Jackendoff's (2002) model or even one of my own, it *is* worth thinking about this at least some shallow depth, I believe, as the conclusions I come to at the end of this thesis interestingly provide some support for, and against, the model of the mind Jackendoff proposed in 2002. In Chapter 4, I presented evidence for phonology's informational encapsulation, and in the rest of Chapter 5 (coming up) I present evidence against phonology's domain specificity, both of which are required, I argue, by Jackendoff (2002).

In what follows, I return to the exploration of whether or not the phonological module can be considered domain specific, through experimental treatment of phonological anomia with semantic and phonological therapies. I begin with a description of anomia in this next section, section 3.

### 3. Anomia

We obviously know now that aphasia is an acquired disorder of language that presents as a result of damage to the brain areas responsible for the production and/or comprehension of language and its components and that, depending on the site or sites of damage and the relative impairment or sparing of the language components, different subtypes of aphasia can emerge (Helm-Estabrooks et al. 2014). As I mentioned in section 2, there are two major subtypes of aphasia: Broca's aphasia and Wernicke's aphasia. These are caused by damage to the lateral frontal, suprasylvian, pre-rolandic area or the adjacent subcortical periventricular white matter, and the posterior third of the superior temporal gyrus, respectively (Helm-Estabrooks et al. 2014).

Broca's aphasia is otherwise known as expressive aphasia and is so-called because it is characterised by impaired language production with unimpaired language comprehension (Kertesz 1982). Wernicke's aphasia, on the other hand, (which goes by the name of receptive aphasia as well), causes problems of a similar vein but in the opposite direction. While a Broca's patient experiences problems with language production but not its comprehension, a Wernicke's patient has them with comprehension but not production (Albert et al. 1981). Other (more minor) subtypes of aphasia include conduction aphasia, transcortical motor aphasia, transcortical sensory aphasia and global aphasia (Helm-Estabrooks et al. 2014), and with all of the different subtypes, different symptoms are associated. Some level of anomia meanwhile seems to be seen in all of the aphasias (Manasco 2014).

Anomia (as I've briefly touched on before but will describe in more detail here) is an impaired ability at accessing words in and retrieving them from the mental lexicon (Goodglass and Wingfield 1997). This can range from a mild difficulty in producing desired words during conversational discourse to a virtual inability to produce them under any conditions at all (Helm-Estabrooks et al. 2014). Being that anomia is a symptom of aphasia, it is thought to result from damage to the regions of the brain associated with it in the left hemisphere (Woollams et al. 2008). This damage can be either traumatic or acquired (Damasio 1992), with causes of traumatic brain injury including the likes of falls, accidents or violence involving a blow to the head, and causes of acquired brain injury poisoning, infection, strangulation, choking, drowning, stroke, heart attacks, brain tumours, aneurysms

123

and even neurodegenerative diseases such as Parkinson's disease, Alzheimer's disease and Huntington's disease (Budd et al. 2010).

In part 2 of this chapter, I provided an overview of some theories of spoken word production. The theories were developed with the aim of accounting for the patterns of speech errors we see in normal subjects and evaluated in terms of their ability to do so, but one could (and indeed Nickels (1997) does) argue that a model of language processing cannot be considered sufficient or efficient unless it can account for impaired performance as well as performance that is unimpaired. Some authors (for example Butterworth (1989, 1992) and Monsell (1987)) have used neuropsychological data as support for the sufficiency/efficiency of some theoretical models over others and this – using neuropsychological data to inform theory about how the mind is modelled, that is – is critical, according to Nickels (1997: 99):

I would argue that a model of language processing cannot be considered adequate unless it can account for patterns of impaired performance in addition to the "normal" speech error and experimental data. The use of neuropsychological data in this way rests on the assumption that patterns of language breakdown reflect the structure of the normal language system. Thus, "the differences between normal language mechanisms and the set of language mechanisms available to aphasics are [considered to be] subtractive that is, the aphasic is simply lacking some of the components available to normal language speakers, rather than inventing new ones" (Saffran, 1982, p.318).

Now, whilst the assumption that aphasics' 'patterns of language breakdown reflect the structure of the normal language system' (Nickels 1997: 99) is by no means a universally accepted one, it is the basis of the neuropsychological approach to aphasia therapy that is used today and is, therefore, implicit in much of what follows here. Discussion of aphasia treatment, however, will be best left to section 4 - in this section, I will first describe the different deficits of spoken word production in aphasia and relate them to the theoretical models that were reviewed in the previous one.

As I've said, in the past people have concerned themselves with relating the type of breakdown in naming to specific aphasic syndrome categories. Benson (1979) was among the first researchers to do this; he drew a distinction between four anomias (word production anomia, paraphasic word production anomia, word selection anomia and nominal anomia) and related them to four aphasic syndromes (Broca's aphasia, Wernicke's aphasia, conduction aphasia and anomic aphasia, respectively). Syndrome labels aren't always reliable indicators of the locus of difficulty in word retrieval, however, (as we'll see in section 4), and so more recently researchers have reduced Benson's four forms down to just three (for example Lambon Ralph et al. (2000). Lambon Ralph et al. (2000) distinguish between semantic anomia (a word finding difficulty caused by damage to the semantic level of language processing), phonological anomia (a word finding difficulty caused by damage to the phonological) and classical anomia (a form of anomia in which patients present with word finding difficulties that are neither caused by damage to the semantic not the phonological levels of language processing).

In this present chapter I'd prefer to be a *bit* more specific about the phonological aspect of word retrieval than Lambon Ralph et al. (2000) are however and differentiate between phonological (or word form) anomia and something called disordered phoneme assembly as Laine and Martin (2006) do in *Anomia: Theoretical and Clinical Aspects*. Like Lambon Ralph et al. (2000), Laine and Martin (2006) also argue for three anomias, but whereas Lambon Ralph et al. (2000) argue for semantic, phonological and classical anomias, Laine and Martin (2006) argue for semantic, phonological and classical anomia and disordered phoneme assembly, which correspond to the main stages of word production in a standard logogen model: retrieval of meaning (at the semantic stage), retrieval of the word form (from the phonological output lexicon) and programming the relevant phonological output (in the phonological output buffer).

These anomia types also find their correspondences in the more recently founded functional models Laine and Martin (2006) point out (such as Levelt et al. (1999's) model of word retrieval). In Levelt et al's (1999) model, 'semantic anomia would roughly correspond to conceptual- and lemma-level deficits' and '[w]ord form [that is, phonological] anomia [...] to difficulties in accessing the lexeme that specifies [...] phonological information of the to-be-produced target word' they say (Laine and Martin 2006: 37). Disordered phoneme assembly, meanwhile, corresponds to 'impairments in the later processes of [e.g.] phonetic encoding that are needed to create the "articulatory score" of the target word for motor output' (Laine and Martin 2006: 37). This, of course, has nothing to do with lexical access (a point I'll come back to in section 3.2). To reiterate what I said in section 2.4, though, aphasiology models of word retrieval's phonological levels are grounded in substance. If phonology is substance-free, then a model which makes use of an interface theory like Jackendoff's (2002) would account for the inner workings of the mind more accurately. Phonological anomia and disordered phoneme assembly could still be differentiated if the phonological component of

the model were to be substance-free. In Jackendoff's (2002) model, semantic anomia could be explained by damage to the semantic integrative module, phonological anomia the phonological integrative module and disordered phoneme assembly the interface module between phonology and phonetics.

Now, regardless of the differences in terminology, Benson's (1979) differentiation between semantic anomia and word selection anomia, Lambon Ralph et al. (2000)'s between semantic and phonological and Laine and Martin's (2006) between semantic and word form are all alike in the way that they separate semantic processes from phonological ones, and I will take this division as the starting point in my discussion. Discussions of the breakdown of word retrieval in aphasia have often centred around two specific papers: Howard and Orchard-Lisle's (1984) and Kay and Ellis' (1987) which each detail a single case study of an aphasic subject with impaired picture naming. Ellis and Young (1988) used these two subjects (JCU from Howard and Orchard-Lisle (1984) and EST from Kay and Ellis (1987)) to illustrate the division between semantic and phonological levels of impairment in naming, and in the following subsections, I will do the same.

The remainder of this section is split into two halves. In the first, I consider semantic anomia and in the second phonological, as well as deficits beyond phonological access in section 3.2.2 of 3.2 to explain why they won't be a topic of discussion in section 4.

### 3.1 Semantic anomia

Unsuccessful attempts at word retrieval often result in speech errors that resemble the target in meaning (semantic speech errors) or in sound (phonological speech errors). These are taken to be symptomatic of anomia. The first, semantic speech errors (or more accurately, semantic *paraphasias* with which irretrievable target words are substituted with words that are semantically related to them such as *refraction* for *reflection* (which both have to do with the dispersion of light (Damasio 1992; Fitzgerald 1996; Marshall et al. 1998)) are most commonly observed in people with semantic anomia. Phonological paraphasias, on the other hand, which are the substitutions of target words with words or nonwords that resemble them phonologically (like *viscosity* ([viskpsiti]) for *velocity* ([vilpsiti]) (a real word, formal error), for example, or *chromosome* ([krəuməsəun]) for *chromosome* ([krəuməsəum]) (a nonword phonemic error) (Brookshire 1996)) are said to be symptomatic of phonological anomia. Semantic paraphasias suggest that while phonological information about a target word can be successfully retrieved by the patient, not all of its semantic information can be (Nickels 1997). Phonological paraphasias, on the other hand, sometimes suggest that while semantic information about a target is available for retrieval, there are problems with a patient's access to the sounds constituting its spoken form (Nickels 1997), (though that isn't *always* the case, as we'll soon see).

This section starts by reviewing some case studies of patients who produce semantic paraphasias in their attempts at word retrieval and are as such thought to have deficits at the semantic level of language processing. This is shortly followed by a review of studies in which participants produce phonological paraphasias (in 3.2), who are thought to have deficits at the phonological one. I must point out here, however, that I use the term 'semantic system' somewhat loosely throughout the thesis, as the nature of semantic representation is a matter of dispute. While some researchers (e.g. Levelt 1989) have argued for separate conceptual- and lexical-semantic levels in word retrieval, others (e.g. Riccoch et al. 1988; Caramazza et al. 1990; Hillis et al. 1990) have argued for a single, modality-independent central system. People have even 'argued for a multicomponent semantic system that separates visual and verbal semantics (Paivio 1991) or is organised along the different sensory-functional domains (Warrington & Shallice, 1984)' (Laine and Martin 2006: 40).

I will use semantic system to refer to all of these things, but will draw distinctions in what follows for purposes of clarity. A question that begs to be answered in this thesis as well (and so I do that in sections 3.1.1 and 3.1.2), though, is 'which semantic errors reflect impaired access to conceptual-semantic representations and which reflect permanent erasure of conceptual-semantic representations from long-term memory (Laine and Martin 2006)'? Progressive neurological diseases are thought to lead to either impaired access to (as in cases of progressive fluent aphasia or progressive non-fluent aphasia) or loss of (as in cases of semantic dementia) representations. Aphasias associated with non-progressive neurological diageases (Laine and Martin 2006).

As you've likely deduced from the discussion thus far, then, I'll be describing the difference between these through an exploration of cases found in the aphasiology literature. I'll study Howard and Orchard-Lisle's (1984) JCU and Hillis et al.'s (1990) KE in section 3.1.1, and Murre et al.'s (2001) AM in section 3.1.2.

127

#### 3.1.1 Impairment in access to conceptual-semantic representations

Howard and Orchard-Lisle (1984) describe the case of JCU, a global aphasic who was rendered with a large fronto-temporo-parietal haemorrhage lesion in her left hemisphere and a smaller one in her right following a stroke. Both her comprehension and production was impaired, they said, with spontaneous speech being restricted to simple *yes/no* responses and a repertoire of recurrent utterances like *I don't understand*. Picture naming was poor too (JCU was only able to name 5 out of a possible 30 items from the Western Aphasia Battery (an instrument for the assessment of aphasics) correctly (Kertesz and Poole 1974)) but she was sensitive to phonemic cueing (a type of anomia therapy). When provided with phonemic cues (such as the first sound /t/ of the target *tiger*) JCU was able to name a further 18 out of 25 items correctly. Howard and Orchard-Lisle (1984) investigated in detail the effect of phonemic cues on her naming and found them to not only be highly effective for JCU, but that she could be induced to produce semantic errors when provided with false cues that were semantically related to the target. The target *tiger* could be cued by /l/ for example, for /l/ is the initial phoneme of *lion, tiger*'s semantic coordinate.

When these errors were re-presented to her in a picture verification task (e.g. she was shown a picture of a tiger and asked *Is this a lion?*), JCU incorrectly accepted 55% of her semantic errors as the correct name for the picture. These observations, together with the one that JCU's output phonology (which was measured by single word repetition tasks) was preserved, more or less, led Howard and Orchard-Lisle (1984) to conclude that JCU's semantic errors were due to underspecified semantic representations at the semantic level of language processing.

Hillis et al. (1990), on the other hand, described the case of KE, a 52 year old right handed male who'd had a thrombo-embolic stroke six months before the study started. Like JCU, KE's speech output was limited, but for KE this was to single nouns and well-known phrases. Repetition of words and nonwords was preserved, and phonological paraphasias were few and far between. Hillis et al. (1990) tested KE's performance on a number of single word production tasks (auditory word to picture matching, visual word to picture matching, oral naming, written naming, oral reading and writing to dictation) and found that he made significant semantic errors in all (as can be seen by the results in Table 5.1, which depicts his performance on semantic tasks):

Task	Total correct (n=144)	% correct	Semantic errors/normal errors	% semantic errors
Auditory word to picture matching	83	58	58/61	95
Visual word to picture matching	91	63	39/53	74
Oral naming	80	56	59/64	92
Written naming	77	53	50/67	75
Oral reading	84	58	52/60	87
Writing to dictation	84	58	40/60	67

Table 5.1 KE's performance on semantic tasks (adapted from Hillis et al. 1990: 203)

KE's semantic processing was so impaired, in fact, that he made different errors for the same stimulus across a number of the tasks (when presented with a picture of an arm in an oral naming task and asked its name KE said *finger*, e.g., but when shown the same picture in a written naming task he said *leg*, when reading aloud he said *ear* and when writing to dictation *hand*). This led Hillis et al. (1990) to conclude that KE had a central semantic processing deficit that could be explained within a single modality-independent semantic system that represents the perceptual, functional and relational components of concepts like the one in Morton's (1970) logogen model.

Howard and Orchard-Lisle's (1984) patient JCU's deficits couldn't, on the contrary, be accounted for by this sort of structure though Butterworth et al. (1984) point out, for although JCU was impaired in verbal comprehension tasks, she performed perfectly normally in a nonverbal picture association task making her word retrieval difficulties lexical-semantic in nature. JCU's pattern of impairments could, however, be explained by two-step models like their own, they say, that propose partially independent modality-specific semantic systems.<sup>7</sup> According to Butterworth et al. (1984), within their model, an impairment at the semantic lexicon level with unimpaired semantic/conceptual representations would give rise to the kind of deficits JCU displays.

<sup>&</sup>lt;sup>7</sup> Clearly, then, it could be described by Levelt et al.'s (1999) WEAVER++ model as well.
3.1.2 Permanent erasure of conceptual-semantic representations from long-term memory In their article Slowly Progressive Aphasia without Generalized Dementia Mesulam (1982) described six patients with progressive aphasia ('a slowly progressing aphasic disorder without the additional intellectual and behavioural disturbances of dementia' (Mesulam 1982: 592)). While these were not the first reported cases of the condition (see Pick (1892), Warrington (1975) and Schwartz et al. (1979) for other representative examples), Mesulam's (1982) paper was the first to highlight the fact that patients with neurodegenerative diseases could present with focal cognitive deficits (Murre et al. 2001). Following the publication of Mesulam's (1982) seminal report, over a hundred patients with progressive aphasia have been recorded. It's become clear from these, though, that the term has been used to describe two very different disorders: progressive non-fluent aphasia and progressive fluent aphasia (see Hodges and Patterson 1996 for a more thorough review).

Now, whilst the phonological and syntactic aspects of language are damaged in patients with non-fluent aphasia, patients with progressive fluent aphasia's speech structures are not (Croot et al. 1998). They do, progressive fluent aphasics, that is, display difficulty in producing the names of what were once familiar people, places and objects (i.e. noun retrieval) however, as well as deficits in word comprehension, failing to understand the most straightforward of questions sometimes and/or keep up with conversational discourse.

Deficits are seen on a number of different verbal based semantic tasks such as picture naming, word to picture matching, category fluency (e.g. 'producing as many exemplars from a semantic category (e.g. animals) in 1 min' (Murre et al. 2001: 649)), picture sorting (i.e. 'grouping black-and-white line drawings depending on various pre-specified criteria, such as living versus non-living; electrical versus non-electrical and so on' (Murre et al. 2001: 649)), naming an item when given a verbal description of it (e.g. 'toaster' for 'an electrical kitchen appliance that is used for browning bread' (Murre 2001: 649)) and the other way round, verbally describing an item when given its name (i.e. 'an electrical kitchen appliance that is used for 'toaster') (Hodges et al. 1992; Hodges and Patterson 1996; Murre et al. 2001).

On non-verbal testing of semantic memory, patients show deficits when they're asked to select the appropriate colours for black and white line drawings of what were once familiar objects (such as 'green' for an apple and 'orange' for an orange), to use once familiar objects appropriately (such as a comb to comb their hair), to draw once familiar objects from memory

and to match common animal sounds to appropriate pictures (such as a meow to a cat and and a bark to a dog) (Bozeat et al. 2000; Hodges et al. 2000; Murre et al. 2001). Patients typically perform well on tests of non-verbal problem solving, visuo-perceptual and spatial ability and working memory though, even in the latter stages of the disease (Breedin et al. 1994; Hodges et al. 1994; Waltz et al. 1999; Murre et al. 2001).

The relatively selective loss of semantic memory seen in progressive fluent aphasia has led researchers (e.g. Snowden et al. 1989; Hodges et al. 1992; Breedin et al. 1994) to adopt the term 'semantic dementia' for it instead, and in 1992 Hodges et al.'s put forward five diagnostic features, which were succinctly summarised in 1997 by Nickels as follows (Nickels 1997: 113):

- (i) selective impairment of semantic memory causing severe anomia, impaired spoken and written single-word comprehension, reduced generation of exemplars on category fluency tests and an impoverished fund of general knowledge;
- (ii) relative sparing of other components of language output and comprehension, notably syntax and phonology;
- (iii) normal perceptual skills and nonverbal problem-solving abilities;
- (iv) relatively preserved autobiographical and day-to-day (episodic) memory and
- (v) a reading disorder with the pattern of surface dyslexia.

Hodges et al. (1992) went on to describe five patients who presented with these features, and whilst all five of the patients' semantic systems were impaired, one patient (PP) was affected more than most. When asked whether she'd ever been to America, for example, PP replied 'what's America?', and when asked what her favourite food was replied 'food, food, I wish I knew what that was'. What's most interesting though is that while PP could match a photograph of an object to an unusual view of that object (i.e. another photograph of it taken from above, below or the side) when presented with visually similar distractors, she was unable to name any of the objects in question.

With regards to picture sorting tests from the Semantic Memory Test Battery (which, rather self explanatorily, is used to test participants' abilities to remember meaning) all five of Hodges et al.'s (1992) patients performed poorly, able to pictures using broad distinctions (such as living vs. not living) but unable to do so with subtler distinctions (such as British vs. foreign animal or fierce vs. non-fierce animal). When it came to intermediate level

distinctions (such as separating household items from musical instruments from vehicles and land animals from sea creatures from birds), PP was impaired on those too (though the others weren't significantly affected).

Perhaps the most representative case of semantic dementia, however, is the one described in Murre et al. (2001) of AM. AM was a 64 year old well-educated man with an undergraduate degree in engineering and a master's degree in science. He worked in management (responsible for over 450 employees) for a renowned international company and was referred for clinical examination after presenting to his general practitioner with progressive word finding and comprehension difficulties. A structural MRI revealed temporal atrophy in the left hemisphere of his brain, and upon examination, his speech was semantically empty (though fluent with preserved syntax and phonology).

When it came to picture naming AM was considerably compromised, scoring only 3 out of 48 correct when tested on a set of high frequency items that should have been familiar to him (Knott et al. 1997). He most frequently responded with nothing at all, but on occasion produced some short, circumlocutory responses that were very vague. Semantic paraphasias were rare, and no phonological paraphasias were observed at all in AM's naming. AM's performance on a word-picture matching task using the same items that were elicited by the picture naming task was relatively normal, and the same was true for other word-picture matching tasks (though he did display a frequency effect with better comprehension of higher-frequency items than lower-frequency ones). Of the errors AM produced on the word-picture matching tasks semantic ones were most common, and he was also impaired on a semantic feature questionnaire which required him to answer with a yes or no to simple questions such as 'Does an ostrich have a long neck?'.

The nonverbal version of the Pyramids and Palm Trees test (which was made to measure conceptual knowledge) proved particularly difficult for AM: whilst his repetition of single words and nonwords was within normal limits for items up to three syllables long, he had a strong tendency for phonological errors in longer words and nonwords, and words of a low frequency. Further evidence for a conceptual-level impairment came from AM's difficulties in the appropriate selection and use of everyday items (he put a closed umbrella over his head horizontally in a rain storm and orange juice into his lasagne, for example). A sample of his AM's speech demonstrating the extent to which he was unable to retrieve

words is given below (Murre et al. 2001: 651):

*Examiner:* Can you tell me about a time you were in hospital?
AM: Well one of the best places was in April last year here (ha ha) and then April, May, June, July, August, September and then October, and then April today.
Examiner: Can you remember April last year?
AM: April last year, that was the first time, and eh, on the Monday, for example, they were checking all my whatsit, and that was the first time, when my brain was, eh, shown, you know, you know that bar of the brain (indicates left), not the, the other one was okay, but that was lousy, so they did that and then doing everything like that, like this and probably a bit better than I am just now (indicates scanning by moving his hands over his head).

Now, although the three cases presented here (JCU from Howard and Orchard-Lisle (1984), KE from Hillis et al. (1990) and AM from Murre et al. (2001)) all exemplify patterns of deficit that can be called semantic anomia, AM differs from JCU and KE in that his semantic anomia is due to dementia, not aphasia, and therefore cannot be treated. Treatments for semantic anomia in aphasia are designed to strengthen connections between semantic and lexical representations (Jefferies and Lambon Ralph 2006; Antonucci and Reilly 2008), but in dementia, it is too late for that as those connections haven't been weakened through damage but lost altogether (Peach and Shapiro 2012). This thesis will, therefore, look at cases of semantic anomia in aphasia only, from here on out. No more will be said about disorders of semantic memory, and cases of them that are in the collection of texts I extract my data from will be excluded from my analysis in section 4.

#### 3.2 Phonological anomia

In this section, I turn my attention to the wide range of phenomena that have to do with deficits in the retrieval of a lexical item's phonological form. Patients with phonological anomia are those that have intact processing at the semantic level, but fail to retrieve the phonological forms of target words in output (Nickels 1997). Phonological errors can occur for one of two reasons, though: a) because there is impaired access to the phonological representations themselves (the reason which will be explored in section 3.2.1) or b) because of impairments in the processes involved in phonological encoding (which is the one that'll be focus of 3.2.2) (Nickels 1997). Only the first kind of error can be attributed to phonological anomia, however, as only these have to do with lexical access. The second kind of error is due to some post-lexical impairment, as section 3.2.2 will explain.

#### 3.2.1 Impairment in access to output lexicons from the lexical-semantic system

The first possible post-semantic place of impairment that could lead to anomia is in the connections between semantics and the output lexicon, Laine and Martin (2006) say. With this, one would expect to find word retrieval impairments with unimpaired comprehension and normal performance on word production tasks that can be completed without semantic support (Laine and Martin 2006).

The most clear-cut cases of impairment in access to output lexicons from the lexical-semantic system are perhaps those that were provided by Lambon Ralph et al. (2000), who reported on two traumatic head injury patients (GM and JS). GM's word retrieval difficulties were milder than those of JS: his score on the Boston Naming test (a tool that's used in speech and language pathology to determine patients' abilities to name) was 31/60 while JS' was 16/60. When GM failed to retrieve a target, he tended to produce circumlocutions such as 'the burial chamber for Egyptian kings... it's got three 'beats'' for 'pyramid', though semantic errors (that were spontaneously rejected) were sometimes produced as well. GM evidenced tip of the tongue behaviour too, in that he was able to provide phonological information about targets (such as the number of syllables they were made up of, as evidenced by 'three 'beats'' for 'pyramid') in the aforementioned example. JS, on the other hand, didn't evidence any phonological knowledge, and circumlocutions for JS were produced upon request only. For JS omissions were the most frequently occurring error type, though he did produce (and like GM, spontaneously reject) semantic errors as well.

Neither GM nor JS produced any phonological paraphasias in speech output tasks, and they both performed within the normal range on word comprehension and semantic-associative tasks (with the exception of GM on something called the PALPA Word Semantic Association task (which is used to check for impairments in semantic processing) where he had problems with abstract items). Repetition and reading performances on both words and nonwords were also within normal limits, which (together with the rest of the results) would suggest semantic processing (but not phonological processing) was intact in both patients.

The data paints a picture, Lambon Ralph et al. (2000) argue, of patients who have access to the meanings of target words (and sometimes even partial information about them), but not enough phonological information to retrieve the targets as nonanomics would be able to. GM and JS' lack of access to full phonological information leads to a selection problem, they say,

whereby semantically close alternatives to targets with more easily accessible phonologies compete for output (and sometimes win).

Similar patterns were reported in 1987 by Kay and Ellis who described the case of EST, a patient whose word retrieval deficits were dramatic, but speech was so fluent that he was able to conceal them in spontaneous speech by using alternative words to, or circumlocutions in place of, the target. In picture naming tasks he produced both semantic and phonological errors and in auditory-verbal repetition tasks better repetition of real words than non-words, but a semantic deficit couldn't be the cause of EST's anomia Kay and Ellis (1987) argued, for he showed complete comprehension of the object names he was unable to process for production.

Kay and Ellis (1987) attributed EST's anomia to a difficulty in activating spoken word forms in an intact phonological (output) lexicon instead, stressing that it was his access to items that was impaired not the items themselves, as evidenced by his ability to retrieve names successfully sometimes after extensive effort. They also assembled a more formal argument for their assertion, pointing out that EST showing a superiority for words over non-words in auditory-vocal repetition tasks could very well suggest 'that [the] lexical phonological entries for such words [could] in fact function in response to [that] type of stimulus and [...] contribute to successful word repetition performance' (Kay and Ellis 1987: 626).

## 3.2.2 Impaired phoneme assembly

A distinction must be made, however, between phonological errors that occur as a result of difficulties with lexical access as in 3.2.1 and phonological errors that occur as a result of something further 'down' models of spoken word retrieval, after the phonological form has been retrieved. Lexical access (as you'll recall from section 2 of this chapter) is generally thought of by Butterworth (1989) and Levelt et al. (1999) as being made up of a number of hierarchically organised processes. According to Levelt et al.'s (1999) lexical access theory, as an example, whereby after the phonological form has been retrieved it is encoded with information interpretable by the system that's responsible for its articulation (Laine and Martin 2006). To my mind, deficits in this domain should not be thought of as phonological anomia, then, since phonological form, and these deficits appear post-lexically after the phonological form has already been retrieved.

The terms phonological encoding and phonetic encoding and phonological assembly and phoneme assembly are used interchangeably in the literature to mean the same thing, but the danger, I feel, in doing this is that it blurs the distinction between phonetics and phonology (which, to my mind, are, of course, two entirely different things). The terms phonological encoding and phonological assembly being used in the aphasiology literature to describe a disorder that is post-phonological in nature can be confusing, I feel, and so I'll use the terms phonetic encoding and phoneme assembly only from this point onwards. Recall also that disorders of phonetic encoding/phoneme assembly can easily be described in Jackendoff's (2002) model which argues for a phonological integrative module (which could be the locus of phonological anomia) and a phonology-phonetics interface module (which could be the locus of disorders of phonetic encoding/phoneme assembly.

The classical aphasic syndrome that's been linked to anomia due to impaired phoneme assembly is conduction aphasia (Laine and Martin 2006). This is a form of fluent aphasia in which comprehension is fairly well preserved, but speech output (which is otherwise well articulated, actually) is punctuated by phonological paraphasias (Laine and Martin 2006). These errors are often quite close to the target, which suggests that the patient producing them was able to retrieve the correct lexical-phonological representations for them, but failed with prearticulatory (i.e. phonemic) planning (Kohn 1984). People with conduction aphasia are particularly sensitive to word length, with longer words causing more production problems than shorter ones due to the larger amounts of phonological information they're made up of (Laine and Martin 2006).

A typical case of conduction aphasia has been described by Laine et al. (1992) who explored the nature of naming deficits in 10 people with classical aphasia syndromes. C2, a 64 year old patient with symptoms of conduction aphasia was mildly anomic, scoring 44/60 on the Boston Naming Test and below normal performance on two others (a 106 picture naming task and a synonym production task). Over half of C2's naming errors were phonological, and of those that were left, only one was semantic (though even that he corrected). C2 scored perfect performance on a semantically based picture classification task, a word to picture matching task and a visual odd-one-out semantic-associative task, and these things, coupled with the fact that estimated familiarity of targets had no bearing on success, led Laine et al. (1992) to conclude that C2 had a phoneme assembly deficit that affects spontaneous speech, oral naming and word repetition.

The phonological errors e.g. 'beelwharrow' for 'wheelbarrow' that occur in *nonaphasic* slips of the tongue, note, are generally attributed to errors in post-lexical phonological processes (such as the copying of segments into slots, for example, in a slot-and-filler model like Shattuk-Hufangel (1979)'s, for their speaker's phonological knowledge is intact (Nickels 1997). When aphasic subjects are in tip of the tongue states however, only partial phonological information (such as the initial phoneme of the word or the number of syllables it has) is available for lexical retrieval, and so more complex phonological real word and nonword errors (such as 'Siam' (a real word error) and 'sympoon' (a nonword error) for the target 'sampan' (Brown and McNeill 1966)) are produced.

The phonological errors observed can help researchers reach a conclusion about the cause of them, but there are many more reliable ways of distinguishing deficits of lexical access and post-lexical phoneme assembly deficits than that, such as 'silent phonology' tasks, for instance. In these types of task, subjects are asked questions such as 'Do 'write' and 'right' sound the same?' (a task that requires a judgement of homophony) and 'Do 'white' and 'bright' rhyme?' (a task requiring, rather self explanatorily, a judgment of rhyme). Pseudo-homophone detection judgement tasks (e.g. 'Which one sounds like a real word - 'brane' or 'plane'?') may be used too, as may phonological lexical decision judgement tasks (e.g. 'Does 'brane' sound like a real word or not?'). Whether or not patients can perform these types of task demonstrates whether or not they can access phonological representations. If they are better at these tasks than they are at producing the same words, we can conclude that the problem is post-lexical in nature.

Since this chapter has to do with lexical access (and therefore word retrieval deficits that are lexical and not post-lexical in nature), I won't be including participants with disorders of phoneme assembly in my analysis of anomia therapy in section 4. Researchers don't always specify the cause of their participants' phonological anomias, however; so if the cause of a participant's phonological anomia is unclear, I won't include them in my analysis either. An alternative option would be for me to assess the phonological errors myself in order to work out whether they are due to impaired access to the phonological representations themselves or because of impairments in the processes involved in retrieving the representations for phone assembly, but given that I'm not a clinician I'm clearly not qualified to do so. My analysis will, therefore, include the following: participants with semantic anomia (which always does have to do with lexical access) and participants with true phonological anomia (i.e. participants whose phonological disorder been diagnosed (by aphasiologists) as being at the

137

lexical level). This I'll, of course, come back to when I discuss my data source and sampling in 4.1.

#### 4. Anomia therapy

In this section I will be comparing the effects of semantic and phonological therapies at treating semantic and phonological anomia in order to see whether the semantic and phonological processors are domain specific and therefore modular by Carruthers (2006a)'s definition. If semantic therapy was found to treat phonological anomia (recall from the introduction to this chapter (and indeed the introduction to the thesis in general)) this would count as clear evidence that the phonological processors, recall from chapter 3, are those that are able to only operate on input that's relevant to their function. Input relevant to the function of the phonological processor would, of course, be phonological, not semantic, and so it will be interesting to see how the different therapy's efficacies at treating phonological anomia pattern.

#### 4.1 Data source and sampling

Now, since impairments of word retrieval are a common symptom of aphasia, much clinical time has been spent attempting their remediation. Anomia therapy has focused on improving access to words' meanings in the semantic system (semantic therapy) and improving access to words' representations in the phonological output lexicon (phonological therapy), but there is a complex relationship between the types of language deficits patients have and the types of tasks that are used in therapy to remediate them (Whitworth et al. 2013). Semantic tasks, for example, have been used with the aim of facilitating word retrieval in people with semantic impairments as well as people with phonological impairments. Phonological tasks, meanwhile, have been used to treat both phonological anomia and semantic anomia too, and so tasks aren't always targeted to the level of language breakdown like they were a long time ago.

Since semantic tasks focus on word meanings, examples include word to picture matching (both spoken and written), word to picture verification, feature matching and feature verification, categorisation, relatedness judgements, generation of semantic features and the provision of semantic information in cues (Whitworth et al. 2013). Phonological tasks, on the other hand, which promote access to word forms, include the likes of repetition and reading aloud, as well as tasks which ask participants to reflect more explicitly on a words'

phonological features, e.g. by counting syllables, identifying initial phonemes or syllables or performing rhyme judgments. The provision of phonological and orthographic cues by therapists are counted as phonological tasks too, in the literature, and although orthographic approaches to treatment mightn't at first glance look very phonological for they have to do with graphemes, not phonemes, I feel this warrants a bit of discussion here as there is evidence to suggest that orthographic cues work just as well as phonological ones do (e.g. Bruce and Howard 1987; Best et al. 1997; Basso et al. 2001; Hickin et al. 2002a).

In 1987, Bruce and Howard successfully treated five people with word finding difficulties with an aid that made use of orthographic cues. This was replicated by Best et al. (1997) in a single case study, and it was found that both graphemic and phonemic cues improved naming. Best et al. (1997) suggested that the reason why orthographic cues work is because they enable the client to develop an (unconscious) strategy whereby they link graphemes to their respective insufficiently activated phonemes so that enough activation to facilitate their retrieval can be delivered: '[i]t appears that the treatment affect[s] a fundamental change in [...] word finding, altering automatic [...] not strategic processes in [the] linguistic system' (Best et al. 1997: 134). With regards to orthographic approaches to treatment, then, like Bruce and Howard (1987), Best et al. (1997), Basso et al. (2001) and Hickin (2002a) (among others) I'll be counting them as phonological tasks in this chapter.<sup>8</sup>

In this thesis, the emphasis will be on studies that involve investigation of single cases or series of single cases which enable a detailed examination of patterns of deficit. In studies which combine patients into groups, the results are uninterpretable and therefore uninformative (Ellis and Young 1988). A number of research papers that are devoted to single case studies (and series of single case studies) spanning 1980 to 2002 that examine the efficacy of treatments for word retrieval impairments have been collected by Nickels (2002), and it's from that collection of papers that I compile my own.

The scope of my review has to be narrower than Nickels' (2002), though, for whilst Nickels (2002) restricted her review of the literature to papers published in peer reviewed journals and books that address impairment-level treatments of spoken word production in aphasia (and so did not review those that 'address[ed] treatments that [...] use alternative (non-linguistic)

<sup>&</sup>lt;sup>8</sup> I will remove orthographic therapies from the data set at the end of Chapter 5, though, to see whether without them a different effect is observed for phonological therapy's efficacy at treating phonological anomia.

means of communication to overcome a word-retrieval/production impairment' (e.g. drawing (Hunt 1999; Sacchet et al. 1999)) (Nickels 2002: 936), those that 'focus[ed] on impairments at an articulatory level' (e.g. apraxia of speech (Ballard et al. 2000; Wambaugh and Doyle 1994)) (Nickels 2002: 937), 'treatments "beyond the single word"' (e.g. sentence processing (Mitchum et al. 2000)) (Nickels 2002: 937) or 'rehabilitation [that's] primarily focused at reducing levels of handicap/disability where the rehabilitation is not impairment-based' (e.g. Lesser and Algar 1995) (Nickels 2002: 937), my review restricts itself to those papers in Nickels (2002) that focus on the remediation of spoken word retrieval only (excluding importantly case studies where the causes of patients' deficits have to do with disorders of phoneme assembly).

As I said in section 3, this chapter is concerned with the retrieval of the phonological form, which means that word retrieval deficits that are post-lexical in nature and therefore don't have to do with lexical access such as disorders of phoneme assembly must be excluded from the analysis. If there were phonetics/phonology mismatches in the data set, what the efficacy of semantic therapies at treating phonological anomias could contribute to debates about the modularity of phonology is unclear.

Something I should also point out though, is that whilst Nickels (2002) reports on cases that have more than one damaged level of language impairment (e.g. AER, TRC and PA from Nickels and Best (1996) who have semantic and post-lexical, semantic and phonological and semantic and post-lexical impairments, respectively), I will include only the cases that have semantic impairments only, or phonological impairments only, in my analysis. Nickels (2002) also reports on cases that attempt to facilitate word retrieval with both semantically and phonologically targeted treatments (e.g. Aftonomos et al. 1997), but since I need to answer the question of whether the phonological processor is domain specific (and therefore able to operate only on phonological input) or not, I'd do better to compare the effects of semantically targeted treatments (that act as semantic input) and phonologically targeted treatments (that act as semantic input) and phonologically targeted treatments (that act as semantic input) and phonologically targeted treatments (that act as phonology wouldn't reveal anything about the domain specificity of phonology, for if they helped with word retrieval it would be unclear whether it was the semantic input or the phonological input in them that was responsible for the remediation.

Now, with that being said, of the 43 papers reporting treatment for impaired word retrieval

summarised in Nickels (2002), only 10 met these criteria. The 33 papers that have been excluded from the analysis (and my reasons for excluding them) are presented in Table 5.2 below, but even in the 10 papers I do present I have had to exclude some *results* along the way (for being statistically insignificant), on the one hand, and *participants* (for their treatment having been mixed or for their impairments or treatment outcome being mixed or unknown) on the other. Where the latter was the case, I've made sure to explain the reasoning behind my exclusion of them in my qualitative analysis of the data sample to follow. And, when the reasons for exclusion were even more complicated than that (as was the case with, e.g. Pederson et al. (2001) and Fink et al. (2002)), I've mapped those out in the analysis as well.

Paper	Participant	Impairment	Treatment	Outcome	Reason for
				(treated)	exclusion
Primarily sem	antic tasks	1	I	T	T
Lowell et al.	BB	Conduction	Semantic	Yes	Impairment
(1995)		aphasia			unknown
	BG	Anomic	Semantic	Yes	Impairment
		aphasia			unknown
	SB	Conduction	Semantic	No	Impairment
		aphasia			unknown
McNeil et al.	BO	Anomic	Semantic	Yes	Impairment
(1997)		aphasia			unknown
McNeil et al.	BO	Anomic	Semantic	Yes	Impairment
(1998)		aphasia			unknown
Nickels and	AER	Semantic and	Semantic	No	Impairment
Best (1996)		post-lexical		**	mixed
			Semantic	Yes	Impairment
					mixed
			C	Ver	T
			Semantic	res	Impairment
	TDC	Somentia and	Somentia	Vac	Impoirmont
	IKC	semantic and	Semantic	1 65	mixed
		phonological	Somentie	No	Impoirmont
			Semantic	INO	mixed
	РА	Semantic and	Phonological	Ves	Impairment
	17	post-lexical	Thomoregieur	105	mixed
		post lexiedi	Semantic	No	Impairment
			Semantie	110	mixed
Semantic and	nhonological	tasks			minted
Aftonomos	VAMC	?	Semantic and	9	More than
et al. (1997)	patients 1.	•	phonological		one patient.
()	2.3		F8		impairment
	_,_				unknown,
					treatment
					mixed and
					outcome
					unknown
	Patients 1,	?	Semantic and	?	More than
	2 9, 11, 18,		phonological		one patient,
	20				impairment
					unknown,
					treatment
					mixed and
					outcome
					unknown
	Patients 4,	?	Semantic and	?	More than
	8, 16, 17		phonological		one patient,
					impairment
					unknown,
					treatment
					mixed and

					outcome
Annoni et al. (1998)	JNH	Semantic and phonological	Semantic and phonological	?	Impairment mixed, treatment mixed and outcome unknown
	GE	Semantic	Semantic and phonological	?	Treatment mixed and outcome unknown
	EG	?	Semantic and phonological	?	Impairment unknown, treatment mixed and outcome unknown
Deloche et al. (1993)	RB	Surface dysgraphia	Orthographic and semantic	Yes	Impairment unknown and treatment mixed
	GC	Morpholexical and word retrieval	Orthographic and phonological	Yes	Impairment unknown and mixed and treatment mixed
Deloche et al. (1997)	В	Word finding	Orthographic, semantic and phonological	Yes	Impairment unknown and treatment mixed
	L	Word finding	Orthographic, semantic and phonological	Yes	Impairment unknown and treatment mixed
	A, D, J, Q, C, K	Word finding	Orthographic, semantic and phonological	Yes	More than one patient, impairment unknown and treatment mixed
	I, O	Word finding	Orthographic, semantic and phonological	Yes	More than one patient, impairment unknown and treatment mixed
	H, G, P, M, R	Word finding	Orthographic, semantic and phonological	Yes	More than one patient, impairment unknown and

					treatment
					mixed
	Ν	Word finding	Orthographic,	No	Impairment
			semantic and		unknown and
			phonological		treatment
					mixed
	Е	Word finding	Orthographic,	No	Impairment
			semantic and		unknown and
			phonological		treatment
					mixed
	F	Word finding	Orthographic,	No	Impairment
			semantic and		unknown and
			phonological		treatment
					mixed
Eales and	MB	Semantic	Semantic and	Yes	Treatment
Pring (1998)			phonological		mixed
	NO	?	Semantic and	Yes	Impairment
			phonological		unknown and
					treatment
					mixed
	PU	Semantic	Semantic and	Yes	Treatment
			phonological		mixed
	SK	?	Semantic and	Yes	Impairment
			phonological		unknown and
					treatment
					mixed
Greenwald et	SS	Semantic and	Semantic and	Yes	Impairment
al. (1995)		phonological	phonological		mixed and
					treatment
					mixed
			Semantic and	Yes	Impairment
			phonological		mixed and
					treatment
					mixed
			Semantic and	No	Impairment
			phonological		mixed and
					treatment
				**	mixed
	MK	Semantic and	Semantic and	Yes	Impairment
		phonological	phonological		mixed and
					treatment
			Comentia - 1	Vac	mixed
			semantic and	res	impairment
			phonological		treatment
					mixed
			Somentia and	No	Impoint
			phonological		mixed and
			phonological		treatment
					mixed
Hillis (1008)	HG	Semantic and	Semantic	Ves	Impairment
111113 (1770)		nhonological	Semantic	105	mixed and
	I	phonological	I		mixed and

					treatment
					mixed
			Phonological	Yes	Impairment
					mixed and
					treatment
					mixed
			Phonological	Yes	Impairment
					mixed and
					treatment
					mixed
			Semantic and	Yes	Impairment
			phonological		mixed and
					treatment
					mixed
Le Dorze	LR	Phonological	Semantic and	Yes	Treatment
(1991)			orthographic		mixed
Le Dorze	RT	Semantic and	Semantic and	Yes	Impairment
and Pitts		phonological	phonological		mixed and
(1995)					treatment
					mixed
			Semantic and	?	Impairment
			phonological		mixed and
					treatment
					mixed
			Semantic	?	Impairment
					mixed and
					treatment
					mixed
			Semantic	No	Impairment
					mixed and
					treatment
					mixed
Li et al.	-	Conduction	Semantic and	No	Impairment
(1988)		aphasia	phonological		unknown and
					treatment
					mixed
			Semantic	Yes	Impairment
					unknown and
					treatment
<b>D</b> 1			· · · · ·		mixed
Pederson et	KB	Phonological	Unsupervised	Yes	Treatment
al. (2001)			computerised		unsupervised
			semantic		and patients
			<b>T</b> T · · ·		Danish
			Unsupervised	Yes	Treatment
			computerised		unsupervised
			phonological		and patients
			<b>T</b> T • •	37	Danish
			Unsupervised	Yes	Ireatment
			computerised		unsupervised
			orthographic		and patients
					Danish

	JI Semantic	Unsupervised computerised semantic	Yes	Treatment unsupervised and patients Danish	
			Unsupervised computerised phonological	Yes	Treatment unsupervised and patients Danish
			Unsupervised computerised orthographic	Yes	Treatment unsupervised and patients Danish
	RI	Phonological	Unsupervised computerised semantic	Yes	Treatment unsupervised and patients Danish
			Unsupervised computerised phonological	Yes	Treatment unsupervised and patients Danish
			Unsupervised computerised orthographic	Yes	Treatment unsupervised and patients Danish
Sbisa et al. (2001)	ML	?	Phonological	Yes	Impairment unknown
			Semantic	Yes	Impairment unknown
Spencer et al. (2000)	NR	Phonological and post- lexical	Semantic and phonological	Yes	Impairment mixed and treatment mixed
Thompson and Kearns (1981)	-	?	Semantic and phonological	Yes	Impairment unknown and treatment mixed
Primarily phonological tasks					
Basso et al. (2001)	RF	Anomia	Orthographic	Yes	Impairment unknown
			Phonological	No	Impairment unknown
			Phonological	Yes	Impairment unknown
	MR	Word retrieval	Orthographic	Yes	Impairment unknown
			Phonological	Yes	Impairment unknown
			Phonological	Yes	Impairment unknown
Beeson (1999)	ST	Post-lexical	Orthographic	Yes	Impairment neither

					semantic nor
Carlomagno et al. (2001)	Patients 1-8	?	Phonological	?	More than one patient, impairment unknown and outcome
			Phonological	?	More than one patient, impairment unknown and outcome unknown
Fink et al. (2002)	GM	Post-lexical	Phonological	Yes	Impairment neither semantic nor phonological
	AS	Phonological and semantic	Phonological	Yes	Impairment mixed
	BM	Phonological and semantic	Phonological	Yes	Impairment mixed
	EL	Phonological and post- lexical	Phonological	Yes	Impairment mixed and one is neither semantic nor phonological; treatment unsupervised
	EG	Post-lexical	Phonological	Yes	Impairment neither semantic nor phonological; treatment unsupervised
	RH	Phonological	Phonological	Yes	Treatment unsupervised
Kiran et al. (2001)	RD	Conduction aphasia	Orthographic and phonological	Yes	Impairment unknown
	RN	Conduction aphasia	Orthographic and phonological	Yes	Impairment unknown
Miceli et al. (1996)	RBO	Phonological and post- lexical	Phonological	Yes	Impairment mixed
	GMA	Phonological and post- lexical	Phonological	Yes	Impairment mixed
Murray and Karcher (2000)	HR	?	Orthographic	Yes	Impairment unknown

Raymer et al. (1993)	CG	Phonological and post- lexical	Phonological	Yes	Impairment mixed
	RJ	Phonological, semantic and post-lexical	Phonological	Yes	Impairment mixed
	MR	Phonological, semantic and post-lexical	Phonological	Yes	Impairment mixed
	RE	Phonological and semantic	Phonological	Yes	Impairment mixed
Robson et al. (1998)	GF	Phonological, post-lexical	Phonological	Yes	Impairment mixed
		and semantic	Phonological	No	Impairment mixed
			Phonological	No	Impairment mixed
Sugishita et al. (1993)	HK, MH, SK	?	Phonological	Yes	More than one patient and impairment unknown
	MK, BY, ST, MT, TI, YA, YK	?	Phonological	Yes	More than one patient and impairment unknown
	AI, TD	?	Phonological	Yes	More than one patient and impairment unknown
	YH	?	Phonological	Yes	Impairment unknown
	TZ, TE	?	Phonological	No	More than one patient and impairment unknown
Pashek (1997)	KR	Apraxia of speech	Phonological	Yes	Impairment unknown
			Phonological	Yes	Impairment unknown
Pashek (1998)	WT	Noun and verb retrieval	Phonological	Yes	Impairment unknown
			Phonological	Yes	Impairment unknown
Rose et al. (2002)	AB	Phonological and post- lexical	Phonological and orthographic	Yes	Impairment mixed
			Phonological	Yes	Impairment mixed

			Phonological	Yes	Impairment mixed
Other tasks					IIIACU
MB	MB	Phonological	Semantic	?	Outcome unknown
Thompson et al. (1986)	S1	Word retrieval	Semantic	?	Impairment unknown and outcome unknown
	S2	Word retrieval	Semantic	Yes	Impairment unknown
	S3	Word retrieval	Semantic	No	Impairment unknown

Table 5.2 A breakdown of the 33 single case (and series of single case) studies of word retrieval in Nickels (2002) that I've excluded from my analysis and my reasons for doing so

Something else I'd like to point out before I delve into the discussion of the case studies leftover though is this: whilst in 1985 Howard et al. drew a distinction between three terms: cueing (which has to do with the immediate effects of treatment), facilitation (which has to do with the effects of treatment measured some time later (e.g. an hour, say)), and therapy (which has to do with the long-term effects of treatment), I should point out here that when I say therapy I don't necessarily mean it in Howard et al. 1985's sense. Rather I use treatment and therapy interchangeably to mean the same thing, facilitate to mean remediate and cueing to refer specifically to types of semantic and phonological (treatments/therapies – see) that make use of semantic and phonological cues, respectively.

## 4.2 Data analysis

## 4.2.1 A qualitative analysis of the data

4.2.1.1 Primarily semantic tasks

#### 4.2.1.1.1 Marshall et al. (1990)

Marshall et al. (1990) undertook three single case studies to examine the effects word to picture matching tasks had on word retrieval in patients with aphasia. The patients studied differed in their deficits (one having a deficit in access to the phonological output lexicon and the other two in semantics), but all three were treated with semantic therapy (semantic discrimination tasks).

The first case, RS, was a 45 year old company director who'd suffered a left hemisphere cerebrovascular accident (CVA) in November 1986 which resulted in both dysphagia and

right hemiplegia. When seen ten months later in 1987 RS had good comprehension still but had developed anomia, with a particular difficulty in retrieving verbs. On a word to picture matching test (owing to Kay et al. (1989) in which words are matched with pictures (one being the target and the other four semantic distractors)), RS scored 39/40, and on a spoken synonym matching test (whereby pairs of high- or low-imagery words are judged for similarity of meaning) RS scored better on high-imagery words than he did low- ones (36/38 compared with 26/38, respectively). On the pictorial version of the Pyramids and Palm trees test (Howard and Orchard-Lisle 1984) (which looks at semantic associations between objects), RS made three errors only, which, it should be noted, is within the normal range.

RS was able to readily discriminate between correct and inappropriate names for pictures, but could not be induced into making semantic errors with semantic or phonological cueing. With regards to written stimuli, on these words, he also performed adequately. When it came to RS' ability to read aloud single words he was almost unimpaired, reading all classes of words well and both high- and low-imagery words without error. RS' deficits led Marshall et al. (1990) to conclude that RS had a deficit in accessing words from the phonological output lexicon. Not in semantics, not in output phonology (i.e. phoneme assembly) but in the route connecting these, rather.

The type of treatment used to remediate RS' word retrieval was semantic. Given 50 drawings of low frequency words to name, 25 of these were to be treated whilst the other 25 acted as controls. Over a 2 week period around 3 hours of therapy time was devoted to semantic matching tasks with the treated groups, and at the end of the 2 weeks it became evident that RS was significantly better at naming treated items than untreated items (20/25 compared with 10/25,  $x^2 = 5.16$ , p < 0.05). This suggests, at first glance, that the phonological processor isn't domain specific and therefore massively modular, for semantic therapy was found to facilitate naming.

IS, on the other hand, was a 76 year old retired civil servant who had been left dysphasic and right hemiplegic too following a left hemisphere CVA in the January of 1988. After a period of intensive speech therapy, IS was discharged to a nursing home 4 months later in May, where she remained up until Marshall et al. (1990) conducted their study. Like RS IS presented with good comprehension of simple conversation, but her deficits seemed more likely to be semantic in nature for she had poor expressive language that was non-fluent and

150

limited to just social phrases such as greetings and salutations and high frequency nouns. IS had quite severe word retrieval difficulties and performed poorly on semantic tasks.

On Kay et al.'s (1989) word to picture matching task IS performed poorly (scoring 25/40 for spoken stimuli and 19/40 for written). On a synonym matching task with high- and low-imagery items IS scored just 24/38 for spoken presentations and 21/38 for written ones and on the Pyramids and Palm Trees test (Howard and Orchard-Lisle 1984) just 34/50. IS' poor performance on semantic tasks in conjunction with the fact that IS made 0 errors on a test when phonemic distractions were used but only 14/20 when semantic distractors were made for a diagnosis of semantic anomia, which Marshall et al. (1990) tried to treat with a semantic therapy similar to that they used with RS.

A hundred drawings were presented to IS for naming which were divided into 4 groups of 25. Group 1 was treated after a 2 week period and Group 2 after 4 weeks. Groups 3 and 4 (the control groups) remained untreated. Around 2 to 2 and a half hours was devoted to therapy in each 2 week period, and two types of picture matching tasks were used.

The scores (out of 25) for the 4 groups showed that improved naming followed treatment. IS scored 10/25 on Group 1 before treatment, and 22/25 after it. On Group 2, meanwhile, IS scored 10/25 before treatment as she did on Group 1, and after treatment this score increased to 23/25. After the first 2 weeks, IS performed significantly better on Group 1, which she'd received treatment for than she did Group 2, which she hadn't ( $x^2 = 6.10$ , p < 0.02). Following the second period of treatment (after 4 weeks) on Group 2, IS performed significantly better on this than she did on Groups 3 and 4 ( $x^2 = 11.86$ , p < 0.01) see Table 5.3 for a breakdown of IS' scores which demonstrate the positive effect semantic therapy had on word retrieval in semantic anomia, which one would expect if semantics were a domain specific module of mind:

	Before treatment		After 2 weeks		After 4 weeks
Group 1	10	Treated	22	Not treated	20
Group 2	10	Not treated	13	Treated	23
Group 3	9	Not treated	12	Not treated	14
Group 4	7	Not treated	13	Not treated	10

Table 5.3 IS' scores (out of 25) for the four groups at successive testing (adapted from Marshall et al. 1990)

The third case (of FW, a 76 year old woman who'd suffered a left hemisphere CVA in the May of 87) was similar to that of IS. FW's speech was marked by severe word finding difficulties that were said by Marshall et al. (1990) to be due to a semantic comprehension deficit (as evidenced by semantic errors in naming (13 were produced when trying to name 120 line drawings) and poor performance on Kay et al.'s (1989) word to picture matching test (in which she scored 28/40 for auditory presentation and 23/40 for visual).

During therapy, FW was given 50 drawings but was only able to name 17 of them. These were divided into a treatment group and a control group (9/25 and 8/25 of the correctly named items in each of them, respectively). Picture matching tasks, though, administered for around 3 and a half hours each week over a 3 week period showed that unlike IS', FW's semantic impairment couldn't be treated by semantic therapy. A slight improvement was seen in naming, for the treated group (from 8/25 correct to 11/25 correct), but this result was not a significant one.

Marshall et al. (1990) also conducted a group study with RS, IS and FW which made use of a similar therapeutic task, but was required to be carried out in their own homes with a relative or volunteer, as opposed to a therapist, assisting. Semantically treated items showed significant gains when compared with controls, which is interesting, but since no therapist was present at the time of testing and testing was not recorded I've decided not to include the results of the group studies in my analysis, for since the accurate administration of treatment can't be guaranteed, there is nothing to say that quality was controlled.

## 4.2.1.1.2 Marshall et al. (1998)

Marshall et al. (1998) reported on four patients diagnosed with jargon aphasia (RMM, TD, JC and CM). Of these four patients, all had problems with word retrieval as was evidenced by their respective scores on naming tests (0/40, 7/40, 7/40 and 14/40). A diagnosis of deficit

was only given for one of the patients (CM), though, and so I won't be including the results of the others' treatments in my analysis.

CM's speech, Marshall et al. (1998) said, largely consisted of neologistic jargon which was, for the most part, incomprehensible. Social phrases and the occasional relevant noun or noun phrase were produced, and he was better at reading aloud than he was repeating. CM showed an awareness of his speech and language disorder and, recognising that his wife was struggling to understand him, was keen to receive therapy. Nevertheless, when it came to talking, CM seemed not to have good comprehension. Unsure whether he was making sense or not during conversational discourse, he often looked to his conversational partner for confirmation, but his poor comprehension was confirmed by his scores on a series of tasks (a minimal pair discrimination requiring picture selection task (37/40), an auditory lexical decision test (127/160), spoken and written word to picture matching tests (39/40 and 32/40) and a synonym matching test (47/60: 27/30 for high image words and a lower 20/30 for low image words).

In order to determine the exact nature of CM's deficit, Marshall et al. (1998) asked him to name or repeat items and judge the correctness of his responses by saying *yes* or *no* or by pointing at cards displaying either a tick or a cross. On experiment 1 (a naming experiment) CM was asked to name 40 pictures from the PALPA test and judge whether or not his responses were correct. CM often produced more than one response per picture, which is why there are 49 judgements included in the data. His total score of correct judgements was only 32/49 with 14 of those being a perfect score on his correct responses, but of the 35 incorrect responses that CM had to judge, he incorrectly did so for half (scoring 6/13 on his verbal paraphasias, 11/20 on his neologisms, 1/1 on his phonological errors and 0/1 on one that was labelled by Marshall et al. (1998) as 'other').

It was found that CM was better at judging repetition than naming, however, (59/66 (26/26 for correct responses, 3/3 for verbal paraphasias, 19/20 for neologisms and 11/17 for phonological errors)), which indicated to Marshall et al. (1998) that CM's problems more than likely had to do with access to semantics or from semantics to the phonological output lexicon than they did something post-lexical (for repetition doesn't make use of the two stages of word production that naming does: the need to access semantics or the need to access phonology from semantics).

A picture selection task (experiment 3) was used to identify which impaired naming process it was that impacted CM's monitoring. In the task, CM was shown two semantically related pictures and heard one of their names, and was required to point to the picture that matched the name heard. Marshall et al. (1998) hypothesised that if it was the impaired retrieval of semantic representations that disrupted CM's monitoring, a decline in awareness should be observed, but if, on the other hand, it was the accessing of phonology from semantics that was impaired, monitoring should be preserved, actually.

40 items in the task stimulated 48 responses, and of the whole 48, CM made only one error when judging his picture selection responses. With regards to the responses themselves, most of his failures occurred with phonological errors as opposed to semantic ones. These results, coupled with those of experiment 3 brought about a diagnosis of an anomia that was due to damage to access to the phonological output lexicon for CM, which Marshall et al. (1998) tried to treat with semantic therapy tasks in experiments 5 and 6.

In experiment 5, CM was shown two semantically related pictures (for example of a hammer and a nail) and asked to point to one of the pictures upon hearing its spoken name before trying to name the other (i.e. pointing to the picture of the hammer when hearing the word *hammer* before trying to name the picture of the nail). It was expected that this task might assist naming by priming the relevant semantic field and increasing activation to phonology from the semantic lexicon, but results showed that this wasn't the case (see his pre-therapy scores out of 40 compared with his post-therapy scores out of 40 in Table 5.4):

Responses	Naming	
	Unprimed	Primed
Correct	6/40	6/40
Semantic errors	3/40	5/40
Phonological errors	6/40	6/40
Verbal paraphasias	12/40	10/40
Neologisms	12/40	11/40
No response	1/40	2/40

Table 5.4 Unprimed and primed naming in CM (adapted from Marshall et al. 1998: 99)

In experiment 6, CM was required to associate the concepts of target pictures with those of semantically related written words before selecting the name of the picture from four options

(the target itself and three semantic distractors). Six one hour therapy sessions were provided, and the efficacy of it was evaluated by asking CM to name the same 40 pictures before and after therapy. An untreated control group of pictures (which were matched for frequency) was tested in the same way, but therapy was found to have no effect on naming (in fact, CM's performance at naming both sets of words declined over the therapy period as the results in Table 5.5 show):

Responses	Controls		Treated items	
	Pre-therapy	Post-therapy	Pre-therapy	Post-therapy
Correct	14/40	7/40	13/40	11/40
Phonological	8/40	6/40	6/40	7/40
error				
Semantic error	8/40	8/40	6/40	3/40
Verbal	4/40	6/40	2/40	7/40
paraphasia				
Neologism	6/40	13/40	13/40	12/40

Table 5.5 Pre- and post-therapy naming of treated and control items (adapted from Marshall et al. 1998: 100)

In short, then, neither of the semantic therapies seemed to be effective at treating CM's phonological anomia. This is, of course, of interest to this thesis, because it's what a massively modular model of phonology would predict.

## 4.2.1.1.3 Nettleton and Lesser (1991)

In Nettleton and Lesser's (1991) study of anomia therapy, six aphasic patients with naming difficulties were selected according to the following criteria (Nettleton and Lesser 1991: 142):

- (a) diagnosed as aphasic on the Boston Diagnostic Aphasia Examination (BDAE)
   (Goodglass and Kaplan 1983);
- (b) at least six months after a single cerebrovascular accident, i.e. well past the period of substantial spontaneous recovery;
- (c) with good hearing and visual acuity;
- (d) without dysarthric difficulties;
- (e) within the age range 50-70 years;
- (f) with a naming difficulty, as assessed on the Boston Naming Test (BNT) (Goodglass and Kaplan 1983).

Two subjects were diagnosed as having problems with the semantic system, two were diagnosed as having problems relating to the phonological lexicon and two with problems at the level of phoneme assembly. After Lesser (1989), the criteria used to distinguish the three types of naming disorder were as follows:

- (1) For a naming disorder relating to the semantic system (Nickels et al. 1991: 142-143):
  - Performance on the auditory word-picture matching test of semantic ability from Psycholinguistic Assessments of Language Processing in Aphasia (PALPA) (Kay *et al.* 1991) fell below 39/40, i.e. the cut-off level for normal performance. [...]
  - ii. The majority of errors on the BNT consisted of semantic paraphasias.
  - iii. When cued by a phonemic cue for a semantic associate of a target word, the subject generally produced the cued associate and accepted it as correct for the target.
- (2) For a naming disorder relating to the phonological output lexicon (Nickels et al. 1991: 143):
  - i. The score in the auditory word-picture matching subtest from PALPA was at least 39/40.
  - ii. The majority of errors on the BNT consisted of anomic circumlocutions. [...]
  - iii. The percentile for the repetition subtests in the BDAE was at or above the percentile for the auditory and comprehension subtests.
- (3) For a naming disorder relating to the phonological output buffer (Nickels et al. 1991: 143):
  - i. The score on the auditory word-picture matching subtest from PALPA was at least 39/40.
  - ii. The majority of errors on the BNT were phonemic paraphasias [...].
  - iii. Repetition percentile on the BDAE was below the percentile for auditory comprehension.

Two subjects were studied in each of the three categories. Two men, PD and FF, met the criteria for a naming disorder relating to the semantic system and two women, DF and MC, met the criteria for a naming disorder relating the phonological output lexicon. For a naming disorder relating the phonological output buffer another two women, MH and NC, met the

criteria, but since the latter two patients' problems had to do with assembly and not access, I shall be excluding them from my analysis, and so no more will be said about them here.

PD was a 55 year old car park attendant with fluent aphasia following a cerebral infarct six months prior in the left parietal area. FF was a 68 year old former labourer who had a fluent aphasia that was marked by empty speech and poor auditory comprehension. DF was a 63 year old housewife who'd been left with a moderate aphasia following a stroke the year before and had relatively good social communication skills and MC a 57 year old housewife whose eight year ago stroke due to an embolism following mitral stenosis left her with non-fluent, agrammatic and dysprosodic speech as can be seen in the patient profiles in Table 5.6:

Subjects	
Naming disorder related to the semantic system	tem
PD	Male, aged 55, car park attendant
	Six months post stroke, with fluent aphasia
	BDAE comprehension 59%
	PALPA comprehension 75% (nine close
	semantic errors)
	BNT 20% (+17% with a semantic cue,
	+10% with a phonemic cue)
	On PTM pictures accepted most cued
	semantic associates as correct
FF	Male, aged 68, retired labourer
	Three years post stroke, with fluent aphasia,
	hemiplegia and hemianopia
	BDAE comprehension 15%
	PALPA comprehension 62% (12 close
	semantic errors)
	BNT 12% ( $+7\%$ with a semantic cue, $+0\%$
	with phonemic cue)
	On PTM pictures accepted all cued
	semantic associates as correct
Naming disorder related to the phonological	lexicon
DF	Female, aged 63, housewife
	One year post stroke, with severe anomic
	aphasia
	BDAE comprehension 90%
	PALPA comprehension 97%
	BNT 10% (with +0% with a semantic cue,
	+0% with a phonemic cue)
MC	Female, aged 57, housewife
	Eight years post stroke, with non-fluent
	agrammatic aphasia
	BDAE comprehension 50%
	PALPA comprehension 97%
	BNT 19% (+0% with a semantic cue, +25%
	with a phonemic cue)

Table 5.6 Subjects (adapted from Nettleton and Lesser 1991: 145)

The materials to be used in therapy were taken from the Cambridge pictures (Howard et al. 1985) (which are 300 drawings of common objects). These pictures were given the name Set A, to distinguish them from the subsets AT, AU and AUS. Each subject was asked to name the 300 pictures in Set A, and from the items they were unable to name, a random selection of 100 was made (meaning that each subjects' subsets were different). For each subject's 100 drawing subset, the 100 items were randomly split down the middle to make a further two sets of 50, a set to be treated (AT) and a set to remain untreated (AU). The remaining 200 items from Set A for each patient acted as a control group, and this was given the label AUS (A

UnSeen), so Set A (300) comprised three subsets of 50, 50 and 200, AT, AU and AUS, respectively.

The next eight weeks constituted the therapy phase; each subject attended clinic for an hour of therapy twice a week during this period, whereby either semantic naming therapy or phonological therapy was given. Each subject experienced only one type of therapy though, with the two patients diagnosed with semantic anomia (PD and FF) receiving model appropriate semantic therapy and the two patients diagnosed with phonological anomia (DF and MC) model appropriate phonological therapy. Patients' abilities to name the items in Set A post-therapy were compared with their ability to name them pre-therapy to measure their effects. The semantic and phonological therapies were found to largely work, with PD seeing a significant improvement at naming AT (the treated items) after the therapy phase p < 0.05 ( $x^2 = 4.4.840$ ) but not the untreated items or the unseen items (AU or AUS).

DF scored significantly better on AT after the therapy phase than she did before it ( $x^2 = 5.121$ ; p = 0.025) but not AU or AUS too, and just like the others MC saw significant improvement in her AT scores post-therapy ( $x^2 = 6.428$ ; p = 0.01) but no significant changes in her AU or AUS ones. When it came to FF's scores, however, although he saw some improvement in naming the AT items post-therapy (going from 2 pre-therapy to 6 post-), this result was not statistically significant. It can be said, then, that whilst semantic therapy facilitated word retrieval in one patient with semantic anomia (PD) which is consistent with the thesis that semantics is a domain specific module of the mind, it did not do the same for the other (FF). Phonological therapy, meanwhile, saw naming improve in both patients with phonological anomia (DF and MC). This, of course, is what one would expect if phonology were domain specific and therefore massively modular.

# 4.2.1.2 Semantic and phonological tasks 4.2.1.2.1 Best et al. (1997)

In 1997 Best et al. reported the case of a man with dysphagia following a stroke. He presented with severely impaired word finding abilities, and so they intervened with a number of treatments, including a case study that compared the effects of lexical (both semantic and phonological treatment) with phonological treatment and a case study that looked at whether naming could be treated with a phonological cueing aid (see Bruce and Howard 1987). A set of pilot treatments were conducted prior to the two treatment studies, but since the methods of

these weren't reported in any great depth, I've made the decision to exclude them from my analysis.

The patient, JOW, was 62 years old at the time of the study. He was a former entrepreneur, and had suffered a left hemisphere stroke in the September of 1990 (though there was no scan information to specify the exact site of his lesion). JOW had had speech and language therapy from one year post-trauma (the September of 91), when he also began to attend a local stroke support group for people with dysphagia. Testing on the Western Aphasia Battery classified JOW as an anomic aphasic, with a quotient of 69.

JOW performed within normal limits on a number of input processing tasks such as auditory discrimination and auditory lexical decision. Semantic processing of concrete items was also good (JOW scored 95% on both the auditory and written versions of word to picture matching tests with semantic and visual distractors (n = 40) (Kay et al. 1992)) and on the three picture version of the Pyramids and Palm Trees test (Howard and Patterson 1992) JOW's score was within normal limits.

When asked by Best et al. (1997) to describe the goings on in a picture, JOW said the following (where the words in the parentheses represent the intended targets determined on the basis of pointing, and the words in the square parentheses the therapists' prompt question):

'A tree, a plag, am er kite there's bun and a girl and a boy and what do you call it? A bicycle, no sorry what's it called? (dog) A garage and um a bais oh (radio) no not a bicycle. [What are they doing?] They're sat on a bike and pouring out something. He's reading a book. Is that enough?

Word finding difficulties were clearly obvious in JOW's pre-treatment speech, then, but further testing confirmed this to be the case. On a picture naming task, it was evident JOW's performance was impaired; he was only able to name 28% of a set of 194 pictures correctly. Interestingly, he was better at naming items that were high in operativity than those that weren't (Rank sum test, z = 2.00, P = 0.05) and those that were rated high in imageability/concreteness than those that rated low on those measures (Rank sum test, z = 2.29, p = 0.025). These findings, together with the fact that JOW's naming accuracy was not significantly affected by animacy, frequency, familiarity, age of acquisition, number of syllables or number of phonemes pointed Best et al. (1997) to a diagnosis of a breakdown in language processing at the lexical-semantic level. His error responses suggested the same thing, they said, with his main error types being of no response (34% of the time) or the provision of semantically related information (11% of it). Only 2% of JOW's responses were phonologically related to the target, which Best et al. (1997) took to be evidence against a phoneme assembly deficit. The fact that JOW's naming wasn't influenced by variables such as frequency or length led them to conclude that his breakdown couldn't be anything other than semantic, for those variables are generally held to reflect processing beyond that.

The effect of lexical therapy (both semantic and phonological) was considered first in study 1, but since this thesis is concerned with the effects of separate semantic therapies and phonological therapies only, I won't be including the lexical therapy part to study 1 in my analysis. The second part to study 1 was concerned with the effects of semantic therapy, however, which is something I definitely will be paying attention to.

Provided with a set of pictures (n = 36) that had a picture of the target on the face of the card and four semantically related names written on the reverse, JOW's task for the semantic therapy part to study 1 was to underline the target name and write it below the picture (e.g. when the target was *apple*, a picture of an apple appeared on the front of the picture card, and the words *apple*, *banana*, *orange* and *pear* on the back, requiring JOW to underline the written word *apple* on the back of the card and write it beneath the picture on the front).

After this, the effects of phonological treatment was considered in study 2. 50 items JOW found hard to name were selected for therapy, and matched in difficulty to 50 control items. The control items were used to detect whether any improvement found on treated items would generalise to untreated ones, and treatment took place for an hour a week, over five weeks using an electronic cueing aid that had nine keys labelled with letters that produced their corresponding phoneme (plus schwa) when pressed. JOW was instructed to use the aid every time he was unable to name a picture, and interestingly enough the phonological treatment with a cueing aid resulted in a highly significant improvement (from 8% targets being correctly named pre-therapy to 40% post- (z = 3.86, p = 0.001)) which generalised to naming of untreated items (z = 2.05, p < 0.05). Similarly significant results were seen with the

semantic treatment with 45 targets being named correctly post-treatment from 21 pretreatment evidencing improved naming.

This is interesting to a thesis of modularity, as if semantics was a domain specific massive module then we would only expect the semantic therapy to have had an effect on naming in semantic anomia and, of course, both the semantic and the phonological therapy were found to have had. This could count as evidence against the semantic processor's domain specificity, then, and by extension its massive modularity.

#### 4.2.1.2.2 Hillis (1989)

In 1989, Hillis looked at the efficacy of treatment at reducing aphasic naming errors too. Two severely aphasic patients who made frequent semantic errors in both spoken and written naming were studied, but contrasting patterns of errors across a variety of tasks revealed that their naming errors arose from different deficits underlyingly. Whilst patient 1's errors were said to be caused by a single, semantic deficit, patient 2's errors were said to have two sources (an inability to retrieve the correct phonological representations from the lexicon for spoken naming and an inability to maintain the graphemic representation of the word when it was being written). Because patient 2's errors are caused by a mixed deficit, I shan't be including the results of their treatment in the analysis. The results of treatment for patient 1 will still be considered, however, and so we'll look at their case report in more detail in what follows.

Patient 1 was a 51 year old, college educated male who'd suffered a thromboembolic stroke three months prior to the investigation. A computerised tomography (CT) scan revealed that this was due to infarction in the left fronto-parietal area of his brain, and a right hemiplegia followed (which resolved over the course of the next two weeks to a mild paresis of the right arm only). Patient 1's ambulation improved to a normal gait but apraxia persisted, and by the time he'd reached seven weeks post-onset it was clear that his expression and comprehension of language was compromised. On the Minnesota Test for Differential Diagnosis of Aphasia (which tests comprehension), patient 1 was able to point to named objects with only 30% accuracy, his yes or no responses to acontextual questions were only at chance level of accuracy and his reading comprehension was severely impaired too. Most errors made on word to picture matching tasks were semantic confusions (e.g. *dog* for *cat*), and when tested on a total of 144 items with a series of tasks (spoken naming, written naming, reading aloud, writing to dictation, spoken word to picture matching and written word to picture matching)

around 40% of his responses for each were errors (most of which were semantic as can be seen in Table 5.7):

Task	Error rate (%)	Semantic errors (%)
Spoken word to picture matching	42.4	40.3
Written word to picture matching	36.8	27.1
Spoken naming	44.4	40.9
Written naming	46.5	34.7
Writing to dictation	41.7	27.7
Reading aloud	41.7	36.1

Table 5.7 Patient 1's performance on lexical tasks (where the stimuli for each task was 144 names for or pictures of concepts)

Patient 1's homogenous pattern of semantic errors across all six tasks, Hillis (1989) said, indicated a unitary impairment in semantics – an impairment which went on to be treated with phonological therapy.

A set of 50 black and white line drawings of objects and events that were familiar to the patient served as stimuli in the phonological therapy. Patient 1 was asked to name all 50 stimuli in three consecutive sessions, and the ten he was unable to name each time were selected for treatment. The remaining 40 were used as controls, to measure the efficacy of the treatment. Change in performance on naming across the spoken and written modalities was evaluated. Patient 1 was presented with a random picture from the stimulus set (of ten items) followed by the following hierarchy of cues (adapted from Hillis 1989: 634) until the target was written accurately:

- 6. Pictured stimulus (independent written name).
- 5. Scrambled anagram and two distractors.
- 4. Scrambled anagram without distractors.
- 3. Initial letter cue.
- 2. Spoken name (correct writing to dictation).
- 1. Written name presented briefly (correct delayed reproduction).

The subject was then asked to say the word, but no feedback was given about whether they had done either of those things rightly or wrongly. The percentages of accurately written and spoken names produced in response to the picture stimuli were recorded, and results showed that not only did both oral and written naming improve significantly with phonological treatment on trained words, but that this generalised to untrained words as well. This, of course, isn't consistent with what a massively modular model of semantics would predict. If semantics were massively modular, then it would have to be domain specific, and, as I've said before, if semantics were domain specific, then it wouldn't be able to operate on input irrelevant to its function as it does here with phonological information.

#### 4.2.1.2.3 Hillis and Caramazza (1994)

Hillis and Caramazza, meanwhile, present a series of single subject studies of brain-damaged patients who each make semantic errors in naming. A cognitive analysis of each case's performance provided evidence to propose relatively selective impairment to components of lexical processing, but whilst Hillis and Caramazza diagnosed cases (KE and JJ) as having one primary deficit (a semantic impairment) they diagnosed two others (HW and SJD) as having word retrieval impairments that were due to damage to two levels of lexical processing (at the level of one modality specific output lexicon (the phonological for HW and the orthographic for SJD) and to sublexical procedures for converting sound to print and print to sound). PM, on the other hand, who was their fifth patient, was diagnosed with substantial impairments in accessing information from both the phonological and orthographic lexicons and in using orthography to phonology and phonology to orthography procedures.

Neither the results from HW and SJD's cases nor the results from PM's will be included in my own data analysis. Attention will be paid to KE and JJ, however, and so I describe their diagnoses and treatments here. KE was a 51 year old, well educated (to college level) male who suffered a left fronto-parietal infarct six months prior to the study. His identical pattern of performance across lexical tasks was interpreted as evidence for selective impairment to the semantic system. A set of 144 items were presented to him for spoken naming, written naming, writing to dictation, reading aloud, spoken word to picture matching and written word to picture matching and of all the incorrect responses that KE made, nearly all were semantic.

A set of 50 drawings of familiar objects (to test nouns) and 10 drawings of familiar actions (to test verbs) was presented for consecutive oral and written naming without feedback. When the

patient was unable to name the item, a hierarchy of phonological cues were provided. Following a correct response, each of the cues were provided again but in reverse order, until the patient was able to produce the name for the item independently when looking at the picture, and results showed that treatment with the phonological cueing strategy resulted in improved naming of both trained and untrained items for KE.

JJ, on the other hand, was a 67 year old ex-corporate executive who sustained a left temporal and basal ganglia stroke 9 months before he began treatment with Hillis and Caramazza (1994). Like KE, JJ had a college education (2 years) and also like KE, he showed selective damage to the semantic component of lexical processing. On the lexical tasks that were presented to KE, JJ also showed frequent semantic errors, though whilst KE showed semantic errors across all categories of words, KE showed semantic errors across all categories of words but animals. According to Hillis and Caramazza (1994) this could be explained by assuming selective damage to the semantic system that selectively spared semantic representations. Why or how that would occur they didn't know, but they did stress that unlike PM, his ability to use sublexical sound to letter and letter to sound mechanisms was spared. The effects on spoken naming performance on two different types of treatment were compared, a semantic word to picture matching treatment and a phonological cued reading treatment. Results of the two treatment approaches showed that whilst the semantic treatment facilitated JJ's ability to name items (of the 20 pictures that were presented to JJ for naming an average of one or two (and therefore 10-20% of) items he was unable to name before treatment he was able to following it) a change in picture naming performance wasn't seen for JJ with the phonological treatment.

The fact that JJ has semantic anomia which was treated by semantic therapy but not phonological therapy suggests, to my mind, that semantics is indeed domain specific (and so massively modular). KE's result, on the other hand, was inconsistent with a thesis of massive modularity, for word retrieval in her semantic anomia was facilitated by phonological therapy. This suggests that the information semantics operated on, in this case, wasn't relevant to its function.

## 4.2.1.2.4 Pederson et al. (2001)

Pederson et al. (2001) investigated the effects of computerised rehabilitation of anomia in aphasia with three single case studies. The therapy was carried out in the patients' own homes without the presence of a speech and language therapist, and it wasn't overseen by any other
clinician, family member or friend either. Just like I excluded Marshall et al. (1990)'s unsupervised group study from my analysis, then, so too will I exclude this one. I also exclude this one because the patients were selected from the neurological departments of the Bispebjerg and Hvidovre hospitals in Copenhagen, Denmark and through the Danish Stroke and Aphasia Association, meaning that the Western Aphasia Battery (WAB) (Kertesz 1982) and Psycholinguistic Assessment of Language Processing Abilities (PALPA) (Kay et al. 1992) tests used in the assessments of their anomias had to be translated from English and adapted for Danish for use in this study.

Whilst I've listed Pederson et al. (2001) as one of the studies excluded in Table 5.2, given that my reason for doing so didn't have to do with the patients' impairments and treatments being mixed or the patients' impairments and treatment outcomes being unknown (as the rest of them do) I decided to elaborate here. No more will be said about Pederson et al. (2001) now, and so I'll turn my attention to Raymer and Ellsworth (2002).

#### 4.2.1.2.5 Raymer and Ellsworth (2002)

Raymer and Ellsworth (2002) compared the effects of semantic and phonological therapies on verb retrieval in a patient (WR) with semantic verb retrieval impairments. Accuracy of picture naming was examined on trained and untrained verbs before and after therapy, and results showed that the treatments significantly improved naming (with trained verbs showing a significant improvement and untrained verbs no improvement). The effects of a rehearsal type of treatment were also studied, but since this involved repetition of targets only it had to with post-lexical processing (i.e. phoneme assembly) it doesn't meet the criteria to be included in my analysis.

The star of the single subject investigation WR was a 54 year old woman who discontinued schooling aged 13 to become a hairdresser. A left hemisphere CVA left her with a right hemiparesis and non-fluent aphasia that was marked by word retrieval difficulties. A CT scan taken at the time of the CVA revealed a left dorsolateral frontal lesion which encompassed a portion of Broca's area (as well as the anterior insular region and subcortical white matter), so it made sense that her aphasia was one that caused problems with production. Standardised testing with the WAB (Kertesz 1982) 3 months post-stroke led to a more formal diagnosis of transcortical mortal aphasia, and results of the Bosting Naming Test (owing to Kaplan et al. 1983) and Action Naming Test (Obler and Albert 1986), respectively, indicated that her naming problems had more to do with verbs than they did nouns (naming for nouns was

affected minimally for her level of education (44/60 correct) whereas picture naming for verbs was quite a way reduced (37/62 correct)).

To evaluate her word retrieval difficulties further, WR was tested on a noun and verb battery made up of 30 verbs and two sets of 30 nouns. The battery included four tasks, the first two owing to Zingeser and Berndt (1990) and the other two to Williamson et al. (1995) which Raymer and Ellsworth (2002: 1034) described as follows:

- Picture naming: WR viewed each of the 90 black and white line drawings of the noun and verb stimuli and provided a one word label for the object or action depicted in each picture.
- (2) Sentence completion: The examiner read aloud a phrase as WR also read along. She then completed each of the 90 sentences with an appropriate noun or verb.
- (3) Crossmodal picture-to-word matching: WR pointed to one of three words that corresponded to the target picture (e.g., write). In the related condition, both distractor words were semantic coordinates of the 90 targets (e.g., read, draw), and in the unrelated condition, both distractor words were unrelated to the 90 targets (e.g., ask, take).
- (4) Picture-picture associate matching: WR pointed to one of two pictures that was most closely related to a target picture (e.g., baking a pie). The distractor pictures were semantic coordinates of the 90 correct pictures in one condition (e.g., correct—grilling a steak; distractor—peeling a potato) and were unrelated in a second condition (e.g., correct—peeling a potato; distractor—leaking water). For half of the picture associate pairs, in the unrelated condition, the distractor picture from the unrelated condition became the correct answer when a completely unrelated foil picture replaced the other associate picture.

WR's results on the battery are displayed in Table 5.8:

Task	Nouns	Verbs
Spoken picture naming	52/60	22/30
Sentence completion	52/60	24/30
Picture-picture associate match		
Related	45/60	21/30
Unrelated	57/60	39/30
Crossmodal picture-word match		
Related	56/60	28/30
Unrelated	60/60	30/30

Table 5.8 Number of items correct for WR on the Florida Noun/Verb Battery, a set of lexical tasks for noun and verb comprehension and retrieval (adapted from Raymer and Ellsworth 2002: 1035)

To determine whether her naming impairments related to the semantic or phonological level of language processing WR's performance on picture matching tasks (which rely on semantic processing for correct responses) was examined, for impairment on them would suggest that semantic dysfunction contributed to her naming difficulties. It was found that her performance was slightly below levels observed with normative subjects. This led Raymer and Ellsworth (2002) to conclude that WR's word retrieval deficits were at the semantic level of language processing, and after that, they began her treatment.

WR was asked to name 222 black and white line drawings of various actions. Responses that were the target or a synonym for the target were accepted as correct. Analysis of WR's errors in two baseline tests indicated that WR made 44-45% errors in verb naming and on the basis of this performance, 60 verbs that WR had consistent difficulty in naming were selected as experimental stimuli. The 60 items were split into three groups: 20 for use with phonological treatment, 20 for use with semantic treatment and 20 for a control set (which later became the rehearsal treatment set, though this has no place in this thesis).

WR was given two to three 1 hour sessions per week over the course of four months. In treatment phase 1, WR underwent phonological treatment for the first set of verbs and after a week-long break underwent semantic treatment with the second set. In both the phonological and the semantic treatment, WR was presented with a picture to name aloud and given feedback with regards to her response accuracy. When her response was inaccurate, she was asked two questions designed to help her develop a word retrieval strategy consistent with that of a normative subject's. In phonological treatment, the questions encouraged WR to think about the sounds of the target words (e.g. for the word *pay* she was asked an initial

phoneme question 'Does *pay* start with [p]?' and a rhyming word question 'Does *pay* sound like *way*?') and in semantic treatment, the questions encouraged her to think about the meanings of them (e.g. for *bake* the coordinate verb question 'Is this similar to *grilling*?' was asked, as was the associated noun question 'Does this have to do with *pie*?').

Results showed that when phonological treatment was applied to the first set, MR's verb naming performance rose from an average of 8.33% correct in the baseline sessions to a 90% correct criterion level, which was found to be statistically significant on a McNemar's test ( $x^2$ = 14.06, p < 0.001). This is clearly inconsistent with what a domain specific and therefore massively modular model of phonology would predict. WR also saw her verb naming performance improve with semantic treatment, though: when semantic treatment was applied to the second set an average of 17.5% correct across the baseline sessions rose to a 100% criterion level thereafter and a significant McNemar's test result was found for this too: ( $x^2$  = 13.07, p < 0.001).

# 4.2.1.3 Primarily phonological tasks 4.2.1.3.1 Fink et al. (2002)

Now, although Fink et al.'s (2002) study is to be excluded from my analysis it's worth mentioning here because my reasoning for excluding it is more complicated than that of other studies. Fink et al. (2002) assessed the benefits of a computer delivered, hierarchical phonological cueing protocol under two conditions of instruction with six different patients: (1) with full clinician guidance for GM, AS, and BM, and (2) with full independence for EL, EG and RH. Since no speech and language therapist was present during the second condition of instruction with EL, EG and RH, I've decided to exclude the results of it from my analysis, but I've also decided to exclude the results of those that were clinically guided (GM, AS and BM) too, but for reasons having to do with their deficits.

Testing revealed that GM's primary locus of impairment was at the level of phoneme assembly, which meant that he had to be excluded from the analysis because his impairment was neither semantic nor phonological. While AS and BM's primary deficits were at the access to phonological output lexicon level of language processing, however, both of them had some involvement of the semantic system meaning that their impairments were mixed and they couldn't be studied. This meant that for a number of different reasons, not one of the six patients studied in Fink et al. (2002) could be included in my analysis: GM's impairment was post-lexical, AS and BM's impairments were mixed, and when it came to EL, EG and RH regardless of their level of impairment (which were mixed as well, actually (impaired access to the phonological output lexicon and phonological encoding (that is, phoneme assembly)), a deficit in phoneme assembly and a deficit in accessing phonology from semantics, respectively), not one of them could be considered because of the study design that their therapy revolved around (they were part of a self-guided group).

## 4.2.1.3.2 Hickin et al. (2002a)

Hickin et al.'s (2002a) study set out to investigate whether the use of phonological and orthographic cues in the treatment of word finding difficulties was effective or not. The study used a case series design, and the participants involved were 8 people with acquired aphasia who were all at least a year post-onset, had a single left CVA and had word finding difficulties as a significant symptom of their aphasia. Detailed assessment of each participant was carried out to diagnose the type of aphasia they had (see Table 5.9), but a diagnosis of the impairment underlying the word retrieval difficulties with mapping between semantics and phonology, that is, a deficit with access to the phonological output lexicon) and SC (who was diagnosed with two sources of word retrieval difficulties, problems at the semantic level of language processing and problems with phoneme assembly)).

Participant	Years post-onset	Age	Aphasia type
HM	6	45	Broca's
PH	3	77	Anomic
SC	5	65	Mixed/Wernicke's
DC	5	70	Anomic
OL	2	65	Anomic
NK	3	52	Anomic
IK	3	68	Broca's
KR	8	38	Broca's

Table 5.9 Background information for the participants included in the study, including aphasia type as assessed by the Comprehensive Aphasia Test (Swinburn et al. 2004) (adapted from Hickin et al. 2002a: 986)

Since SC's impairment diagnosis was mixed and a diagnosis of impairment wasn't given for GM, OL, IK or KR, I shan't be including any of them in my data analysis. I'll be concerning myself with the results of PH, DC and NK only, who, interestingly, were said to be the three who benefited most from therapy.

The study consisted of three phases of around eight weeks in length. First, there was an assessment phase, and after that, there was a treatment phase (which focused on improving word finding by using phonological cues in a picture naming task). Progress was monitored over a number of assessments, with one at the beginning and one at the end of the study.

As part of phase 1, naming was assessed using a set of 200, black and white line drawings. Naming was tested at the beginning and the end of the assessment phase to establish baseline naming performance. Written naming was assessed using a subset of items (40 in total) taken from the set and comprehension, auditory discrimination, short-term memory, reading and repetition were assessed using the tests detailed in Table 5.10:

Test	n	Parti	cipant	S
		РН	DC	NK
Picture naming tests 1 and 2: mean	200	0.36	0.67	0.56
Semantic tests				
CAT spoken word to picture matching test	30	0.93	1.00	0.93
CAT written word to picture matching test	30	0.97	0.97	0.97
Pyramids and Palm Trees test (three picture version)	52	0.90	0.92	0.87
Picture naming: semantic errors as a proportion of total errors		0.25	0.50	0.33
Phonological tests				
ADA auditory discrimination test	40	0.68	0.85	0.90
Short-term memory test: phoneme span		2.50	2.30	2.70
Repetition of words	152	0.97	0.95	0.99
Repetition of nonwords	26	0.58	0.50	0.81
Repetition of nonwords: initial phoneme correct	26	0.88	0.85	0.96
Picture naming: phonological errors as a proportion of total errors		0.05	0.11	0.00
Reading real words	152	0.97	0.97	0.92
Reading nonwords	26	0.35	0.15	0.08
Reading nonwords: initial phoneme correct	26	0.85	0.92	0.92

Table 5.10 Participants' performances on a number of background assessments (adapted from Hickin et al. 2002a: 987)

During the treatment phase, 200 items were divided into two sets, one a treatment set and one a control one. The treated set was divided again into two sets of 50 (one to be treated with

phonological cues and the other orthographic), and assessments were made before and after treatment to test the treatments' efficacies.

As I said earlier, treatment took place for around 8 weeks, with each session lasting around one to one and a half hours. Items were presented for naming, and if they were unsuccessfully retrieved so too were the phonological or orthographic cues. In the phonological cueing therapy, the first phoneme (plus schwa) and an unrelated distractor were presented first and if the patient was still unable to retrieve the target the phonological information provided in cues increased to the first syllable and then the whole word. In the orthographic cueing therapy the same process was followed, except instead of the cues being spoken by the therapist they were written down by them.

As a result of the treatment, eight participants (including PH, DC and NK) showed a significant improvement in naming the treated items. Only one showed significant improvement on the untreated items (DC), but all three showed overall more significant improvement on the treated items than the untreated items, which is consistent with a massively modular model of phonology that requires its domain specificity. None of the participants showed a significant advantage of one cue over the other.

## 4.2.1.3.3 Hickin et al. (2002b)

Hickin et al. (2002b) investigated the effects of phonological treatment on word finding in aphasia too. They present the results of therapy from two patients: HM and PH. HM is a married male with two children in his forties who was working as a carpenter/cabinet maker when he suffered a single, left hemisphere stroke 5 years prior to his involvement in the study. Following his stroke HM and his wife separated, meaning he gets to see his children less often than he would have liked to. Living alone now in a flat, he attends a day care centre three times a week, where he's busy building a model railway. HM's other interests include steam trains and rock music, and he is said to have an expressive aphasia (that is, an aphasia that causes problems with production).

PH, meanwhile, is a seventy something year old female, who has lived in London all her life but now lives there in sheltered accommodation. PH is very sociable, engaging in bingo, lunches and day trips out to the south coast. PH has a large and supportive family that live nearby, with whom she has regular contact. Like HM, PH is said to have suffered a left hemisphere stroke three years before being involved in the study. PH's diagnosis is of

172

aphasia. Unlike HM, her expressive speech is fluent, and she frequently experiences word finding problems.

Table 5.11 shows the results of the assessments that were carried out to diagnose the specific deficits in language processing that led to HM and PH's aphasias.

Deficit	HM	РН
1. Naming 200 pictures	44	36
(number of correctly named		
items)		
2. CAT spoken word to picture	100	93
matching $(n = 30)$		
3. CAT written word to picture	87	97
matching $(n = 30)$		
4. Pyramids and Palm Trees ( <i>n</i>	94	90
= 52)		
5. Percentage of semantic	52	25
naming errors (out of total		
naming errors)		
6. Imageability effect in	N	Y
naming		
7. ADA spoken discrimination	82	68
(n = 40)		
8. Short-term memory	1.4	2.5
phonemes		
9. Repetition of words ( $n =$	73	97
152)		
10. Repetition: non-words ( $n =$	31	58
26)		
11. Repetition non-words: 1 <sup>st</sup>	54	88
phoneme correct $(n = 26)$	20	-
12. Percentage of phonemic	20	5
naming errors (out of total		
naming errors)	×7	
13. Length effect in naming	Y	N
14. Reading words $(n = 152)$	70	97
15. Reading non-words ( $n =$	0	35
26)	20	0.5
16. Reading non-words: 1 <sup>st</sup>	38	85
phoneme correct	1.(	20
17. Written naming $(n = 25 \text{ for})$	16	38
HM and 40 for PH)		

Table 5.11 Language assessment in percentages prior to therapies (from Hickin et al. 2002b: 72)

Hickin et al. (2002b) concluded of the results that whilst HM and PH both made semantic errors in naming, neither of them had semantic processing deficits. HM's difficulties with naming and repetition, coupled with the fact that his naming was affected by length led Hickin et al. (2002b) to conclude that his deficit could either have been in phonology or in access to the output lexicon, and so I will eliminate him from my own analysis because his exact diagnosis is unclear. PH, meanwhile, was said to have a deficit in access to the phonological output lexicon. Her superior performance to HM's in reading and repetition suggested to Hickin et al. (2002b) that her output phonology was intact but not easily accessed, and this, together with the fact that her semantic processing was relatively intact as well, led to a diagnosis of a phonological access deficit in the mapping of semantics to phonology.

Phase 1 of treatment (which was phonological) focused on picture naming involving a choice of cue. In the second phase of treatment, Hickin et al. (2002b) moved away from picture naming, however, instead encouraging the use of targeted words in interaction related to PH's interests. During phase 1 of treatment, 50 words were treated with phonological cues. When PH was unable to name a picture (or made an error in picture naming), she was exposed to the first sound of the word along with that one or more distractors. Targets and distractors were matched for syllable structure but were different in terms of their first sounds, as well as not being semantically related to the target, e.g. banana and piano for tornado. In phase 2, on the other hand, the emphasis was on the use of target words in everyday speech. Tasks included naming to definition, naming in a pseudo-realistic speech situation, making lists, reminiscing, telling anecdotes and conversational discourse around certain subjects. In all of the sessions, PH had access to pictures and written semantic cues if they needed them. It's clear, them, that phase 2 of PH's treatment, although semantically targeted, would involve some implicit phonological activation. For this reason, I have decided to define that phase of treatment as being mixed (semantic-phonological) and have struck it from my analysis. Results showed that PH made gains in picture naming in phase 1 (the phonological task).

#### 4.2.1.3.4 Jokel and Rochon (1998)

In Jokel and Rochon (1998), finally, treatment for an aphasic patient PD with severe naming difficulties is described. The 90 year old woman (who had suffered a left frontal haemorrhage in the July of 95)'s initial diagnosis was of global aphasia with a probable phonological output impairment, but Jokel and Rochon (1998) went on to conduct a number of assessments on to confirm whether or not this was the case.

Pictureable nouns were administered over four different tasks (confrontation naming, spoken word to picture matching, repetition and reading aloud). In the confrontation naming tasks PD's ability to name 134 items was tested three times in three separate sessions to establish a baseline naming performance for her (this was only 10% correct). Another set of 120 items were used in the word to picture matching, repetition and reading aloud tasks, and PD achieved 100% in the naming of these items. Because of the pervasive word retrieval deficit and having demonstrated intact comprehension, repetition and reading aloud performance for stimuli she was unable to retrieve in confrontation naming, it was hypothesised that she had an impaired phonological output lexicon.

Three different types of phonological treatment were provided: repetition, reading and sentence completion. The 120 (above) words were divided into four separate sets, with 30 in each of the treatment conditions (e.g. repetition (30), reading (30) and sentence completion (30)) and 30 acting as a control group. Treatment on each set was given for five days, with all three types of treatment being provided each day. Performance on a confrontation naming task which contained all 120 items was measured before and after each treatment condition, and it was found that confrontation naming pre- vs. post-repetition 2/30 vs. 2/30, pre- vs. post-reading 4/30 vs. 12/30 and pre- vs. post-sentence completion 2/30 vs. 0/30. The only significant difference found was for that of the reading treatment (Fisher's Exact test p = 0.04), and when it came to the sentence completion treatment, the post-treatment score was lower than the pre-treatment, suggesting that treatment had the opposite effect of what was wanted.

This left a total of 10 case studies made up of 18 participants and 24 results that were relevant to this thesis:

Paper	Participant	Impairment	Treatment	Outcome (treated)
Primarily se	emantic tasks	·		· · · ·
Marshall	RS	Phonological	Semantic	Yes
et al.	IS	Semantic	Semantic	Yes
(1990)	FW	Semantic	Semantic	No
Marshall	СМ	Phonological	Semantic	No
et al.			Semantic	No
(1998)				
Nettleton	PD	Semantic	Semantic	Yes
and Lesser	FF	Semantic	Semantic	No
(1991)	DF	Phonological	Phonological	Yes
	MC	Phonological	Phonological	Yes
Semantic a	nd phonologic	al tasks	1	1
Best et al.	JOW	Semantic	Semantic	Yes
(1997)			(Study 1: semantic	
			therapy	
			Phonological (Study 2:	Yes
		~ ·	cueing aid therapy)	
Hillis (1989)	Patient 1	Semantic	Phonological	Yes
Hillis and	JJ	Semantic	Semantic	Yes
Caramazza			Phonological	No
(1994)	KE	Semantic	Phonological	Yes
Raymer	WR	Semantic	Phonological	Yes
and			Semantic	Yes
Ellsworth				
(2002)		-		
Primarily p	honological ta	usks	<b>D1 1 1 1</b>	
Hickin et	PH	Phonological	Phonological	Yes
al. (2002a)	DC	Phonological	Phonological	Yes
TT' 1 '	NK	Phonological	Phonological	Yes
Hickin et al. (2002b)	PH	Phonological	Phonological (phase 1)	Yes
Jokel and	PD	Phonological	Phonological	No
Rochon			Phonological	Yes
(1998)			Phonological	No

Table 5.12 A breakdown of the 10 single case studies of word retrieval relevant to this thesis

Some others had to be struck out too though, along the way. This was a decision that was made on statistical significance grounds, and so I go on to explain that in what follows in 4.2.2.

## 4.2.2 Outcome classification (treated vs. untreated)

Now, something needs to be said about the outcome column here. In this thesis, I chose to classify therapies as having treated anomias when the patients' naming was better on a set of items post-therapy than it was pre-, and the improvement was found to be statistically significant. When patients named a set of items better after they'd received therapy for their anomia than they had before but no statistical significance was found, then the classification of not-treated was given. Rather self explanatorily, though, if the number of correctly named items stayed the same before and after therapy or if fewer items were correctly named after it than there were before the patient received therapy for their anomia, the therapy was classified as having not treated the patient.

However, whilst some researchers calculated statistical significance in this way (i.e. through tests that made use of 2x2 contingency tables to see whether there existed a relationship between the independent variable (otherwise known as the cause, which in this case was treatment (pre- and post-)) and the dependent variable (otherwise known as the effect, which was of course, in this case was outcome (number of items named correctly and incorrectly)) such as the chi-squared test or Fisher's exact test, for example (Kim 2017), others were more interested in whether the effects observed could be generalised to control items (e.g. Marshall et al. 1990), and so didn't calculate statistical significance in the way that is of interest to this thesis. When this was the case (or when statistical significance wasn't calculated at all, for that matter), I made sure to extract the data myself from the studies (well, where I was able to at least), in order to calculate the statistical significance of the effects of treatment on outcome using the Fisher's exact test calculator within the SPSS statistical package for Mac.<sup>9</sup>

#### 4.2.2.1 Marshall et al. (1990)

As I pointed out in section 4.2.1.1.1 Marshall et al. (1990) did indeed find their results for RS to be statistically significant, but the statistical significance reported had to do with whether his performance on treated items was better than that on untreated items. Now, whilst this was interesting from a qualitative perspective in that it looked like his naming improved with treatment, in order to make for a fair test I had to see whether this could be supported by statistics by running the numbers of items he did and didn't name pre-treatment and post-treatment through a Fisher's exact test in SPSS. This was easy enough to do as the authors

<sup>&</sup>lt;sup>9</sup> An explanation of why the Fisher's exact test was used is given in 4.2.3.1.

made it known to the reader that in the to-be-treated set of 25 items, only 8 were named correctly pre-treatment compared with 20/25 post-.

The effect observed was one that could be considered statistically significant; a p value of 0.0014 was produced (which is <.05 (the p value taken to be the level of statistical significance in the social sciences)) and, as such, the semantic therapy for phonological anomia, in this case, was classified as having treated it.

The results reported by Marshall et al. (1990) for patients IS and FW were again not really relevant to this study, and so I conducted Fisher's exact tests for those too to see whether there was a relationship between treatment and naming for them. Marshall et al. (1990) point out in their paper that IS' score on Group 1 items prior to therapy was 10/25 and after it 22/25, which gives us a significant result: p = 0.0009 and so the semantic therapy for IS' semantic anomia was classified treated too. It was said about FW meanwhile, that she scored 8/25 on a set of items pre-treatment and 11/25 on the same set after 3 weeks of treatment. This, as one would probably predict, produced a statistically insignificant result (of p = 0.5607), and so I classified the semantic therapy that was used to facilitate word retrieval in FW's anomia as untreated.

#### 4.2.2.2 Marshall et al. (1998)

Marshall et al. (1998), on the other hand, didn't test for statistical significance – but then again, nor did they have to. CM, who had phonological anomia, received two types of semantic treatment. In one experiment (experiment 5) semantic treatment made no difference to naming (6/40 items were named correctly pre-treatment and the same number again, 6/40, post-treatment), and in the other (experiment 6) CM was actually worse at naming after treatment than they were before (13/40 and 11/40 items were correctly named pre- and post-treatment, respectively). This led me to classify neither the first nor the second treatment as having treated CM's phonological anomia without further tests for statistical significance, as can be seen in Table 5.12.

## 4.2.2.3 Nettleton and Lesser (1991)

Nettleton and Lesser (1991) conducted their tests of statistical significance in the same way that I would have. They compared the number of items that were named pre- and post-semantic treatment by each of the semantic subjects (PD and FF), and before and after phonological treatment by the patients with phonological anomia (DF and MC). As I said in

178

section 4.2.1.1.3 results indicated that semantic treatment did indeed facilitate word retrieval in semantic anomia in PD's case (14% of the items were named correctly pre-treatment compared with 36% post-), and these results were found to be statistically significant by Nettleton and Lesser (1991) (p < 0.05 ( $x^2 = 4.840$ )) which led me to immediately classify semantic therapy as having treated semantic anomia in the case of PD.

For FF, though, whose anomia was also semantic, although some improvement was seen in naming following therapy (6% of items were named after compared with 2% before), the results did not reach significance. Thus semantic anomia was classified as having not treated semantic anomia in this case.

For DF, whose anomia was phonological, scores were significantly better after semantic therapy than they were before ( $x^2 = 5121$ ; p = 0.025). Before therapy for DF, percentage success in naming was 20% and after therapy it was 46%, so a classification of treated was given.

Last but by no means least MC, who like DF was a phonological anomic who received phonological therapy, performed significantly better after it than she did before ( $x^2 = 6.428$ ; p = 0.01). Their percentage success went from 20% to 50% before and after naming and so their therapy, like PDs and DFs, was classified as having treated anomia in this instance.

#### 4.2.2.4 Best et al. (1997)

Now, Best et al.'s (1997) report required me to calculate statistical significance for their results too, because although for JOW, a patient with semantic anomia, they found phonological therapy to significantly facilitate word retrieval (taking him from an average of 8% of items named correctly before treatment to an average of 40% (without an aid) after it (Wilcoxon, z = 3.86, p = 0.01)), when referring to the before and after results of semantic therapy, they reported that semantic therapy significantly improved naming but did specify what p value led them to come to that conclusion.

Statistically significant results for phonological therapy's (p = 0.0003) and semantic therapy's (p = 0.0178) effects on naming were found through Fisher's exact tests (after the percentages were put back to raw token numbers, that is) by myself, however. And so I made sure to classify the two as having treated the patients' anomias.

#### 4.2.2.5 Hillis (1989)

It was when I looked into the statistical significance of Hillis (1989) that I really ran into problems. Not only did Hillis not report raw token numbers for items named before and after treatment, the percentages provided were done so on two figures which were difficult to decipher. As such, I made the decision to exclude this study from my own, which is why it's missing from Table 5.12. I couldn't conclude anything about how many words had been retrieved at each point with any accuracy at all, and I didn't want to run the risk of any mistakes being made interpreting their results impacting my own.

## 4.2.2.6 Hillis and Caramazza (1994)

The semantic therapy that was used to facilitate word retrieval in JJ's semantic anomia was classified as having successfully treated it, whereas the phonological therapy that was used was not. As Hillis did in her 1989 study, though, Hillis and Caramazza (1994) depicted JJ and KE's performances pictorially. Percentages of correctly named nouns were plotted, but those percentages weren't (for the most part) provided (and nor were the raw token numbers that those percentages were figured out from).

Hillis and Caramazza (1994) did say, however, that phonological treatment did not facilitate word retrieval in JJ who had semantic anomia, and so I was automatically able to put that in the untreated pile. Semantic therapy, on the other hand, was said to improve naming in JJ though and a p value of <0.02 was reported, and so I classified that as having treated JJ's anomia. Improvement in naming in KE was seen by Hillis and Caramazza (1994), but since no statistical tests were conducted by them nor were percentages/numbers reported to the reader that I could have used to carry out tests of my own, effects for KE were excluded.

#### 4.2.2.7 Raymer and Ellsworth (2002)

Raymer and Ellsworth's (2002) study didn't need any intervention. Only one participant was involved (MR), and both the therapies she had (semantic and phonological) were found to facilitate word retrieval in her semantic anomia. WR saw her naming performance significantly improve from 17.5% correct to 100% correct with phonological treatment (p < 0.001). Naming performance rose from an average of 8.33% of words named correctly pretreatment to a 90% correct criterion level for semantic treatment. This was found by Raymer and Ellsworth (2002) to be statistically significant also (again, p < 0.001), and so the semantic therapy was classified by myself as having treated her semantic anomia too.

#### 4.2.2.8 Hickin et al. (2002a)

Hickin et al. (2002a), who studied the effects of phonological treatment on phonological anomia in three participants (PH, DC and NK) reported statistically significant results for all. PH saw an improvement to 55% correctly named items of 200 post-treatment from an average of 35% pre-, DC an improvement from 72% of 200 items correctly to 89% post-treatment and NK from 57% to 71% post-treatment. All 3 results were said to have generated *p* values of <0.01, and so the therapies were all classified as having treated phonological anomia in this thesis.

## 4.2.2.9 Hickin et al. (2002b)

In Hickin et al. (2002b), PH's anomia was studied once more. They received phonological treatment. n equalled 50, and PH was able to name items an average of 34% of the time before phonological treatment compared with 60% of the items after it was administered. A Fisher's exact test result comparing the 17/50 words named before treatment and 33/50 words after treatment (calculated by myself) ruled the increase statistically significant with a p value of 0.0025.

## 4.2.2.10 Jokel and Rochon (1998)

The final report's results to be reviewed here, are that of Jokel and Rochon's (1998) who compared the efficacies of three phonological therapies at treating phonological anomia in a participant called PD. 120 words were divided into 4 sets, and 3 of those were treated with repetition, reading and sentence completion. Scores for naming were reported as 0/30 vs. 2/30 pre- vs. post-repetition, 4/30 vs. 12/30 pre- vs. post-reading and completion 2/30 vs. 0/30 pre- vs. post-sentence completion. With the third phonological treatment, then, a negative effect on naming was observed and so this was classified as not having treated PD's phonological anomia. With the first two (repetition and reading) positive effects were observed, but the reading treatment was the only one a significant difference in naming (Fisher's Exact test p = 0.04) was found for, and so whilst the second treatment was classified as having treated PD, like the third treatment, the first was not.

This left 9 case studies made up of 16 participants and 22 results to be quantitatively analysed:

Paper	Participant	Impairment	Treatment	Outcome (treated)	
Primarily se	Primarily semantic tasks				
Marshall	RS	Phonological	Semantic	Yes	
et al.	IS	Semantic	Semantic	Yes	
(1990)	FW	Semantic	Semantic	No	
Marshall	СМ	Phonological	Semantic	No	
et al.			Semantic	No	
(1998)					
Nettleton	PD	Semantic	Semantic	Yes	
and Lesser	FF	Semantic	Semantic	No	
(1991)	DF	Phonological	Phonological	Yes	
	MC	Phonological	Phonological	Yes	
Semantic at	nd phonologic	al tasks			
Best et al.	JOW	Semantic	Semantic	Yes	
(1997)			(Study 1: semantic		
			therapy		
			Phonological (Study 2:	Yes	
			cueing aid therapy)		
Hillis and	JJ	Semantic	Semantic	Yes	
Caramazza			Phonological	No	
(1774) Roymer	WP	Semantic	Phonological	Vas	
and	W IX	Semantic	Semantic	Vec	
Fllsworth			Semantic	105	
(2002)					
Primarily p	honological ta	usks			
Hickin et	PH	Phonological	Phonological	Yes	
al. (2002a)	DC	Phonological	Phonological	Yes	
	NK	Phonological	Phonological	Yes	
Hickin et	РН	Phonological	Phonological (phase 1)	Yes	
al. (2002b)					
Jokel and	PD	Phonological	Phonological	No	
Rochon			Phonological	Yes	
(1998)			Phonological	No	

Table 5.13 A breakdown of the 9 single case studies of word retrieval that I've included in my analysis

# 4.2.3 A quantitative analysis of the case studies

# 4.2.3.1 Statistical significance testing

We can, of course, determine whether relationships exist between independent variables (otherwise known as causes) and dependent variables (otherwise known as effects) by testing for statistical significance. In data sets such as mine, where the distribution of one categorical variable (in this case, treatment type) is to be compared with the distribution of another categorical variable (e.g. like mine, treatment outcome) either the chi-squared test or Fisher's exact test tend to be used (Kim 2017).

Now in order to determine whether a statistically significant relationship existed between treatment type (the independent variable) and treatment outcome (the dependent variable) in this thesis, a couple of tests (one looking at the relationship between treatment type and treatment outcome in patients with phonological anomia and another between treatment type and treatment outcome in patients with an anomia that was semantic) had to be conducted. The chi-squared test, however, should only be used in cases of large sample sizes, for it applies an approximation and mine (n=22) was definitely not that. Fisher's exact test, on the other hand, runs a procedure that works for small sized samples too (Kim 2017), and so that was the one I used.

I calculated the statistical significance of the effect of treatment type on treatment outcome in both phonological and semantic anomia with two Fisher's exact tests using the SPSS statistical package for Mac; results and discussion can be found in the following section.

# 4.2.3.2 Results and discussion

Of the 9 studies I decided to include in my analysis, 22 different results for the effects of therapy type on anomia type were elicited. Results showed that phonological anomia could be treated with phonological therapy 7 times out of 9 (or 78%) of the time and semantic therapy 1 out of 3 (33%):

Therapy type	Therapy result		
	Treated	Not treated	
Semantic therapy	1	2	
Phonological therapy	7	2	

Table 5.14 The effects of semantic therapy and phonological therapy at treating phonological anomia in the sample

Semantic anomia, on the other hand, could be treated with semantic therapy 71% of the time (5 times out of 7) and phonological therapy 67% (2 out of 3 times):

Therapy type	Therapy result		
	Treated	Not treated	
Semantic therapy	5	2	
Phonological therapy	2	1	

Table 5.15 The effects of semantic therapy and phonological therapy at treating semantic anomia in the sample

This is interesting at first glance, because although phonological therapy looks to be much more effective than semantic therapy at treating phonological anomia, the fact that semantic therapy has a successful effect 33% of the time is indicative that the phonological processor is able to act on semantic input after all (a third of the time, in fact), which would suggest that phonology is neither domain specific nor massively modular.

Fisher's exact tests which considered the effects the two therapy types had on the treatment outcome in patients with semantic and phonological anomia showed that the above results weren't statistically significant, however. The results for patients with phonological were p = 0.2364, which, being >0.05 (the *p* value taken to be the level of statistical significance in the social sciences), counts as weak evidence against the null hypothesis, and so it is unlikely that there exists a relationship between treatment type and treatment outcome in patients with phonological anomia.

The p value for the outcomes of therapy types happening by chance in people with semantic anomia was even greater at 1, which means that the result for people with semantic anomia is even less statistically significant than the one for people with phonological anomia is. For patients with semantic anomia, it is even less likely that treatment type and treatment outcome are related than they are for phonological anomia.

Now, the reason I studied the efficacies of semantic and phonological therapies at treating semantic anomia when this thesis questions whether phonology can be considered a module of the mind is because, if both semantics and phonology were part of the grammar, we would

expect them to pattern the same way. Noel Burton-Roberts (2011) argues that phonology, in being grounded in phonetics, is neither modular nor part of the grammar (while semantics is), but since a treatment type/treatment outcome relationship was found for neither phonology nor semantics, this is indicative, to my mind, that whatever their statuses are, modular or ammodular, it's likely that they're the same – either both of them are parts of a modular grammar or neither of them are parts of a modular grammar.

However, there are a number of variables at play here that have the ability to affect the statistical significance of these results. The size of the sample is the one that first comes to mind, but word frequency and length and orthography are known, in the aphasiology literature, to also affect naming and so I consider those in 4.2.3.2.1. It's also important to consider the effects that publication bias (if there's been any) could have on the conclusion I come to in this thesis, and so I will do that too in section 4.2.3.2.2.

# 4.2.3.2.1 The effects on statistical significance

## 4.2.3.2.1.1 Sample size

When constructing a sample, one should really consider the number of results necessary to achieve representativeness of the sample in the wider context, but agreeing on the optimum number of subjects that guarantees this representativeness has always been a bit of an issue for linguists (Silva-Corvalán 2001). In early work, it was thought that linguistic behaviour was relatively homogenous so small samples would be sufficient to measure correlation (e.g. Labov (1966: 181) proposed that '10-20 instances of a given variable is sufficient to assign a value that fits into a complete matrix'), but more recently people (e.g. Milroy (1987)) have proposed that sample sizes this small are insufficient with Milroy (1987) arguing that n=30 should be taken as the watershed between sufficient and insufficient samples.

Of course, it should go without saying that the smaller the sample size the less sure we can be about the accuracy of the conclusions we draw from that sample, but as Milroy and Gordon (2003) rightly point out, sample sizes are partly dictated by practicalities (as mine was). A great deal of data was considered; it just wasn't all relevant to the answering of my research question, it turned out. This is because clear cases of semantic anomia and phonological anomia are few and far between – as are purely semantic and purely phonological therapies. The ways in which researchers diagnosed phonetic disorders as phonological ones posed a problem too, as did the ways in which they tested for statistical significance. It makes sense to say then that while we can draw qualitative conclusions from my data sample (e.g. the fact that semantic therapy was found to treat anomia 33% of the time could cast doubt on theories

that phonology is a domain specific massive module), because n only equalled 22 (and the efficacies of only 3 semantic therapies at treating phonological anomia were considered) we should draw quantitative conclusions only tentatively. More robust results would be produced from a sample that was greater in size than mine, and so I suggest that for future research.

# 4.2.3.2.1.2 Factors that affect naming 4.2.3.2.1.2.1 Word frequency and length

There have also been a number of other factors found to affect naming such as word frequency, word length, familiarity, age of acquisition, imageability and concreteness, operativity, and animacy (Nickels 1997).

Newcombe et al. (1965), for example, found there to exist a linear relationship between the latency of word naming and the words to be named's frequencies, with lower frequency words taking longer to name than higher ones. Something similar was seen in a number of other studies (e.g. Howes 1964; Rochford and Williams 1965; Butterworth et al. 1984; Howard et al. 1985), but whilst Newcombe et al. (1965) found there to be a negative correlation between word frequency and response latencies (as word frequency went up, response latencies went down), others found word frequency and naming accuracy to be positively correlated (the higher in frequency a word was, the more accurately it was named by participants).

Nevertheless, despite the seeming robustness of these results, not everyone (e.g. Howard et al. 1985; Caramazza and Hillis 1990; Nickels and Howard 1995) has found word frequency to affect naming. Nickels and Howard (1995), for instance, studied the effects of frequency on naming in 27 aphasic subjects and found that frequency was only a significant predictor of performance for 2 of them; the other 25 did not evidence any effects of frequency on naming. Nickels and Howard (1995) suggest that the stark contrast of their results compared with other people's might well be because word frequency, in other studies, is oftentimes confounded with other variables (especially length, to which I'll now turn). In order to determine the amount of variance accounted for by frequency alone, Nickels and Howard (1995) argue, authors would do well to conduct multiple regression analyses.

With regards to word length, a negative correlation between that and naming accuracy has been found by most authors (Dubois et al. 1964; Goodglass et al. 1976; Caplan 1987; Ellis et al. 1983). As was the case with word frequency, though, not every aphasic has been found to

show an effect of word length on naming (EST, from Kay and Ellis (1987), didn't, for example, and Best (1995) actually describes the case of a patient who showed a reverse length effect (that is, found shorter words harder to name than longer ones)).

# 4.2.3.2.1.2.2 Orthography

Now, regardless of whether word frequency and word length or indeed other variables which have the potential to affect naming that I briefly mentioned but that I didn't go into in section 4.2.2.2.1 (such as familiarity, age of acquisition, imageability and concreteness, operativity, and animacy (see Nickels (1997) for a thorough review) had an effect on naming in the participants in my data set, I am particularly interested in figuring out whether treating orthographic therapy as being phonological therapy might have affected the results also.

Of the 7 studies that made use of phonological treatments (Nettleton and Lesser 1991; Hillis and Caramazza 1994; Best et al. 1997; Jokel and Rochon 1998; Hickin et al. 2002a; Hickin et al. 2002b; Raymer and Ellsworth 2002), 3 of them made use of graphemes as well as phonemes in the phonological treatment (Best et al. 1997; Hickin et al. 2002a; Hickin et al. 2002b). In these 3 studies were 5 results, and so I excluded them from my analysis to see how much of an effect including orthographic treatments had. 4 had to do with phonological anomia, and, interestingly enough, it did affect the results, bringing the efficacy of phonological therapy at treating phonological anomia down to 60% of the time from 78%.

Therapy type	Therapy result		
	Treated	Not treated	
Semantic therapy	1	2	
Phonological therapy	3	2	

Table 5.16 The effects of semantic therapy and phonological therapy at treating phonological anomia in the sample (excluding those phonological therapies that made use of orthography)

The effect of semantic therapy on phonological anomia stayed the same, of course, at 33% and so, with phonological therapies that make use of orthography excluded, there is even less of a difference between the efficacies of phonological treatments and semantic treatments at

treating phonological anomia, which could count as evidence that phonology is, indeed, ammodular.

I will point out, though, that once again, the *p* value produced by a Fisher's exact test was 1, which is, of course, not significant at p < 0.5. This means that, although we can conclude that spoken phonological therapies and semantic therapies have a similar success rate for the treatment of phonological anomia (3/2 compared with 2/2, respectively) in this particular sample, it is difficult to determine whether the same success rates would be seen beyond it.

## 4.2.3.2.2 The effects of publication bias

Publication bias is defined as the failure to publish the results of a study on the basis of the negativity of its findings (DeVito and Goldacre 2018). In empirical research this is a common occurrence – studies that have positive results are more likely to be published than studies that have negative ones. Meta-analyses like mine (which involve statistical analysis of pooled data) should therefore ask themselves whether their data collection has been affected by publication bias, as the answer 'yes' to his question would (and should) affect their data analysis. One way to test for this is by using a funnel plot. If the data when plotted looks to be in the shape of a symmetric funnel, then publication bias is unlikely, it is said. If the plotted data looks asymmetric, on the other hand, then the opposite is thought to be true (Hedin et al. 2016).

Egger's regression is the standard statistical measure for quantifying funnel plot asymmetry, DeVito and Goldacre (2018) explain, but the minimum required number of studies for the use of it is 10. Of course, in this thesis, only 2 studies (which elicited a total of 3 results) were collected for the analysis of the effects of semantic therapy at treating phonological anomia, and so my set is too small for this assessment.

It is important to bear in mind the possible impact publication bias may have on the conclusions I come to in this thesis regardless, I think, for if a number of negative results, that is, a number of studies evidencing that semantic therapy did not effectively treat phonological anomia, went unpublished, had they been so, the percentage of semantic therapy effectively treating phonological anomia in the data may have been much lower. It may have even been so low that I was led to a different conclusion, that phonology is indeed domain specific, and therefore modular.

Now, this is not an argument I'm making. There isn't enough evidence to do so. But this is a variable that neither I can (nor any scientist can) control for that has the potential to affect the analysis of the results and therefore the conclusion of this thesis, and so this is something that should be brought attention to in any scientific study.

## 5. Concluding remarks

I guess it's possible to argue from the above, then, that, according to Carruthers' (2006a) definition of modularity, phonology could well be considered ammodular. This is because for Carruthers (2006a) the domain specificity (the inability of it to operate on anything that is irrelevant to its function) of a cognitive processor is necessary for its modularity, and semantic therapy, in being able to treat phonological anomia 33% of the time, demonstrates that on those occasions, the patients' phonological processors were operating on semantic information. This would suggest that that phonology is neither domain specific nor massively modular, as semantic information is of course, irrelevant to the function of phonology.

As I said though, we should be careful how much we make of this conclusion, as a Fisher's exact test produced a p value of 0.2364 when orthography was included in the phonological therapies and 1 when it wasn't, which means that the results weren't statistically significant either way, and so we should proceed with caution in trying to generalise what was observed in this narrow sample to the wider population.

What I will say is this, though: with such small numbers you're unlikely to get a significant p value from Fisher's exact tests *anyway*, regardless of the ways in which the data patterns. Interested to see what pattern would produce a significant p value I ran some hypothetical Fisher's exact tests and found that in order for the results to be significant, phonological therapy would have had to have treated phonological anomia 100% of the time and semantic anomia 0% of the time. Even if phonological therapy treated phonological anomia 8 times out of 9 and semantic therapy 1 time out of 3 the p value would have been > 0.05 (0.1273):

Therapy type	Therapy result		
	Treated	Not treated	
Semantic therapy	1	2	
Phonological therapy	8	1	

Table 5.17 The hypothetical effects of semantic therapy and phonological therapy at treating phonological anomia if the data patterned differently

For this reason, then, I think it's safe to say then that my results might not necessarily be statistically insignificant because of the data's patterning. Rather, it might be because the sample was simply too small. We could, therefore, (very) tentatively conclude that phonology is amodular, but in order to reach a clearer conclusion than the one I have here, further research must be conducted.

# **Chapter 6: Conclusion**

This philosophy of mind (and language) thesis was unique in that it drew on data from different domains of linguistics (phonology and language change, in particular) in Chapter 4, and a different field of inquiry altogether (that of speech and language pathology) in Chapter 5, in order to shed some light on the question of whether or not phonology can be considered a module of the mind. My hope was that an answer would one day inform the models of word retrieval that are used in aphasiology, and in turn, the work of speech and language pathologists to improve treatment for phonological anomia.

In Chapter 4, I explored the extent to which phonology could be considered substance-free, and therefore an informationally encapsulated, Fodorian (1983, 2000) module of the mind, and in Chapter 5, whether or not word retrieval in phonological anomia could be facilitated by phonological therapy alone (or semantic therapy too), and therefore whether or not phonology could be considered a domain specific, Carrutherian (2006a) module.

At first glance, it looked as though each Chapter led me to a different conclusion; that phonology is substance-free and therefore can be considered modular according to Fodor's (1983, 2000) definition of modularity on the one hand (in 4), but isn't domain specific and therefore can't be considered modular according to Carruthers' (2006a) definition on the other (in 5).

This finding has important implications for Jackendoff's (2002) parallel architecture for the language faculty. Jackendoff's (2002) model, recall, is made up of three informationally encapsulated integrative modules (phonology, syntax and semantics) with likewise informationally encapsulated interface modules between them, that convert phonological to syntactic information (and vice versa), syntactic to semantic information (and vice versa) and semantic to phonological information (once again, and vice versa). The phonological, syntactic and semantic modules, therefore, operate on information specialised for their domain only (phonology on phonological information, syntax or syntactic information and semantics on semantic information).

Cognitive systems are informationally encapsulated, remember, to the extent that in the course of *processing* a given set of inputs (that are relevant *or* irrelevant to their function)

they cannot access information stored beyond the module; all they have to go on is the information that's contained in those inputs, and what's stored in the system itself. Phonology is informationally encapsulated in this model, in that is free of phonetic substance. It is at the phonetics-phonology interface, he proposes, that underlying phonological representations are turned into surface phonological representations, and at the phonetics-phonology interface (for he proposes two different ones) that phonetic forms become phonological.

Domain specific cognitive systems, remember, operate only on certain sorts of inputs – those that are specialised for their domain. They can operate only on inputs relevant to their function, as phonology, syntax and semantics all do in this model. The two features, although similar, are decidedly different then in that domain specificity restricts what information is accepted as input – what kind of information makes a system operate – and informational encapsulation restricts what kind of information is able to influence its operation.

To reiterate, Jackendoff's model (2002) therefore requires phonology to be both informationally encapsulated *and* domain specific. To his mind, phonological data is free of phonetic substance (because phonetic data is computed to become phonological data before it enters the module), and the phonological processor is domain specific, in that it is specialised to operate on the domain of phonological data only.

The finding in Chapter 5 of this thesis that the phonological processor was able to operate on semantic information in that phonological anomia could be facilitated by semantic anomia therapy, in evidencing that phonology is not domain specific leads us to a question, also, Jackendoff's parallel architecture of the language faculty. If phonology is not domain specific, it is neither modular, according to Carruthers (2006a), nor does Jackendoff's (2002) model appropriately capture its character.

Once again though, I will point out that Chapter 5's conclusion wasn't as clear-cut as the one in Chapter 4 was. The results in my dataset were found through Fisher's exact tests to be statistically insignificant, and so a definitive answer as to whether semantic therapy consistently facilitates word retrieval in phonological anomia (making phonology non-domain specific and ergo not massively modular) will require more results to be analysed than have been so here. Using statistical procedures, one is able to determine whether the null hypothesis (that what's observed was due to chance, and chance only) is accepted or rejected, and since my results weren't statistically significant, scientifically speaking, we should accept that. However, there is something to be said about the sample size here. In focusing exclusively on the statistical significant results of studies that involved investigation of single cases or series of single cases in which patients had semantic impairments only, or phonological impairments only, and received therapies that were purely semantic and/or purely phonological, a large number of papers I extracted from Nickels' (2002)'s catalogue had to be excluded from the analysis in order to ensure accuracy. Of the 43 papers available, only 9 turned out to meet the criteria for inclusion. This left me with a sample size of n = 22 (which is perhaps what made for the statistically insignificant test results, rather than the patterning).

As I said in Chapter 5, with a sample size as small as mine was (just 12 results for the treatment of phonological anomia) it's unlikely a Fisher's exact test would have yielded any statistically significant p values, no matter the patterning of the data. If the *same* pattern were observed in a sample ten times the size than mine (i.e. if 120 results were yielded instead of 12 with phonological therapy treating phonological anomia successfully 70 times out of 90 instead of 7 times out 9, and semantic therapy treating phonological anomia 10 times out of 30 successfully instead of 1 time out of 3, then the Fisher's exact test p value would have indeed been statistically significant (at 0, no less):

Therapy type	Therapy result		
	Treated	Not treated	
Semantic therapy	10	20	
Phonological therapy	70	20	

Table 6.1 The hypothetical effects of semantic therapy and phonological therapy at treating phonological anomia if the data patterned the same, but the sample were ten times the size

If this were the case, one would be able to conclude confidently that although phonological therapy was better at facilitating word retrieval in people with phonological anomia, semantic therapy did too 33% of the time, which would count as clear evidence against phonology's domain specificity (and massive modularity).

My results might well have posed a challenge to the view that phonology is domain specific if my sample size was larger then, which is interesting because if that were the case, one might conclude that whether phonology can be considered modular or not depends, really, on your definition of modularity. The fact of the matter is though; it was not. The sample size was small, and so whilst we can say that there is no real difference between the efficacies of semantic and phonological therapies at treating phonological anomia in *this* data set, it's difficult to say too much about what this means for modularity without being able to generalise what's been seen in this portion of the population to the whole population (which is what a statistically significant result would have allowed us to do).

It's safe to say then that this thesis has raised a number of questions. On the one hand, I've argued that phonology is substance-free and therefore can be considered an informationally encapsulated Fodorian (1983, 2000) module of the mind but on the other have found (an albeit small and statistically insignificant amount of) evidence to suggest that phonology might not be domain specific after all and therefore ammodular by Carruthers' (2006a) definition.

My results incline me to conclude (albeit tentatively) that phonology is substance-free and therefore modular in the modest sense of the word, but not domain specific (and therefore not modular) in the massive one – which of course casts some doubt on whether the model of the mind described by Jackoff is adequate after all, since Jackendoff (2002) requires phonology to be both.

However, whilst we can define a module as being informationally encapsulated but not domain specific in *theory*, it's not clear to me how that would work in *practice*. A non-domain specific phonological module would be able to operate on, say, semantic and syntactic information. But, in it being informationally encapsulated, semantic and syntactic knowledge wouldn't be able to influence its operations. This doesn't make much sense.

I should also probably point out, that even in a substance-free model of phonology, domain specificity is expected. For if phonology is able to operate only on substance-free phonological computation, then it is specialised for the domain of substance-free phonological computation, not substance-free semantic computation, say, or substance-free syntactic computation. To me, it appears that the two, informational encapsulation and domain specificity, that is, would go hand in hand, though that's not what the results of this thesis predict.

Of course, a study of a similar methodology to this one but of a larger sample size than mine would have to be carried out in order to confirm or deny whether that latter component of my conclusion (that phonology is not domain specific) is indeed the case. Because although I have found some evidence to suggest that phonology is able to operate on semantic information and is therefore not domain specific, as I've said, the sample size was small making for a statistically insignificant result. There are a number of other variables that have the potential to have affected these results too, including, but not limited to, the possibility that publication bias could have influenced which effects of semantic anomia therapy at treating phonological anomia were published, and which ones weren't. As I said in Chapter 5, studies that have positive results are more likely to be published than studies that have negative ones in the sciences, and had there been more studies with negative results been published and included in my data set, then the percentage of semantic therapies that were found to treat phonological anomia in thesis may well have been lower than 33%.

I regret that a more definitive answer to my research question has not been found, but hope that what I have done lies some significant groundwork for future study by establishing a direction of inquiry, and sufficiently summarising the limitations with the current study such that they can be circumvented if a similar one were to be undertaken. Whether phonology is modular or not then, as ever, remains a mystery. But until we have just one thesis of modularity (and phonology, for that matter), I think that's likely to be the case.

## References

Adami, C. 2002. 'Ab initio modelling of ecosystems with artificial life'. *Natural Resource Modeling* 15: 133-145.

Aftonomos, L.B., R.D. Steele and R.T. Wertz. 1997. 'Promoting recovery in chronic aphasia with an interactive technology'. *Archives of Physical Medicine Rehabilitation* 78: 841–846.

Agutter, A. 1988. 'The not-so-Scottish vowel length rule'. In J.M. Anderson and N. Macleoud (eds) *Edinburgh Studies in the English Language*. Edinburgh: John Donald. 120-132.

Aitken, A.J. 1981. 'The Scottish vowel-length rule'. In M. Benskin and M.L. Samuels (eds) So Meny People Longages and Tonges: Philological Essays in Scots and Mediaeval English Presented to Angus McIntosh. Edinburgh: The Middle English Dialect Project. 131-157.

Albert, M., H. Goodglass, N. Helm, A. Rubens and M. Alexander. 1981. *Clinical Aspects of Dysphagia*. New York: Springer-Verlag.

Annoni, J.M., A. Khater, M.C. Custodi, V. Debeauvais, C.M. Michel and T. Landis. 1998.'Advantage of semantic language therapy in chronic aphasia: A study of three cases'.*Aphasiology* 12: 1093–1105.

Antonucci, S.M. and J. Reilly. 2008. 'Semantic memory and language processing: A primer'. *Semin. Speech Lang.* 29: 5-17.

Apoussidou, D. 2006. 'The learnability of metrical phonology'. PhD Thesis. University of Amsterdam.

Applebaum, I. 1998. 'Fodor, modularity, and speech perception'. *Philosophical Psychology* 11: 317-330.

Arbib, M. 1987. 'Modularity and interaction of brain regions underlying visuomotor coordination'. In J.L. Garfield (ed.) *Modularity in Knowledge Representation and Natural-Language Understanding*. Cambridge, MA: MIT Press. 707-711. Archangeli, D. 1988. 'Aspects of underspecification theory'. *Phonology* 5: 183-207.

Archangeli, D. and D. Pulleyblank. 1994. *Grounded Phonology*. Cambridge, MA: The MIT Press.

Atran, S. 1990. Cognitive Foundations of Natural History: Towards an Anthropology of Science. Cambridge: Cambridge University Press.

Atran, S. 2002. 'Modular and cultural factors in biological understanding'. In P. Carruthers, S. Stich and M. Siegal (eds) *The Cognitive Basis of Science*. Cambridge: Cambridge University Press. 41-72.

Avery, P. 1996. 'The representation of voicing contrasts'. PhD Thesis. University of Toronto.

Avery, P. and K. Rice. 1989. 'Segmental structure and coronal underspecification'. *Phonology* 6: 179-200.

Avery, P. and K. Rice. 2004. 'The representational residue: The role of contrast in phonology'. Paper presented at the 12<sup>th</sup> Manchester Phonology Meeting.

Baillargeon, R. 1995. 'A model of physical reasoning in infancy'. In C. Rovee-Collier and L.P. Lipsitt (eds) *Advances in Infancy Research*. Norwood, NJ: Ablex. 305-371.

Baldwin, C.Y. and K.B. Clark. 1997. 'Managing in an age of modularity'. *Harvard Business Review* 75: 84-93.

Ballard, K.J., J.P. Granier and D.A. Robin. 2000. 'Understanding the nature of apraxia of speech: Theory, analysis and treatment'. *Aphasiology* 14: 969–995.

Barbutti, R., R. Giacobazzi and G. Levi. 1993. 'A general framework for semantics based on bottom-up abstract interpretation of logic programs'. *ACM Transactions on Programming Languages and Systems* 15: 133-181.

Bargh, J.A. and T.L. Chartrand. 1999. 'The unbearable automaticity of being'. *American Psychologist* 54: 462-479.

Barkow, J.H., L. Cosmides and J. Tooby (eds). 1992. *The Adapted Mind: Evolutionary Psychology and the Generation of Culture*. New York: Oxford University Press.

Barnes, J. 2006. *Strength and Weakness at the Interface: Positional neutralization in Phonetics and Phonology*. Berlin: Walter de Gruyter.

Baron-Cohen, S. 1995. Mindblindness. Cambridge, MA: MIT Press.

Barrett, H.C. 2005. 'Enzymatic computation and cognitive modularity'. *Mind & Language* 20: 259-287.

Barrett, H.C. 2006. 'Modularity and design reincarnation'. In P. Carruthers, S. Laurence and S. Stich (eds) *The Innate Mind: Culture and Cognition*. New York: Oxford University Press. 199-217.

Barrett, H.C. and R. Kurzban. 2006. 'Modularity in cognition: Framing the debate'. *Psychological Review* 113: 628-647.

Basso, A., P. Marangolo, F. Piras and C. Galluzzi. 2001. 'Acquisition of new 'words'' in normal subjects: A suggestion for treatment of anomia'. *Brain and Language* 77: 45–59.

Baudouin de Courtenay, J. 1972. *A Baudouin de Courtenay Anthology: The Beginnings of Structural Linguistics*. Bloomington, IN: Indiana University Press.

Bechtel, W. and R.C. Richardson. 1993. *Discovering Complexity: Decomposition and Localization as Strategies in Scientific Research*. Princeton, NJ: Princeton University Press.

Beckman, J. 1997. 'Positional faithfulness, positional neutralization, and Shona vowel harmony'. *Phonology* 14: 1-46.

Beeson, P. 1999. 'Treating acquired writing impairment: Strengthening graphemic representations'. *Aphasiology* 13: 767–785.

Benson, D. 1979. Aphasia, Alexia and Agraphia. New York: Churchill Livingstone.

Bermudez-Otero, R. 2017. 'The allophony of English /ai/ reconsidered'. Last accessed on 11 October 2019, from: http://www.bermudez-otero.com/Bloomington.pdf.

Bermudez-Otero, R. and G. Trousdale. 2012. 'Cycles and continua: On unidirectionality and gradualness in language change'. In T. Nevalainen and E. Closs Traugott (eds) *The Oxford Handbook of the History of English*. New York: Oxford University Press. 691-720.

Best, W.M., D. Howard, C. Bruce and C.E.P. Gatehouse. 1997. 'Cueing the words: A single case study of treatments for anomia'. *Neuropsychological Rehabilitation* 7: 105-141.

Blaho, S. 2008. 'The syntax of phonology: a radically substance-free approach'. PhD Thesis. University of Tromsø.

Blevins, J. 2004. *Evolutionary Phonology. The Emergence of Sound Patterns*. Cambridge: Cambridge University Press.

Blevins, J. and A. Garrett. 2004. 'The evolution of metathesis'. In B, Hayes, R. Kirchner and D. Steriade (eds) *Phonetically Based Phonology*. Cambridge: Cambridge University Press. 117-156.

Bloom, P. 2004. *Descartes' Baby: How the Science of Child Development Explains what Makes us Human*. New York: Basic Books.

Boersma, P. 2006. 'A programme for bidirectional phonology and phonetics and their acquisition and evolution'. Ms. University of Amsterdam.

Boersma, P. 2007. 'Cure constraints and their interactions in phonological perception and production'. Ms. University of Amsterdam.

Boersma, P. and S. Hamann. 2007. 'The evolution of auditory contrast'. Ms. University of Amsterdam.

Boersma, P., P. Escudero and R. Hayes. 2003. 'Learning abstract phonological from auditory phonetic categories: An integrated model for the acquisition of language-specific sound categories'. In M. Solé, D. Recasens and J. Romero (eds) *Proceedings of the 15<sup>th</sup>* 

*International Congress of Phonetic Sciences*. Barcelona: Universitat Autònoma de Barcelona. 3-9.

Bolker, J.A. 2000. 'Modularity in development and why it matters to Evo-Devo'. *American Zoologist* 40: 770-776.

Boyer, P. 2001. Religion Explained. New York: Basic Books.

Bozeat, S., M.A. Lambon Ralph, K. Patterson, P Garrard and J.R. Hodges. 2000. 'Non-verbal semantic impairment in semantic dementia'. *Neuropsychologica* 38. 1207-1215.

Breedin, S.D., E.M. Saffran and H.B. Coslett. 1994. 'Reversal of the concreteness effect in a patient with semantic dementia'. *Cogn. Neuropsychol.* 11: 617-660.

Broca, P. 1865. 'Sur le siege de la faculte du langage articule'. *Bulletin de la Societe d'Anthropologie* 6: 337-393.

Brookshire, R.H. 1996. Introduction to Neurogenic Communication Disorders. 5th ed. St Louis: Mosby.

Browman, C. and L. Goldstein. 1986. 'Towards an articulatory phonology'. *Phonology* 3: 219-252.

Brown, A.S. 1991. 'A review of the tip-of-the-tongue experience'. *Psychological Bulletin* 2: 204-223.

Brown, R. and D. McNeill. 1966. 'The "tip-of-the-tongue" phenomenon'. *Journal of Verbal Learning and Verbal Behavior* 5: 325-337.

Browne, D. 1996. 'Cognitive versatility'. Minds and Machines 6: 507-523.

Bruce, C. and D. Howard. 1987. 'Computer-generated phonemic cues: An effective aid for naming in aphasia'. *British Journal of Disorders of Communication* 22: 191-201.

Budd, M.A., K. Kortle, L. Cloutman, M. Newhart, R.F. Gottesman, C. Davis, J. Heidler-Gary, M.W. Seay and A.E. Hillis. 2010. 'The nature of naming errors in primary progressive aphasia versus acute post-stroke aphasia'. *Neuropsychology* 24: 581-589.

Buller, D.J. 2005. 'Get over: Massive modularity'. Biology and Philosophy 20: 881-891.

Buller, D. and V.G. Hardcastle. 2000. 'Evolutionary psychology, meet developmental neurobiology: Against promiscuous modularity'. *Brain and Mind* 1: 307-325.

Burton-Roberts, N. 2011. 'Where and what is phonology? A representational perspective'. In N. Burton-Roberts, P. Carr and G. Docherty (eds) *Phonological Knowledge*. Oxford: Oxford University Press. 39-66.

Buser, P.A. and M. Imbert. 1992. Vision. Cambridge, MA: MIT Press.

Butterworth, B. 1989. 'Lexical access in speech production'. In W. Marslen-Wilson (ed.) *Lexical Representation and Process*. Cambridge, MA: MIT Press. 261-286.

Butterworth, B. 1992. 'Disorders of phonological encoding'. Cognition 42: 261-286.

Butterworth, B.L., D. Howard and P.J. McLouglin. 1984. 'The semantic deficit in aphasia: The relationship between semantic errors in auditory comprehension and picture naming'. *Neuropsychologia* 22: 409–426.

Bybee, J. 2003. *Phonology and Language Use (Vol. 94)*. Cambridge: Cambridge University Press.

Bybee, J. 2006. *Frequency of Use and the Organization of Language*. Oxford: Oxford University Press.

Bybee, J. 2010. Language, Usage and Cognition. Cambridge: Cambridge University Press.

Cain, M.J. 2013. Fodor: Language, Mind and Philosophy. Hoboken, NJ: John Wiley & Sons.

Cain, M.J. 2016. The Philosophy of Cognitive Science. Cambridge: Polity Press.
Calabretta, R. and D. Parisi. 2005. 'Evolutionary connectionism and mind/brain modularity'. In W. Callebaut and D. Rasskin-Gutman (eds) *Modularity: Understanding the Development and Evolution of Natural Complex Systems*. Cambridge, MA: MIT Press. 309-330.

Callebaut, W. 2005. 'The ubiquity of modularity'. In W. Callebaut and D. Rasskin-Gutman (eds) *Modularity: Understanding the Development and Evolution of Natural Complex Systems*. Cambridge, MA: MIT Press. 3-28.

Callebaut, W. and D. Rasskin-Gutman. 2005. *Modularity: Understanding the Development and Evolution of Natural Complex Systems*. Cambridge, MA: MIT Press.

Cam, P. 1988. 'Modularity, rationality, and higher cognition'. *Philosophical Studies* 53: 279-294.

Caramazza, A., D. Chialant, R. Capasso and G. Miceli. 2000. 'Separable processing of consonants and vowels'. *Nature* 403: 428-430.

Caramazza, A. and M. Miozzo. 1997. 'The relation between syntactic and phonological knowledge in lexical access: Evidence from the 'tip-of-the-tongue' phenomenon'. *Cognition* 64: 309-343.

Caramazza, A., A.E. Hillis, B.C. Rapp and C. Romani. 1990. 'The multiple semantic hypothesis: Multiple confusions?' *Cogn. Neuropsychol.* 7: 161-189.

Caramazza, A. and A.E. Hillis. 1990. 'Spatial representation of words in the brain implied by studies of a unilateral neglect patient'. *Nature* 346: 267-269.

Carey, S. 1985. Conceptual Change in Childhood. Cambridge, MA: MIT Press.

Carey, S. and E. Spelke. 1994. 'Domain specific knowledge and conceptual change'. In L. Hirschfeld and S. Gelman (eds) *Mapping the Mind: Domain Specificity in Cognition and Culture*. Cambridge: Cambridge University Press. 169-200.

Carlomagno, S., M. Pandolfi, L. Labruna, A. Colombo and C. Razzano. 2001. 'Recovery from moderate aphasia in the first year post-stroke: Effect of type of therapy'. *Archives of Physical Medicine Rehabilitation* 82: 1073–1080.

Carroll, J.M. 1988. 'Modularity and naturalness in cognitive science'. *Metaphor and Symbol Activity* 3: 61-86.

Carruthers, P. 1992. *Human Knowledge and Human Nature*. Oxford: Oxford University Press.

Carruthers, P. 2003a. 'Moderately massive modularity'. In A. O'Hear (ed.) *Minds and Persons: Royal Institute of Philosophy Supplement 53*. Cambridge: Cambridge University Press. 67-89.

Carruthers, P. 2003b. 'On Fodor's problem'. Mind and Language 18: 502-523.

Carruthers, P. 2004. 'The mind is a system of modules shaped by natural selection'. In C. Hitchcock (ed.) *Contemporary Debates in the Philosophy of Science*. Oxford: Blackwell. 293-311.

Carruthers, P. 2005. *Consciousness: Essays from a Higher-Order Perspective*. Oxford: Clarendon Press.

Carruthers, P. 2006a. The Architecture of the Mind. New York: Oxford University Press.

Carruthers, P. 2006b. 'The case for massively modular models of mind'. In R.J. Stainton (ed.) *Contemporary Debates in Cognitive Science*. Malden, MA: Blackwell Publishing. 3-21.

Carruthers, P. and A. Chamberlain (eds). 2000. *Evolution and the Human Mind: Modularity, Language and Meta-Cognition*. Cambridge: Cambridge University Press.

Carruthers, P., S. Laurence and S. Stich. 2005. *The Innate Mind: Structure and Contents*. New York: Oxford University Press.

Chambers, J.K. 1973. 'Canadian raising'. The Canadian Journal of Linguistics 18: 113-135.

Chambers, J.K. 1989. 'Canadian raising: Blocking, fronting, etc.'. *American Speech* 64: 75-88.

Chambers J.K. 2006. 'Canadian raising: Retrospect and prospect'. *Canadian Journal of Linguistics* 51: 105-118.

Chambers, J.K. and M.F. Hardwick. 1986. 'Comparative sociolinguistics of a sound change in Canadian English'. *English Word-Wide* 7: 23-46.

Charland, L. 1995. 'Feeling and representing: Computational theory and the modularity of affect'. *Synthese* 105: 273-301.

Chen, M. 1970. 'Nasals and nasalization'. Paper presented at the 1970 Linguistic Society of America Meeting.

Cheng, K. 1986. 'A purely geometric module in the rat's spatial representation'. *Cognition* 23: 149-178.

Cheng, K. and N.S. Newcombe. 2005. 'Is there a geometric module for spatial orientation? Squaring theory and evidence. *Psychonomic Bulletin and Review* 12: 1-23.

Chien, A.J. 1996. 'Why the mind may not be modular'. Minds and Machines 6: 1-32.

Chomsky, N. 1959. 'A review of B.F. Skinner's verbal behavior'. Language 35: 26-58.

Chomsky, N. 1965. Aspects of the Theory of Syntax. Cambridge, MA: MIT Press.

Chomsky, N. 1980. 'Rules and representations'. Behavioral and Brain Sciences 3: 1-15.

Chomsky, N. 1988. Language and Problems of Knowledge. Cambridge: MIT Press.

Chomsky, N. and M. Halle. 1968. The Sound Pattern of English. New York: Harper Row.

Churchland, P. 1988. 'Perceptual plasticity and theoretical neutrality: A reply to Jerry Fodor'. *Philosophy of Science* 55: 167-187.

Churchland, P. 1989. *A Neurocomputational Perspective: The Nature of Mind and the Structure of Science*. Cambridge, MA: MIT Press.

Churchland, P., C. Koch and T. Sejnowski. 1990. 'What is computational neuroscience?'. In E. Schwartz (ed.) *Computational Neuroscience*. Cambridge, MA: MIT Press.

Clements, G. and E. Hume. 1995. 'The internal organization of speech sounds'. In J. Goldsmith (ed.) *Handbook of Phonological Theory*. Cambridge, MA: Blackwell. 245-306.

Collins, J. 2005. 'On the input problem for massive modularity'. *Minds and Machines* 15: 1-22.

Coltheart, M., A. Woollams, S. Kinoshita and C. Perry. 1999. 'A position-sensitive Stroop effect: Further evidence for a left-to-right component in print-to-speech'. *Psychonomic Bulletin and Review* 6: 456-463.

Compton, R. and B. Dresher. 2011. 'Palatalization and "strong i" across Inuit dialects'. *Canadian Journal of Linguistics* 56: 203-228.

Cosmides, L. and J. Tooby. 1992. 'Cognitive adaptations for social exchange'. In J. Barkow, L. Cosmides and J. Tooby (eds) *The Adapted Mind*. Oxford: Oxford University Press. 163-228.

Cosmides, L. and J. Tooby. 1994. 'Origins of domain specificity: The evolution of functional organization'. In L.A. Hirschfield and S.A. Gelman (eds) *Mapping the Mind: Domain Specificity in Cognition and Culture*. Cambridge, MA: MIT Press. 85-116.

Cosmides, L. and J. Tooby. 2001. 'Unraveling the enigma of human intelligence'. In R. Sternberg and J. Kaufman (eds) *The Evolution of Intelligence*. Hillsdale, NJ: Erlbaum. 145-198.

Cosmides, L., J. Tooby and J.H. Barkow. 1992. 'Introduction: Evolutionary psychology and conceptual integration'. In J.H. Barkow, L. Cosmides and J. Tooby (eds) *The Adapted Mind*. New York: Oxford University Press. 3-15.

Cowie, F. 1999. What's Within? Nativism Reconsidered. Oxford: Oxford University Press.

Crain, S. and D. Lillo-Martin. 1999. *An Introduction to Linguistic Theory and Language Acquisition*. Oxford: Blackwell.

Croot, K., K. Patterson and J.R. Hodges. 1998. 'Single word production in nonfluent progressive aphasia'. *Brain Lang* 61: 226-273.

Crosswhite, K. 2001. Vowel Reduction in Optimality Theory. New York: Routledge.

Currie, G. and K. Sterenly. 2000. 'How to think about the modularity of mind-reading'. *The Philosophical Quarterly (1950-)* 50: 145-160.

Cusumano, M.A. and K. Nobeoka. 1998. *Thinking Beyond Lean: Multi-project Management*. New York: The Free Press.

Dabrowska, E. 2015. 'What exactly is Universal Grammar, and has anyone seen it?'. *Frontiers in Psychology* 6: 852.

Damasio, H. 1989. 'Neuroimaging contributions to the understanding of aphasia'. In F. Boller and J. Grafman (eds) *Handbook of Neuropsychology*. Amsterdam: Elsevier. 3-46.

Damasio, A. 1992. 'Aphasia'. Medical Progress 326: 531-539.

Damasio, H., T.J. Grabowski, D. Tranel, R.D. Hichward and A.R. Damasio. 1996. 'A neural basis for lexical retrieval. *Nature* 380: 499-505.

Darley, F., A. Aronson and J. Brown. 1975. Motor Speech Disorders. Philadelphia: Saunders.

Daroff, R.B., G.M. Fenichel, J. Jankovic and J.C. Mazziotta (eds). 2012. *Bradley's Neurology in Clinical Practice*. 6<sup>th</sup> ed. New York: Elsevier Saunders.

Davenport, M. and S.J. Hannahs. 2010. *Introducing Phonetics and Phonology*. 3<sup>rd</sup> ed. London: Hodder Education.

Dawkins, R. 1976. The Selfish Gene. Oxford: Oxford University Press.

Day, R.H. 1989. 'Natural and artificial cues, perceptual compromise, and the basis of veridical and illusory perception'. In D. Vickers and P.L. Smith (eds) *Human Information Processing: Measures, Mechanisms, and Models*. New York: Elsevier. 107-129.

Deacon, T. 1998. The Symbolic Species. New York: Norton.

Dell, G. 1986. 'A spreading-activation theory of retrieval in sentence production'. *Psychological Review* 93: 283-321.

Dell, G.S. 1988. 'The retrieval of phonological forms in production: Tests of predictions from a connectionist model'. *Journal of Memory and Language* 27: 124-142.

Dell, G.S. 1989. 'The retrieval of phonological forms in production: Tests of predictions from a connectionist model'. In W. Marsh-Wilson (ed.) *Lexical Representation and Process*. Cambridge, MA: MIT Press. 136-165.

Dell, G.S., M.F. Schwartz, N. Martin, E.M. Saffran and D.A. Gagnon. 1997. 'Lexical access in normal and aphasic speech'. *Psychological Review* 104: 801–838.

Deloche, G., M. Dordain and H. Kremin. 1993. 'Rehabilitation of confrontation naming in aphasia: Relations between oral and written modalities'. *Aphasiology* 7: 201–216.

Deloche, G., D. Hannequin, M. Dordain, M.N. Metz-Lutz, H. Kremin, C. Tessier, K. Vendrell, D. Cardebat, D. Perrier, S. Quint and B. Pichard. 1997. 'Diversity of patterns of improvement in confrontation naming: Some tentative hypotheses'. *Journal of Communication Disorders* 30: 11–22.

DeVito, D., J. and B. Goldacre. 2018. 'Catalogue of bias: Publication bias'. *BMJ Evidence Based Medicine*. pii.

Donegan, P. and D. Stampe. 1979. 'The natural study of phonology'. In D. Dinnsen (ed.) *Current Approaches to Phonology*. Bloomington: Indiana University Press. 126-173.

Dresher, B. 2001. 'Contrast and asymmetries in inventories'. Paper presented at the Asymmetry Conference.

Dresher, B. 2003. 'Contrast and asymmetries in inventories'. In A.M. du Sciullo (ed.) *Asymmetry in Grammar. Volume 2: Morphology, Phonology, Acquisition*. Amsterdam: John Benjamins. 239-257.

Dresher, B., G. Piggott and K. Rice. 1994. 'Contrast in phonology: Overview'. *Toronto Working Papers in Linguistics* 13: 3-17.

Dresher, B. and X. Zhang. 2004. 'Phonological contrast and phonetics in Manchu vowel systems'. *Proceedings of BLS* 19: 103-144.

Duffie, J.A. and W.A. Beckman. 2013. *Solar Engineering of Thermal Processes*. 4<sup>th</sup> ed. Hoboken, NJ: John Wiley & Sons, Inc.

Durand, J., U. Gut and G. Kristofferson. 2014. *The Oxford Handbook of Corpus Phonology*. Oxford: Oxford University Press.

Dwyer, S. 1999. 'Moral competence'. In K. Murasugi and R. Stainton (eds) *Philosophy and Linguistics*. Boulder, CO: Westview Press. 169-190.

Eales, C. and T. Pring. 1998. 'Using individual and group therapy to remediate word finding difficulties'. *Aphasiology* 12: 913–918.

Ellis, A.W. and A.W. Young. 1988. *Human Cognitive Neuropsychology*. Hove: Lawrence Erlbaum.

Ellis, A.W., D. Miller and G. Sin. 1983. 'Wernicke's aphasia and normal language processing: A case study in cognitive neuropsychology'. *Cognition* 15: 111-144.

Eble, G.J. 2005. 'Morphological modularity and macroevolution: Conceptual and empirical aspects'. In W. Callebaut and D. Rasskin-Gutman (eds) *Modularity: Understanding the Development and Evolution of Natural Complex Systems*. Cambridge, MA: MIT Press. 221-238.

Elman, J.L., E.A. Bates, M.H. Johnson, A. Karmiloff-Smith, D. Parisi and K. Plunkett. 1996. *Rethinking Innateness: A Connectionist Perspective on Development*. Cambridge, MA: MIT Press.

Epinosa-Soto, C. and A. Wagner. 2010. 'Specialization can drive the evolution of modularity'. *PLoS Computational Biology* 6: e1000719.

Escudero, P. and P. Boersma. 2003. 'Modelling the perceptual development of phonological contrasts with Optimality Theory and the Gradual Learning Algorithm'. *Penn Working Papers in Linguistics* 8: 71-85.

Farley, A., E. McLafferty and C. Hendry. 2012. 'The cardiovascular system'. *Nursing Standard* 27: 35-39.

Fiddick, L., L. Cosmides and J. Tooby. 2000. 'No interpretation without representation: The role of domain-specific representations and inferences in the Wason selection task'. *Cognition* 77: 1-79.

Fink, R.B., A. Brecher, M.F. Schwartz and R.R. Robey. 2002. 'A computer-implemented protocol for treatment of naming disorders: Evaluation of clinician-guided and partially self guided instruction'. *Aphasiology* 16: 1061-1086.

Fitzgerald, M. 1996. Neuroanatomy: Basic and Clinical. Philadelphia: Saunders.

Flemming, E. 2004. 'Contrast and perceptual distinctiveness'. In B. Hayes, R. Kirchner and D. Steriade (eds) *The Phonetic Bases of Markedness*. Cambridge: Cambridge University Press. 232-276.

Fodor, J.A. 1975. The Language of Thought. Cambridge, MA: Harvard University Press.

Fodor, J.A. 1980. 'Methodological solipsism considered as a research strategy in cognitive psychology'. *Behavioral and Brain Sciences* 3: 63-109.

Fodor, J. 1983. *Modularity of the Mind: An Essay on Faculty Psychology*. Cambridge, MA: MIT Press.

Fodor, J.A. 1987. *Psychosemantics: The Problem of Meaning in the Philosophy of Mind.* Cambridge, MA: MIT Press.

Fodor, J.A. 1998. 'The trouble with psychological Darwinism'. *London Review of Books* 20: 11-13.

Fodor, J.A. 2000. *The Mind Doesn't Work That Way: The Scope and Limits of Computational Psychology*. Cambridge, MA: MIT Press.

Fodor, J. and Z. Pylyshyn. 1988. 'Connectionism and cognitive architecture'. *Cognition* 28: 3-71.

Fodor, J.A., T. Bever and M. Garrett. 1974. *The Psychology of Language: An Introduction to Psycholinguistics and Generative Grammar*. New York: McGraw Hill.

Fogle, P.T. 2008. *Foundations of Communication Sciences and Disorders*. New York: Delmar.

Fontana, W. and L.W. Buss. 1994. "'The arrival of the fittest": Toward a theory of biological organization'. *Bulletin of Mathematical Biology* 56: 1-64.

Francis, D.R., N. Clark and G.W. Humphreys. 2002. 'Circumlocution-induced naming: A treatment for effecting generalisation in anomia?' *Aphasiology* 16: 243–259.

Friedland, J. 1994. 'Car industry: Mini miracle'. Far Eastern Economic Review 157: 76.

Friedmann, N. 2006. 'Speech production in Broca's agrammatic aphasia: Syntactic tree pruning'. In Y. Grodzinsky and K. Amunts (eds) *Broca's Region*. New York: Oxford University Press. 63-82.

Fromkin, V. 1971. 'The non-anomalous of anomalous utterances'. Language 47: 27-52.

Fruehwald, J. 2013. 'Phonological involvement in phonetic change'. PhD Thesis. University of Pennsylvania.

Fruehwald, J. 2016. 'The early influence of phonology on a phonetic change'. *Language* 92: 376-410.

Fulton, J. 2005. 'Chemoreceptors of taste and smell'. Last accessed on 26 July 2017, from: http://neuronresearch.net/smell/files/chemoreceptors.htm.

Futuyma, D. 2009. *Evolution*. 2<sup>nd</sup> ed. Sunderland, MA: Sinauer Associates.

Gallistel, C.R. 1990. The Organization of Learning. Cambridge, MA: MIT Press.

Gallistel, C.R. 2000. 'The replacement of general-purpose learning models with adaptively specialized learning modules'. In M. Gazzaniga (ed.) *The New Cognitive Neurosciences*. 2<sup>nd</sup> ed. Cambridge MA: MIT Press. 1179-1191.

Gallistel, C.R. and J. Gibbon. 2001. 'Time, rate and conditioning'. *Psychological Review* 108: 289-344.

Gallistel, C.R., T.A. Mark, A. King and P. Latham. 2001. 'The rat approximates an ideal detector of changes in rates of reward: Implications for the law of effect'. *Journal of Experimental Psychology: Animal Behavior Processes* 27: 354-372.

Garfield, J.L. (ed.). 1987. *Modularity in Knowledge Representation and Natural Language Understanding*. Cambridge, MA: MIT Press.

Garfield, J.L. 1991. *Modularity in Knowledge Representation and Natural-Language Understanding*. Cambridge, MA: MIT Press.

Garud, R. and A. Kumaraswamy. 1993. 'Changing competitive dynamics in network industries: An exploration of sun microsystems' open systems strategy'. *Strategic Management Journal* 14: 351-369.

Garud, R. and A. Kumaraswamy. 1995. 'Technological and organizational designs for realizing economies of substitution'. *Strategic Management Journal* 16: 93-109.

Geschwind, N. 1965. 'Disconnexion syndromes in animals and man. I'. Brain 88: 237-294.

Gigerenzer, G. 1994. 'Why the distinction between single-event probabilities and frequencies is important for psychology (and vice versa)'. In G. Wright and P. Ayton (eds) *Subjective Probability*. Chichester: Wiley. 129-161.

Gilbert, S.F, J.M. Opitz, R.A. Raff. 1996. 'Resynthesizing evolutionary and developmental biology'. *Dev. Biol.* 173: 357-372.

Gobet, F. 2015. Understanding Expertise: A Multidisciplinary Approach. London: Palgrave.

Goodglass, H. and E. Kaplan. 1983. *The Assessment of Aphasia and Related Disorders*. 2nd ed. Philadelphia: Lea & Febiger.

Goodglass, H. & A. Wingfield. 1997. *Anomia: Neuroanatomical and Cognitive Correlates*. Burlington: Elsevier Science.

Goodglass, H., E. Kaplan, S. Weintraub and N. Ackerman. 1976. 'The "tip-of-the-tongue" phenomenon in aphasia'. *Cortex* 12: 145-153.

Gordon, M. 2004. 'A phonetic and phonological study of word-level stress in Chickasaw.' *International Journal of American Linguistics* 70: 1-32.

Gottschling, V. 2009. 'Modularity'. In J. Symons and P. Calvo (eds) *The Routledge Companion to Philosophy*. London: Routledge. 296-321.

Greenberg, J. 1957. 'The nature and uses of linguistic typologies'. *International Journal of American Linguistics* 23: 68-77.

Greenberg, J. 1978. 'Typology and cross-linguistic generalizations'. In J. Greenberg, A. Fergerson and E. Moravcsik (eds) *Universals of Human Language, vol. 1: Method and Theory*. Stanford, CA: Stanford University Press. 33-59.

Greenwald, M.L., A.M. Raymer, M.E. Richardson and L.J.G. Rothi. 1995. 'Contrasting treatments for severe impairments of picture naming'. *Neuropsychological Rehabilitation* 5: 17–49.

Groom, P.K. and B.B. Lamont. 2015. *Plant Life of Southwestern Australia. Adaptions for Survival.* Warsaw: De Gruyter Open.

ten Hacken, P. 2019. 'The mental lexicon in Jackendoff's parallel architecture'. In P. ten Hacken (ed.) *Word Formation in Parallel Architrecture: The Case for a Separate Component*. Switzerland: Springer Nature. 3-22.

Hale, M. and C. Reiss. 2000a. 'Phonology as cognition'. In N. Burton-Roberts, P. Carr and G. Docherty (eds) *Phonological Knowledge*. Oxford: Oxford University Press. 161-184.

Hale, M. and C. Reiss. 2000b. "Substance abuse' and 'dysfunctionalism': Current trends in phonology'. *Linguistic Inquiry* 31: 157-169.

Hale, M. and C. Reiss. 2003. 'The Subset Principle in phonology. Why the tabula can't be rasa'. *Journal of Linguistics* 39: 219-244.

Hale, M. and C. Reiss. 2008. The Phonological Enterprise. Oxford: Oxford University Press.

Hale, M., M. Kissock and C. Reiss. 2007. 'Microvariation, variation, and the features of universal grammar'. *Lingua* 117: 645-665.

Hannahs, S.J. and A. Bosch (ed.). 2018. *The Routledge Handbook of Phonological Theory*. London: Taylor and Francis.

Harris, J. 1990. 'Segmental complexity and phonological government'. *Phonology* 7: 255-300.

Harris, J. 1994. English Sound Structure. Oxford: Blackwell.

Harris, J. 2005. 'Vowel reduction as information loss'. In P. Carr, J. Duranc and C. Ewen (eds) *Headhood, Elements, Specification and Contrastivity: Phonological Papers in Honour of John Anderson*. Amsterdam: John Benjamins Publishing. 119-132.

Harris, J. 2006. 'The phonology of being understood: Further arguments against sonority'. *Lingua* 116: 1483-1494.

Harris, J. and G. Lindsey. 1995. 'The elements of phonological representation'. In J. Durand and F. Katamba (eds) *Frontiers of Phonology: Atoms, Structures, Derivations*. London: Longman. 34-79.

Hauser, M.D. and S. Carey. 1998. 'Building a cognitive creature from a set of primitives: Evolutionary and developmental insights'. In D. Cummins and C. Allen (eds) *The Evolution of Mind*. New York: Oxford University Press. 51-106.

Hawking, S. 1988. A Brief History of Time. New York: Bantam Books.

Hayes, B. and D. Steriade. 2004. 'Introduction: The phonetic bases of phonological markedness'. In B. Hayes, R. Kirchner, D. Steriade (eds) *Phonetically Based Phonology*. Cambridge: Cambridge University Press. 1-33.

Hayes, B., D. Steriade and R. Kirchner. 2004. *Phonetically Based Phonology*. Cambridge: Cambridge University Press.

Hebb, D.O. 1949. The Organization of Behavior. New York: Wiley and Sons.

Hedin, R.J., B.A. Umberhaum, B.N. Detweiler, L. Kollmorgen and M. Vassar. 2016. 'Publication bias and nonreporting found in majority of systematic reviews and meta-analyses in anesthesiology journals'. *Anesth Analg* 123: 1018-1025.

Heinz, J. 2001a. 'Computational phonology part I: Foundations'. *Language and Linguistics Compass* 5: 140-152.

Heinz, J. 2011b. 'Computational phonology part II: Grammars, learning, and the future'. *Language and Linguistics Compass* 5: 153-168.

Helm-Estabrooks, N., M. Albert and M. Nicholas. 2014. *Manual of Aphasia and Aphasia Therapy*. 3rd ed. Austin, TX: PRO-ED Inc.

Henaff Gonon, M., R. Bruckert and F. Michel. 1989. 'Lexicalization in an anomic patient'. *Neuropsychologia* 27: 391–407.

Herbert, R.E. 2004. 'Therapy for word finding in aphasia. Effects on picture naming and conversation'. PhD Thesis. University College London.

Hermer, L. and E. Spelke. 1994. 'A geometric process for spatial reorientation in young children'. *Nature* 370: 57-59.

Hermer, L. and E. Spelke. 1996. 'Modularity and development: The case of spatial reorientation'. *Cognition* 61: 195-232.

Hermer-Vazquez, L., E.S. Spelke and A.S. Katsnelson. 1999. 'Sources of flexibility in human cognition: Dual-task studies of space and language'. *Cognitive Psychology* 39: 3-36.

Hickin, J., W. Best, D. Howard and F. Osborne. 2002a. 'Phonological therapy for word finding difficulties: A re-evaluation'. *Aphasiology* 16: 981–999.

Hickin, J., R. Herbert, W. Best, D. Howard and F. Osborne. 2002b. 'Lexical and functionally based treatment: Effects on word retrieval and conversation'. In S. Byng, C. Pound and J. Marshall (eds) *The Aphasia Therapy File 2*. Hove, UK: Psychology Press. 69-82.

Hillis, A.E. 1989. 'Efficacy and generalisation of treatment for aphasic naming errors'. *Archives of Physical Medicine Rehabilitation* 70: 632–636.

Hillis, A.E. 1998. 'Treatment of naming disorders: New issues regarding old therapies'. *Journal of the International Neuropsychological Society* 4: 648–660

Hillis, A. and A. Caramazza. 1994. 'Theories of lexical processing and theories of rehabilitation'. In G. Humphreys and M. Riddoch (eds) *Cognitive Neuropsychology and Cognitive Rehabilitation*. Hillsdale, NJ: Lawrence Erlbaum Associates. 449-482.

Hillis, A., B. Rapp, C. Romani and A. Caamazza. 1990. 'Selective impairment of semantics in lexical processing'. *Cognitive Neuropsychology* 7: 191-244.

Hirschfield, L.A. and S.A. Gelman (eds). 1994. *Mapping the Mind: Domain Specificity in Cognition and Culture*. New York: Cambridge University Press.

Hodges, J.R. and K. Patterson. 1996. 'Nonfluent progressive aphasia and semantic dementia: A comparative neuropsychological study'. *Journal of the International Neuropsychological Society* 2: 511.

Hodges, J.R., K. Patterson, S. Oxbury and E. Funnell. 1992. 'Semantic dementia. Progressive fluent aphasia with temporal lobe atrophy'. *Brain* 115: 1786-1806.

Hodges, J.R., K. Patterson and L.K. Tyler. 1994. 'Loss of semantic memory: Implications for the modularity of mind'. *Cogn Neuropsychol*. 11: 505-542.

Hodges, J.R., S. Bozeat, M.A. Lambon Ralph, K. Patterson and J. Spatt. 2000. 'The role of conceptual knowledge in object use: Evidence from semantic dementia'. *Brain* 123: 1913-1925.

Hombert, J., J. Ohala and W. Ewan. 1979. 'Phonetic explanations for the development of tones'. *Language* 55: 37-58.

Hooper, J. 1976. *An Introduction to Natural Generative Phonology*. New York: Academic Press.

Howard, D. and S. Franklin. 1988. Missing the Meaning? Cambridge, MA: MIT Press.

Howard, D. and V. Orchard-Lisle. 1984. 'On the origin of semantic errors in naming: Evidence from the case of a global aphasic'. *Cognitive Neuropsychology* 1: 163-190.

Howard, D. and K. Patterson. 1992. *Pyramids and Palm Trees: A Test of Semantic Access from Pictures and Words*. Bury St Edmunds: Thames Valley Test Company.

Howard, D., K. Patterson, S. Franklin, V. Orchard-Lisle and J. Morton. 1985. 'Treatment of word retrieval deficits in aphasia: A comparison of two therapy methods'. *Brain* 108: 817-829.

Hubel, D.H. and T.N. Wiesel. 1974. 'Sequence regularity and geometry of orientation columns in the monkey striate cortex'. *Journal of Comparative Neurology* 158: 267-294.

Hume, E. 2003. 'Language specific markedness: The case of place of articulation'. *Studies in Phonetics, Phonology and Morphology* 9: 295-310.

Hunt, J. 1999. 'Drawing on the semantic system: The use of drawing as a therapy medium'. In S. Byng, K. Swinburn and C. Pound (eds) *The Aphasia Therapy File*. Hove, UK: Psychology Press. 41-60.

Husken, M., C. Igel and M. Touissant. 2002. 'Task-dependent evolution of modularity in neural networks'. *Connection Science* 14: 219-229.

Hyman, L. 1972. 'Nasals and nasalization in Kwa'. Studies in African Linguistics 3: 167-206.

Hyman, L. 1975. Phonology: Theory and Analysis. New York: Holt, Rinehart & Winston.

Hyman, L. 1976. 'Phonologization'. In A. Juilland (ed.) *Linguistic Studies Presented to Joseph H. Greenberg*. Saratoga: Anna Libri. 407-418.

Idsardi, W. 2006. 'A simple proof that optimality theory is computationally intractable'. *Linguistic Inquiry* 37: 271-275.

Iosad, P. 2013. Representation and variation in substance-free phonology: A case study in Celtic. PhD Thesis. University of Tromsø.

Iosad, P. 2017. A Substance-free Framework for Phonology: An Analysis of the Breton Dialect of Bothoa (Edinburgh Studies in Theoretical Linguistics 2). Edinburgh: Edinburgh University Press.

Jackendoff, R. 2002. Foundations of Language. Oxford: Oxford University Press.

Jakobson, R. 1942. 'The Paleosiberian languages'. American Anthropologist 44: 602-620.

Jakobson, R., C. Fant and M. Halle. 1952. *Preliminaries to Speech Analysis: The Distinctive Features and their Correlates.* Cambridge, MA: MIT Press.

Jastrow, J. 1899. 'The mind's eye'. Popular Science Monthly 54: 299-312.

Jefferies, E. and M.A. Lambon Ralph. 2006. 'Semantic impairment in stroke aphasia versus semantic dementia: A case-series comparison'. *Brain* 129: 2132-2147.

Job, R. and L. Surian. 1998. 'A neurocognitive mechanism for folk biology?'. *Behavioral and Brain Sciences* 21: 577.

Johnson, K. 2007. 'Speech perception without speaker normalization: An exemplar model'. In K. Johnson and J. Mullenix (eds) *Talker Variability in Speech Processing*. London: Academic Press. 145-166.

Jokel, R. and E. Rochon. 1998. 'Treatment for an aphasic naming impairment: When phonology met orthography'. *Brain and Cognition* 32: 299–301.

Jones, G.V. 1989. 'Back to Woodworth: Role of interlopers in the tip-of-the-tongue phenomenon'. *Memory and Cognition* 17: 69-76.

Jones, G.V. and S. Langford. 1987. 'Phonological blocking in the tip of the tongue state'. *Cognition* 26: 115-122.

Joos, M. 1942. 'A phonological dilemma in Canadian English'. Language 18: 141-144.

Jun, J. 2004. 'Place assimilaton'. In B. Hayes, R. Kirchner and D. Steriade (eds) *Phonetically Based Phonology*. Cambridge: Cambridge University Press. 58-86.

Kaisse, E.M. and P.A. Shaw. 1985. 'On the theory of Lexical Phonology'. *Phonology Yearbook* 2: 1-30.

Kandel, E.R., J.H. Schwartz and T.M. Jessell (eds). 1995. *Essentials of Neural Science and Behaviour*. New York: McGraw-Hill.

Kaplan, E.F., H. Goodglass and S. Weintraub. 1983. *The Boston Naming Test.* 2nd ed. Philadelphia: Lea and Febiger.

Karmiloff-Smith, A., E. Klima, U. Bellugi, J. Grant and S. Baron-Cohen. 1995. 'Is there a social module? Language, face processing and theory of mind in individuals with Williams syndrome'. *Journal of Cognitive Neuroscience* 72: 196-208.

Kashino, M. 2006. 'Phoneme restoration: The brain creates missing speech sounds'. *Acoustical Science and Technology* 27: 318-321.

Kashtan, N. and U. Alon. 2005. 'Spontaneous evolution of modularity and network motifs'. *Proceedings of the National Academy of Sciences of the United States of America* 102: 13772-13778.

Kashtan, N., E. Noor and U. Alon. 2007. 'Varying environments can speed up evolution'. *Proceedings of the National Academy of Sciences of the United States of America* 104: 13711-13716.

Kaun, A. 2004. 'The typology of rounding harmony'. In B. Hayes, R. Kirchner and K.Steriade (eds) *Phonetically Based Phonology*. Cambridge: Cambridge University Press. 87-116.

Kay, J. and A. Ellis. 1987. 'A cognitive neuropsychological case study of anomia. Implications for psychological models of word retrieval'. *Brain* 110: 613-629.

Kay, L.E., D. Marion and A. Bax. 1989. J. Magn. Reson. 84: 72.

Kay, J., R. Lesser and M. Coltheart. 1991. *Psycholinguistic Assessments of Language Processing in Aphasia*. Hove: Lawrence Erlbaum.

Kay, J., R. Lesser and M. Coltheart. 1992. *PALPA: Psycholinguistic Assessments of Language Processing in Aphasia.* Hove: Lawrence Erlbaum.

Kearns, K. 2004. 'Broca's aphasia'. In L. LaPointe (ed.) *Aphasia and Related Neurogenic Language Disorders*. New York: Thieme. 117-141.

Keller, S.S., T. Crow, A. Foundas, K. Amunts and N. Roberts. 2009. 'Broca's area: Nomenclature, anatomy, typology and asymmetry'. *Brain and Language* 109: 29-48.

Kempen, G. and P. Huijbers, P. 1983. 'The lexicalization process in sentence production and naming: Indirect election of words'. *Cognition* 14: 185-209.

Kertesz, A. 1982. Western Aphasia Battery. San Antonio: The Psychological Corporation.

Kertesz, A. and E. Poole. 1974. 'The aphasia quotient: The taxonomic approach to measurement of aphasic disability'. *Can J Neurol Sci.* 1: 7-16.

Kertesz, A., W. Harlock and R. Coates. 1979. 'Computer tomographic localization, lesion size, and prognosis in aphasia and nonverbal impairment'. *Brain and Language* 8: 34-50.

Kilmer, W.L., W.S. McCulloch and J. Blum. 1969. 'A model of the vertebrate central command system'. *International Journal of Man-Machine Studies* 1: 267-294.

Kim, H.Y. 2017. 'Statistical notes for clinical researchers: Chi-squared test and Fisher's exact test'. *Restor Dent Endod*. 42: 152-155.

King, R. 1969. *Historical Linguistics and Generative Grammar*. Englewood Cliffs, N.J.: Prentice-Hall.

Kingston, J. 2007. 'The phonetics-phonology interface'. In Paul de Lacy (ed.) *The Cambridge Handbook of Phonology*. Cambridge: Cambridge University Press. 401-434.

Kiparsky, P. 1965. 'Phonological change'. Doctoral Thesis. MIT.

Kiparsky, P. 1968. 'Linguistic universals and language change'. In E. Bach and R. Harms (eds) *Universals in Linguistic Theory*. New York: Holt, Rinehart & Winston. 171-202.

Kiparsky, P. 1985. 'Some consequences of Lexical Phonology'. *Phonology Yearbook* 2: 85-138.

Kiparsky, P. 1995. 'The phonological basis of sound change'. In J. Goldsmith (ed.) *The Handbook of Phonological Theory*. Oxford: Blackwell. 640-670.

Kiran, S., C.K. Thompson and N. Hashimoto. 2001. 'Training grapheme–phoneme conversion in patients with oral reading and naming deficits: A model-based approach'. *Aphasiology* 15: 855–876.

Kirschner, M. and J. Gerhart. 1998. 'Evolvability'. *Proceedings of the National Academy of Sciences* 95: 8420-8427.

Klima, E. and U. Bellugi. 1979. *The Signs of Language*. Cambridge, MA: Harvard University Press.

Knott, R., K. Patterson and J.R. Hodges. 1997. 'Lexical and semantic binding effects in short-term memory: Evidence from semantic dementia'. *Cogn Neuropsychol*. 14: 1165-1216.

Kohn, S.E. 1984. 'The nature of the phonological disorder in conduction aphasia'. *Brain and Language* 23: 97-115.

Kolb, B. and I.Q. Whishaw. 2008. 'Fundamentals of human neuropsychology'. 6<sup>th</sup> ed. New York: Worth Publishers.

Labov, W. 1966. *The Social Stratification of English in New York City*. Washington D.C.: Center for Applied Linguistics.

Labov, W. 1971. 'Methodology'. In W. Orr Dingall (ed.) *A Survey of Linguistic Science*. Maryland: University of Maryland.

Labov, W. and I. Rosenfelder. 2011. *The Philadelphia Neighborhood Corpus of LING 560 Studies, 1972-2010. With Support of NSF Contract 921643*. Philadelphia: University of Pennsylvania.

Labov, W., M. Yaeger and R. Steiner. 1972. *Quantitative Study of Sound Change in Progress*. Philadelphia: US Regional Survey.

Ladefoged, P. 2006. A Course in Phonetics. California: Thomson Wadsworth Corporation.

Laine, M. and N. Martin. 2006. *Anomia: Theoretical and Clinical Aspects*. New York: Psychology Press.

Laine, M., M. Juhola, J. Niemi and P. Koivuselkä-Sallinen. 1992. 'Simulating aphasic naming errors with a discrete stage serial word production model'. Poster presented at the Academy of Aphasia 30th Annual Meeting.

Lambon Ralph, M.A., K. Sage and J. Roberts. 2000. 'Classical anomia: A neuropsychological perspective on speech production'. *Neuropsychologica* 38: 186-202.

Langlois, R.N. 1992. 'External economies and economic progress: The case of the microcomputer industry'. *Business History Review* 66: 1-50.

Langlois, R.N. 1999. 'Modularity in technology, organization, and society'. Paper presented at INFORMS Philadelphia Meeting.

Langlois, R.N. 2000. 'Capabilities and vertical disintegration in process technology: The case of semiconductor fabrication experiment'. In N.J. Foss and P.L. Robertson (eds) *Resources, Technology and Strategy*. London: Routledge. 199-206.

Langlois, R.N. and P.L. Robertson. 1992. 'Networks and innovation in a modular system: Lessons from the microcomputer and stereo component industries'. *Research Policy* 21: 297-313.

Larson, A. and J.B. Losos. 1996. 'Phylogenetic systematics of adaptation'. In M.R. Rose and G.V. Lauder (eds) *Adaptation*. San Diego: Academic Press. 187-220.

Le Dorze, G. 1991. 'Etude des effets de l'intervention aupres d'un cas d'aphasie de conduction avec trouble d'acces au lexique'. Journal of Speech Language Pathology and Audiology 17: 5–14.

Le Dorze, G. and C. Pitts. 1995. 'A case study evaluation of the effects of different techniques for the treatment of anomia'. *Neuropsychological Rehabilitation* 5: 51–65.

Lesser, R. 1989. 'Some issues in the neuropsychological rehabilitation of anomia'. In X. Seron and G. Deloche (eds) *Cognitive Approaches in Neuropsychological Rehabilitation*. Hillsdale: Lawrence Erlbaum. 65-104.

Lesser, R. and L. Algar. 1995. 'Towards combining the cognitive neuropsychological and pragmatic in aphasia therapy'. *Neuropsychological Rehabilitation* 5: 1–16.

Leslie, A.M. 1994. 'ToMM, ToBY, and agency: Core architecture and domain specificity'. In L. Hirschfield and S. Gelman (eds) *Mapping the Mind: Domain Specificity in Cognition and Culture*. New York: Cambridge University Press. 119-148.

Lettvin, J.Y., H. Maturana, W.S. McCulloch and W.H. Pitts. 1959. 'What the frog's eye tells the frog's brain'. *Proceedings of the IRE* 47: 1940-1951.

Levelt, W. 1989. Speaking: From Intention to Articulation. Cambridge, MA: MIT Press.

Levelt, W.J.M. 1992. 'Accessing words in speech production: Stages, processes and representations'. *Cognition* 42: 1-22.

Levelt, W.J.M, A. Roelofs and A.S. Meyer. 1999. 'A theory of lexical access in speech production'. *Behavioral and Brain Sciences* 22: 1-75.

Li, E.C., K. Kitselman, D. Dusatko and C. Spinelli. 1988. 'The efficacy of PACE in the remediation of naming deficits'. *Journal of Communication Disorders* 21: 491–503.

Liberman, A.M., F.S. Cooper, D.P. Shankweiler and M. Studdert-Kennedy. 1967. 'Perception of speech code'. *Psychological Review* 74: 431-461.

Lichtheim, L. 1885. 'On aphasia'. Brain VII: 433-484.

Lissauer, H. 1890. 'Ein fall von seelenblinheit nebst einem beitrag zur theorie derselben'. *Arch Psychiatry* 21: 222-270. Lowell, S., P. Besson and A. Holland. 1995. 'The efficacy of a semantic cueing procedure on naming performance of adults with aphasia'. *American Journal of Speech-Language Pathology* 4: 109–114.

Lyons, J.C. 2001. 'Carving the mind at its (not necessarily modular) joints'. *British Journal for the Philosophy of Science* 52: 277-302.

MacSweeney, M., C. Capek, R. Campbell and B. Woll. 2008. 'The signing brain: The neurobiology of sign language'. *Trends in Cognitive Science* 12: 432-440.

Manasco, H. 2014. *Introduction to Neurogenic Communication Disorders*. Burlington, MA: Jones and Bartlett Publishers, Inc.

Marr, D. 1982. Vision. A Computational Investigation into the Human Representation and Processing of Visual Information. New York: Freeman and Co.

Marr, D. 1983. Vision. San Francisco: W.H. Freeman and Company.

Marshall, J., C. Pound, M. White-Thomson and T. Pring. 1990. 'The use of picture/word matching tasks to assist word-retrieval in aphasic patients'. *Aphasiology* 4: 167-184.

Marshall, R., R. Lazar and J. Mohr. 1998. 'Aphasia'. *Medical Update for Psychiatrists* 3: 132-138.

Marslen-Wilson, W.D. and L.K. Tyler. 1981. 'Central processes in speech understanding'. *Philosophical Transactions of the Royal Society B* B295: 317-332.

Martinet, A. 1960. Eléments de Linguistique General. Paris: Colin.

Mattingly, I.G., and A.M. Liberman. 1988. 'Specialized perceiving systems for speech and other biologically significant sounds'. In G.M. Edelman, W.E. Gall and W.M. Cowan (eds) *Auditory Function: Neurological Bases of Hearing*. New York: Wiley. 775-793.

Mattingly, I.G. and A.M. Liberman. 1990. 'Speech and other auditory modules'. In G.M. Edelman, W.E. Gall and W.M. Cowan (eds) *Signal and Sense: Local and Global Order in Perceptual Maps*. New York: Wiley. 501-519.

Maynard-Smith, J. and E. Szathmáry. 1995. *The Major Transitions in Evolution*. Oxford: Oxford University Press.

McCarthy, J. 1993. 'A case of surface constraint violation'. *Canadian Journal of Linguistics* 38: 169-195.

McCarthy, J. 1996. 'Remarks on phonological opacity in Optimality Theory'. In J. Lecarme, J. Lowenstamm and U. Shlonsky (eds) *Studies in Afroasiatic Grammar: Papers from the 2<sup>nd</sup> Conference on Afroasiatic Linguistics*. The Hague: Holland Academic Graphics. 215-243.

McCulloch, W.S. and W.H. Pitts. 1943. 'A logical calculus of the ideas immanent in nervous activity'. *Bulletin of Mathematical Biophysics* 5: 115-133.

McGilvray, J. (ed.). 2005. *The Cambridge Companion to Chomsky*. Cambridge: Cambridge University Press.

McMahon, A.M.S. 1991. 'Lexical phonology and sound change: the case of the Scottish vowel length rule'. *Journal of Linguistics* 27: 29-53.

McNeil, M.R., P.J. Doyle, K. Spencer, A.J. Goda, D. Flores and S. Small. 1997. 'A doubleblind, placebo controlled, pharmacological and behavioural treatment of lexical-semantic deficits in aphasia'. *Aphasiology* 11: 385–400.

McNeil, M.R., P.J. Doyle, K. Spencer, A.J. Goda, D. Flores and S. Small. 1998. 'Effects of training multiple form classes on acquisition, generalisation and maintenance of word retrieval in a single subject'. *Aphasiology* 12: 575–585.

Mervis, C.B., C.A. Morris, J. Bertrand and B. Robinson. 1999. 'Williams syndrome: Findings from an integrated program of research'. In H. Tager-Flusberg (ed.) *Neurodevelopmental Disorders*. Cambridge, MA: MIT Press. 65-110.

Mesulam, M.M. 1982. 'Slowly progressive aphasia without generalized dementia'. *Ann Neurol.* 11: 592-598.

Mesulam, M. (ed.). 2000. *Principles of Behavioral and Cognitive Neurology*. 2<sup>nd</sup> ed. Oxford: Oxford University Press.

Mielke, J. 2004. 'Moving beyond innate features: A unified account of natural and unnatural classes'. Paper presented at NELS 35.

Mielke, J. 2005. 'Ambivalence and ambiguity in laterals and nasals'. *Phonology* 22: 169-203.

Mielke, J. 2008. *The Emergence of Distinctive Features (Oxford Studies in Typology and Linguistic Theory)*. Oxford: Oxford University Press.

Meyer, A.S. and K. Bock. 1992. 'The tip-of-the-tongue phenomenon: Blocking or partial activation?' *Memory and Cognition* 20: 181-211.

Meyering, T.C. 1994. 'Fodor's modularity: A new name for an old dilemma'. *Philosophical Psychology* 7: 39-62.

Miceli, G., A. Amitrano, R. Capasso and A. Caramazza. 1996. 'The treatment of anomia resulting from output lexical damage: Analysis of two cases'. *Brain and Language* 52: 150-174.

Milroy, L. 1987. Observing and Analysing Natural Language. Oxford: Blackwell.

Milroy, L. and M. Gordon. 2003. *Sociolinguistics: Method and Interpretation*. Hoboken, NJ: Wiley.

Mitchum, C.C., M.L. Greenwald and R.S. Berndt. 2000. 'Cognitive treatments of sentence processing disorders: What have we learned?' *Neuropsychological Rehabilitation* 10: 311–336.

Mohr, B. 1971. 'Intrinsic variations in the speech signal'. *Phonetica* 23: 69-93.

Monsell, S. 1987. 'On the relation between lexical input and output pathways for speech'. In A. Allport, D. MacKay, W. Prinz and E. Scheerer (eds) *Language Perception and Production*. London: Academic Press. 273-311.

Morén, B. 2003a. 'Affricates, palatals and iotization in Serbian: Representational solutions to longstanding puzzles'. *Poljarnyj Vestnik* 6: 46-70.

Morén, B. 2003b. 'The Parallel Structures model of feature geometry'. *Working Papers of the Cornell Phonetics Laboratory* 15: 194-270.

Morén, B. 2006. 'Consonant-vowel interactions in Serbian: Features, representations and constraint interactions'. *Lingua* 116: 1198-1244.

Morén, B. 2007a. 'The division of labor between segment-internal structure and violable constraints'. In S. Blaho, P. Bye and M. Krämer (eds) *Freedom of Analysis?* Berlin: Mouton De Gruyter. 313-344.

Morén, B. 2007b. 'Minimalist/substance-free feature theory: Case studies and implications'. Course held at the 14th EGG Summer School.

Morén, B. 2007c. 'Minimalist/substance-free feature theory: Why and how'. Course held at the 14th EGG Summer School.

Moreton, E. and E. Thomas. 2007. 'Origins of Canadian Raising in voiceless-coda effects: A case study in phonologization'. *Laboratory Phonology* 9: 37-64.

Morgart, A.P. 2015. 'Lexical access in aphasia: Impacts of phonological neighborhood density on accuracy of word production'. M.A. Thesis. University of Iowa.

Morris, C.R. and C.H. Ferguson. 1993. 'How architecture wins technology wars'. *Harvard Business Review* 71: 86-96.

Morton, J. 1970. 'A functional model for memory'. In D.A. Norman (ed.) *Models of Human Memory*. New York: Academic Press. 203-254.

Morton, J. 1979. 'Facilitation in word-recognition: Experiments causing change in the logogen model'. In P. Kolers, M. Wrolstad and H. Bouma (eds) *Processing of Visible Language*. New York: Plenum. 259-268.

Morton, J. and K. Patterson. 1980. 'A new attempt at interpretation, or, an attempt at a new interpretation'. In M. Coltheart, K. Patterson and J.C. Marshall (eds) *Deep Dyslexia*. London: Routledge and Kegan Paul. 91-118.

Moss, L. 2001. 'Deconstructing the gene and reconstructing molecular developmental systems'. In S. Oyama, P.E. Griffiths, R.D. Gray (eds) *Cycles of Contingency: Developmental Systems and Evolution*. Cambridge, MA: MIT Press. 85-97.

Muffatto, M. 1999. 'Introducing a platform strategy in product development'. *International Journal of Production Economics* 60: 145-153.

Murat, A. 2015. 'The language of thought hypothesis'. In E.N. Zalta (ed.) *The Stanford Encyclopedia of Philosophy*. Last accessed on 30 August 2017, from: https://plato.stanford.edu/archives/fall2015/entries/language-thought/.

Murray, L.L. and L. Karcher. 2000. 'A treatment for written verb retrieval and sentence construction skills'. *Aphasiology* 14: 585-602.

Murre, J.M.J., K.S. Graham and J.R. Hodges. 2001. 'Semantic dementia: Relevance to connectionist models of long-term memory'. *Brain* 124: 647-675.

Naeser, M. and R. Hayward. 1978. 'Lesion localization in aphasia with cranial computed tomography and the Boston Diagnostic Aphasia Exam'. *Neurology* 28: 545-551.

Nettleton, J. and R. Lesser. 1991. 'Therapy for naming difficulties in aphasia: Application of a cognitive neuropsychological model'. *Journal of Neurolinguistics* 6: 139-157.

Newell, A. and H.A. Simon. 1976. 'Computer science as empirical inquiry: Symbols and search'. *Communications of the ACM* 19: 113-126.

Newmeyer, F. 2010. 'On comparative concepts and descriptive categories: A reply to Haspelmath'. *Language* 86: 688-695.

Nickels, L.A. 1997. *Spoken Word Production and its Breakdown in Aphasia*. Hove, UK: Psychology Press.

Nickels, L. 2002. 'Improving word finding: Practice makes (closer to) perfect?' *Aphasiology* 16: 1047-1060.

Nickels, L.A. and W. Best. 1996. 'Therapy for naming deficits (part II): Specifics, surprises and suggestions'. *Aphasiology* 10: 109–136.

Nickels, L. and D. Howard. 1995. 'Aphasic naming: What matters?' *Neuropsychologica* 33: 1281-1303.

Nickels, L., S. Byng and M. Black. 1991. 'Sentence processing deficits: A replication of therapy'. *British Journal of Disorders of Communication* 26: 175-199.

Nijhout, H.F. 1991. *The Development and Evolution of Butterfly Wing Patterns*. Washington, D.C.: Smithsonian Institution Press.

Obler, L.K. and M.L. Albert. 1986. The Action Naming Test. Boston: VA Medical Center.

Odden, D. 2006. 'Phonology ex nihilo'. Paper presented at the the Tromsø Phonology Project Group Meeting.

Odden, D. 2013. *Introducing Phonology (Cambridge Introductions to Language)*. Cambridge: Cambridge University Press.

Ohala, J. 1974. 'Experimental historical phonology'. In J. Anderson and C. Jones (eds)*Historical Linguistics II: Theory and Description in Phonology*. Amsterdam: North-Holland.353-379

Ohala, J. 1981. 'The listener as a source of language change'. In C. Masek, R. Hendrick and M. Miller (eds) *Papers from the Parasession on Language and Behavior*. Chicago: Chicago Linguistics Society. 178-203.

Ohala, J. 1998. 'A bibliography of the phonetics of sound change'. Ms. University of California at Berkeley.

Paivio, A. 1991. 'Dual coding theory: Retrospect and current status'. *Canadian Journal of Psychology* 45: 255-287.

Panksepp, J. and J.B. Panksepp. 2000. 'The seven sins of evolutionary psychology'. *Evolution and Cognition* 6: 108-131.

Panksepp, J. and J.B. Panksepp. 2001. 'A continuing critique of evolutionary psychology: Seven sins for seven sinners, plus or minus two'. *Evolution and Cognition* 7: 56-80.

Parter, M., N. Kashtan and U. Alon. 2007. 'Environmental variability and modularity of bacterial metabolic networks'. *BMC Evol. Biol.* 7: 169.

Pashek, G. 1997. 'A case study of gesturally cued naming in aphasia: Dominant versus nondominant hand training'. *Journal of Communication Disorders* 30: 349–366.

Pashek, G. 1998. 'Gestural facilitation of noun and verb retrieval in aphasia: A case study'. *Brain and Language* 65: 177–180.

Pavlov, I.P. 1955. Selected Works. Moscow: Foreign Languages Publishing House.

Peach, R.K. and L.P. Shapiro (eds). 2012. *Cognition and Acquired Language Disorders: An Information Processing Approach*. St. Louis, MO: Mosby.

Pedersen, P.M., K. Vinter and T.S. Olsen. 2001. 'Improvement of oral naming by unsupervised computerised rehabilitation'. *Aphasiology* 15: 151–169.

Phillips, A., H. Wellman and E. Spelke. 2002. 'Infants' ability to connect gaze and emotional expression to intentional action'. *Cognition* 85: 53-78.

Pick, A. 1892. 'Uber die beziehunden der senilen hirnatrophie zur aphasie'. *Prag Med Wschr* 17: 165-167.

Pickles, J. 2012. 'Auditory pathways: Anatomy and physiology'. *Handbook of Clinical Neurology* 129: 3-25.

Pierrehumbert, J. 2000. 'The phonetic grounding of phonology'. *Bulletin de la Communication Parlee* 5: 7-23.

Piggott, G. 1992. 'Variability in feature dependency: The case of nasality'. *Natural Language and Linguistic Theory* 10: 33-77.

van der Pijl, L. and C.H. Dodcon. 1966. *Orchid Flowers: Their Pollination and Evolution*. Coral Gables, Miami: University of Miami Press.

Pine, B.J. 1993. *Mass Customization: The New Frontier in Business Competition*. Boston, MA: Harvard Business School Press.

Pinker, S. 1994. The Language Instinct. New York: William Morrow.

Pinker, S. 1997. How the Mind Works. London: Penguin.

Pinker, S. and P. Bloom. 1990. 'Natural language and natural selection'. *Behavioral and Brain Sciences* 13: 707-784.

Popper, K. 1935. Logik der Forschung. Vienna: Julius Springer Verlag.

Popper, K. 1959. The Logic of Scientific Discovery. London: Hutchinson.

Powell, T.P.S. and V.B. Mountcastle. 1959. 'Some aspects of the functional organization of the cortex of the post-central gyrus of the monkey: A correlation of findings obtained in a single unit analysis with cytoarchitecture'. *Bulletin of the Johns Hopkins Hospital* 105: 133-162.

Prince, A. and P. Smolensky. 1993. 'Optimality Theory: Constraint interaction in generative grammar'. Ms. University of Colorado.

Prinz, J.J. 2006. 'Is the mind really modular?'. In R. Stainton (ed.) *Contemporary Debates in Cognitive Science*. Oxford: Blackwell. 22–36.

Pylyshyn, Z. 1984. Computation and Cognition. Cambridge, MA: MIT Press.

Quartz, S.R. and T.J. Sejnowski. 2002. *Liars, Lovers, and Heroes: What the New Brain Science Reveals about How We Become Who We Are.* New York: Harper Collins.

Raff, E.C. and R.A. Raff. 2000. 'Dissociability, modularity, evolvability'. *Evolution & Development* 2: 235-237.

Raff, R.A. 1996. *The Shape of Life – Genes, Development and Evolution of Animal Forms.* Chicago: University of Chicago Press.

Ramachandran, V.S. and S. Blakeslee. 1998. *Phantoms in the Brain: Probing the Mysteries of the Human Mind*. New York: William Morrow.

Raven, P.H. 2006. 'Sensory systems'. Last accessed on 24 July 2017, from: http://www.mhhe.com/biosci/genbio/raven6b/graphics/raven06b/other/raven06\_55.pdf.

Raymer, A. 2005. 'Naming and word-retrieval problems'. In L.L. LaPointe (ed.) *Aphasia and Related Language Disorders*. 3rd ed. New York: Thieme. 68-82.

Raymer, A. and T. Ellsworth. 2002. 'Response to contrasting verb retrieval treatments: A case study'. *Aphasiology* 16: 1031-1045.

Raymer, S., C. Thompson, B. Jacobs and H. Le Grand. 1993. 'Phonological treatment of naming deficits in aphasia: Model-based generalization analysis'. *Aphasiology* 7: 27-54.

Reed, D.P. 1996. 'Taking it all apart: Principles of network modularity'. In E. Noam and A. Nishuilleabhain (eds) *Private Networks Public Objectives*. Oxford: Elsevier, North-Holland. 117-194.

Rescorla, M. 2017. 'The computational theory of mind'. In E.N. Zalta (ed.) *The Stanford Encyclopedia of Philosophy*. Last accessed on 30 August 2017, from: https://plato.stanford.edu/archives/spr2017/entries/computational-mind/.

Riccoch, M.J., G.W. Humphreys, M. Coltheart and E. Funnell. 1988. 'Semantic systems or system? Neuropsychological evidence re-examined'. *Cogn Neuropsychol.* 5: 3-25.

Rice, K. 1993. 'A reexamination of the feature [sonorant]: The status of "sonorant obstruents". *Language* 69: 308–344.

Rice, K. and P. Avery. 1991. 'On the relationship between laterality and coronality'. In C. Paradis and J.F. Prunet (eds) *Phonetics and Phonology 2. The Special Status of Coronals. Internal and External Evidence.* San Diego, Academic Press. 101-124.

Ritter, B. 1998. Cells, Tissues, Organs, and Systems. Nelson: Nelson Science Technology.

Robbins, P. 2015. 'Modularity of mind'. In E.N. Zalta (ed.) *The Stanford Encyclopedia of Philosophy*. Last accessed on 25 July 2017, from: https://plato.stanford.edu/archives/sum2015/entries/modularity-mind/.

Robson, J., J. Marshall, T. Pring and S. Chiat. 1998. 'Phonological naming therapy in jargon aphasia: Positive but paradoxical effects'. *Journal of the International Neuropsychological Society* 4: 675–686.

Roelofs, A. 2000. 'WEAVER++ and other computational models of lemma retrieval and word-form encoding'. In L. Wheeldon (ed.) *Aspects of Language Production*. Sussex, UK: Psychology Press. 71-114.

Rose, M., J. Douglas and T. Maytas. 2002. 'The comparative effectiveness of gesture and verbal treatments for a specific phonologic naming impairment'. *Aphasiology* 16: 1001–1030.

Rosenfelder, I. 2005. *Canadian Raising in Victoria, B.C.: An Acoustic Analysis*. Heidelberg: Neuphilologischen Fakultät, Ruprecht-Karls-Universität.

Rosenfelder, I. 2007. 'Canadian raising in Victoria, B.C. An acoustic analysis. *Arbeiten aus Anglistik und Amerikanistik* 32: 257-284.

Rumelhart, D. and J. McLelland. 1986. *Parallel Distributed Processing: Explorations in the Microstructure of Cognition*. Cambridge, MA: MIT Press.

Sacchet, C., S. Byng, J. Marshall and C. Pound. 1999. 'Drawing together: Evaluation of a therapy programme for severe aphasia'. *International Journal of Language and Communication Disorders* 34: 265–289.

Samuel, A.G. 1987. 'Lexical uniqueness effects on phoneme restoration'. *Journal of Memory and Language* 26: 36-56.

Samuels, R. 1998. 'Evolutionary psychology and the massive modularity hypothesis'. *British Journal for the Philosophy of Science* 49: 575-602.

Samuels, R. 2000. 'Massively modular minds: Evolutionary psychology and cognitive architecture'. In P. Carruthers and A. Chamberlain (eds) *Evolution and the Human Mind: Modularity, Language and Meta-Cognition*. Cambridge: Cambridge University Press. 13-46.

Samuels, R. 2006. 'Is the human mind massively modular?'. In R. Stainton (ed.) *Contemporary Debates in Cognitive Science*. Oxford: Blackwell. 37–56.

Sanchez, R. 1999. 'Modular architectures in the marketing process'. *Journal of Marketing* 63: 92-111.

Sanchez, R. and J. T. Mahoney. 1996. 'Modularity, flexibility, and knowledge management in product and organizational design'. *Strategic Management Journal* 17: 63-76.

Sanderson, S. W. and M. Uzumeri. 1997. Managing Product Families. Chicago: Irwin.

Sartori, G. and R. Job. 1988. 'The oyster with four legs: A neuro-psychological study on the interaction of semantic and visual information'. *Cognitive Neuropsychology* 5: 105-132.

Saussure, F. 1966. Course in General Linguistics. New York: McGraw-Hill Book Co.

Scheer, T. 2014. 'Spell-Out, post-phonological'. In E. Cyran and J. Szpyra-Kozłowska (eds) *Crossing Phonetics-Phonology Lines*. Newcastle upon Tyne: Cambridge Scholars Publishing. 255–275.

Schlosser, G. and G.P. Wagner (eds). 2004. *Modularity in Development and Evolution*. Chicago: University of Chicago Press.

Schnelle, H. 2010. Language in the Brain. Cambridge: Cambridge University Press.

Scholl, B.J. and A.M. Leslie. 1999. 'Modularity, development and "theory of mind". *Mind and Language* 14: 131-153.

Schwartz, B.D. 1986. 'The epistemological status of second language acquisition'. *Second Language Research* 2: 120-159.

Schwartz, M.F., O.S. Marin and E.M. Saffran. 1979. 'Dissociations of language function in dementia: A case study'. *Brain Lang* 7: 277-306.

Scobbie, J.M., N. Hewlett and A. Turk. 1999. 'Standard English in Edinburgh and Glasgow: The Scottish vowel length rule revealed'. In P. Foulkes and G.J. Docherty (eds) *Urban Voices: Accent Studies in the British Isles*. London: Arnold. 230- 245.

Seeley, T. 1995. *The Wisdom of the Hive: The Social Physiology of Honey Bee Colonies*. Cambridge, MA: Harvard University Press.

Segal, G. 1996. 'The modularity theory of minds'. In P. Carruthers and P.K. Smith (eds) *Theories of Theories of Mind*. Cambridge: Cambridge University Press. 141-157.

Seinfield, J.H. and N. Spyros. 2006. *Atomospheric Chemistry and Physics: From Air Pollution to Climate Change*. 2<sup>nd</sup> ed. Hoboken, NJ: John Wiley & Sons, Inc.

Senghas, A., S. Kita and A. Özyürek. 2004. 'Children creating core properties of language: Evidence from an emerging sign language in Nicaragua'. *Science* 305: 1779-1782.

Shallice, T. 1998. *From Neuropsychology to Mental Structure*. Cambridge: Cambridge University Press.

Shattuck-Hufnagel, S. 1979. 'Speech errors as evidence for a serial-order mechanism in sentence production'. In W.E. Cooper and W.C.T. Walker (eds) *Sentence Processing: Psycholinguistic Studies Presented to Merrill Garrett.* London: Erlbaum. 295-342.

Shattuck-Hufnagel, S. 1987. 'The role of wordonset consonants in speech production planning: New evidence from speech error patterns'. In E. Keller and M. Gopnik (eds) *Motor and Sensory Processes of Language*. Hillsdale NJ: LEA. 17-51.

Shiestl, F.P., M. Ayasse, H.F. Paulus, C. Löfstedt, B. Hannson, F. Ibarra and W. Francke. 2000. 'Sex pheromone mimicry in the early spider orchid (Ophrys sphegodes): Patterns of hydrocarbons as the key mechanism for pollination by sexual deception'. *J. Comp. Physiol.* 186: 567–574.

Silva-Corvalán, C. 2001. *Sociolingüística y Pragmática del Español*. Washington, D.C.: Georgetown University Press.

Simon, H.A. 1962. 'The architecture of complexity'. *Proceedings of the American Philosophical Society* 106: 467-482.

Simon, H.A. 1969. The Sciences of the Artificial. 2nd ed. Cambridge, MA: MIT Press.

Simon, H. 2005. 'The structure of complexity in an evolving world: The role of near decomposability'. In W. Callebaut and D. Rasskin-Gutman (eds) *Modularity: Understanding the Development and Evolution of Natural Complex Systems*. Massachusetts: MIT Press.

Simon, H. and A. Ando. 1961. 'Aggregation of variables in dynamic systems'. *Econometrica* 29: 111-138.

Skinner, B.F. 1957. Verbal Behavior. Acton, MA: Copley Publishing Group.

Smith, N. and I.M. Tsimpli. 1996. 'Modules and quasi-modules: Language and theory of mind in a polyglot savant'. *UCL Working Papers in Linguistics* 8: 1-18.

Snowden, J.S., P.J. Goulding and D. Neary. 1989. 'Semantic dementia: A form of circumscribed cerebral atrophy'. *Behavioral Neurology* 2: 167-182.

Spelke, E. 1994. 'Initial knowledge: Six suggestions'. Cognition 50: 431-445.

Spencer, K.A., P.J. Doyle, M.R. McNeil, J.L. Wambaugh, G. Park and B. Carroll. 2000. 'Examining the facilitative effects of rhyme in a patient with output lexicon damage'. *Aphasiology* 14: 567–584.

Sperber, D. 1994. 'The modularity of thought and the epidemiology of representations'. In L.A. Hirschfield and S.A. Gelman (eds) *Mapping the Mind: Domain Specificity in Cognition and Culture*. New York: Cambridge University Press. 39-67.

Sperber, D. 1996. Explaining Culture: A Naturalistic Approach. Oxford: Blackwell.

Sperber, D. 2002. 'In defense of massive modularity'. In E. Dupoux (ed.) *Language, Brain and Cognitive Development*. Cambridge, MA: MIT Press. 47-57.

Sperber, D., D. Premack and A. Premack (eds). 1995. *Causal Cognition: A Multidisciplinary Debate*. New York: Oxford University Press.

Stampe, D. 1972. 'A dissertation on natural phonology'. Doctoral Thesis. University of Chicago.

Stemberger, J.P. 1985. 'An interactive activation model of language production'. In A.W. Ellis (ed.) *Progress in the Psychology of Language. Vol. 1.* Hillsdale, NJ: Lawrence Erlbaum Associates Inc. 143-186.

Sterenly, K. 1990. The Representational Theory of Mind. Oxford: Basil Blackwell.

Sbisa, S., J. D'Andrea, C. Semenza and P. Tabossi. 2001. 'Rehabilitating anomia: A case study'. *Brain & Language* 79: 46–47.
Steriade, D. 1987. 'Redundant Values'. In A. Bosch, B. Need and E. Schiller (eds) *Papers* from the Parasession on Metrical and Autosegmental Phonology. Chicago: Chicago Linguistics Society. 339-362.

Steriade, D. 1995. 'Underspecification and markedness'. In J. Goldsmith (ed.) *The Handbook* of *Phonological Theory*. Cambridge, MA: Blackwell. 114-174.

Stokoe, W. 1960. 'Sign language structure: An outline of the visual communication systems of the American deaf'. *Studies in Linguistics: Occasional Papers* 8: Buffalo: Dept. of Anthropology and Linguistics, University of Buffalo.

Stromswold, K. 1994. 'Language comprehension without language production: Implications for theories of language acquisition'. Paper presented at the 18th Annual Boston University Conference on Language Development.

Stroop, J.R. 1935. 'Studies of interference in serial verbal reactions'. *Journal of Experimental Psychology* 18: 643-662.

Sugishita, M., K. Seki, S. Kabe and K. Yunoki. 1993. 'A material-control single-case study of the efficacy of treatment for written and oral naming difficulties'. *Neuropsychologia* 31: 559–569.

Sukhatme, S.P. 2008. *Solar Energy: Principles of Thermal Collection and Storage*. 3<sup>rd</sup> ed. New Dehli: Tata McGraw-Hill Publishing Company Limited.

Swinburn, K., G. Porter and D. Howard. 2004. *Comprehensive Aphasia Test*. New York: Psychology Press.

Symons, D. 1987. 'If we're all Darwinians, what's the fuss about?'. In C. Crawford, M. Smith and D. Krebs (eds) *Sociobiology and Psychology: Ideas, Issues and Applications*. Hillsdale, NJ: Erlbaum. 121-45.

Tager-Flusberg, H. (ed.). 1999. Neurodevelopmental Disorders. Cambridge, MA: MIT Press.

Tan, K.H. and R. Nishida. 2000. 'Mutual reproductive benefits between a wild orchid, *Bulbophyllum patens*, and *Bactrocera* fruit flies via a floral synomone'. *Journal of Chemical Ecology* 26: 533-546.

Thomas, E.R. 1991. 'The origins of Canadian raising in Ontario'. *Canadian Journal of Linguistics* 36: 147-170.

Thompson, C.K. and K.P. Kearns. 1981. 'An experimental analysis of acquisition, generalization, and maintenance of naming behavior in a patient with anomia'. In R.H. Brookshire (ed.) *Clinical Aphasiology Conference Proceedings*. Minneapolis, MN: BRK Publishers. 132-144.

Thompson, C.K., H.R. Hall and C.E. Sison. 1986. 'Effects of hypnosis and imagery training on naming behaviour in aphasia'. *Brain and Language* 28: 141–153.

Thurgood, G. and H. Javkin. 1975. 'An acoustic explanation of sound change: \*-*at* to -*e*, \*-*ap* to -*o*, and \*-*ak* to -*œ* in Lisu'. *Journal of Phonetics* 3: 161-165.

Tooby, J. and L. Cosmides. 1992. 'The psychological foundations of culture'. In J. Barkow, L. Cosmides and J. Tooby (eds) *The Adapted Mind*. New York: Oxford University Press. 19-136.

Tooby, J. and L. Cosmides. 1995. 'Foreword'. S. Baron-Cohen (ed.) *Mindblindness*. Cambidge, MA: MIT Press. xi-xviii.

Trubetzkoy, N. 1939. *Grundzüge der Phonologie. Travaux du Cercle Lingquistique de Prague 7.* Prague: Le Cercle Linguistique de Prague.

Trubetzkoy, N. 1969. Principles of Phonology. Berkeley: University of California Press.

Vaux, B. 2008. 'Why the phonological component must be serial and rule-based'. In B. Vaux and A. Nevins (eds) *Rules, Constraints and Phonological Phenomena*. Oxford: Oxford University Press. 243-248.

Vennemann, T. 1972a. 'Phonetic analogy and conceptual analogy'. In T. Vennemann and T. Wilbur (eds) *Schuchardt, the Neogrammarians, and the Transformational Theory of Phonological Change.* Frankfurt: Athenäum. 181-204.

Vennemann, T. 1972b. 'Rule inversion'. Lingua 29. 209-242

Vennemann, T. 1974. 'Words and syllables in natural generative phonology'. *Papers from the Parasession on Natural Phonology* 10: 346-374.

Vigliocco, G., T. Antonini and M.F. Garrett. 1997. 'Grammatical gender is on the tip of Italian tongues'. *Psychological Science* 8: 314-317.

Vigliocco, G., D. Vinson, R.C. Martin and M.F. Garrett. 1999. 'Is "count" and "mass" information available when the noun is not? An investigation of tip of the tongue states and anomia'. *Journal of Memory and Language* 40: 534-558.

Wagner, G.P. and L. Altenberg. 1996. 'Perspective: Complex adaptations and the evolution of evolvability'. *Evolution* 50: 967-976.

Waltz, J.A., B.J. Knowlton, K.J. Holyoak, K.B. Boone, F.S. Mishkin, M. de Menezes Santos, C.R. Thomas and B.L. Miller. 1999. 'A system for relational reasoning in human prefrontal cortex'. *Psychol Sci.* 10: 199-125.

Wambaugh, J.L. and P.J. Doyle. 1994. 'Treatment for acquired apraxia of speech: A review of efficacy reports'. *Clinical Aphasiology* 22: 231–243.

Wang, W. and C. Fillmore. 1961. 'Intrinsic cues and consonant perception'. *Journal of Speech and Hearing Research* 4: 130-136.

Warren, R.M. and R.P. Warren. 1970. 'Auditory illusions and confusions'. *Scientific American* 223: 30-36.

Warrington, E.K. 1975. 'The selective impairment of semantic memory'. *Q J Exp Psychol*. 27: 635-657.

Warrington, E. and R. McCarthy. 1983. 'Category specific access dysphagia'. *Brain* 106: 859-878.

Warrington, E. and T. Shallice. 1984. 'Category specific impairments'. Brain 107: 829-854.

Wernicke, C. 1874. Der Aphasische Symptomenkomplex: Eine Psychologische Studie auf Anatomischer Basis. Breslau: Cohn & Weigert.

West-Eberhard, M.J. 1996. 'Wasp societies as microcosms for the study of development and evolution'. In M.J. West-Eberhard and S. Turillazzi (eds) *Natural History and Evolution of Paper Wasps*. Oxford: Oxford University Press. 290-317.

West-Eberhard, M.J. 2003. *Developmental Plasticity and Evolution*. New York: Oxford University Press.

Whitworth, A., J. Webster and D. Howard. 2013. *A Cognitive Neuropsychological Approach to Assessment and Intervention in Aphasia: A Clinician's Guide*. 2nd ed. Hove: Psychology Press.

Wilhem, B. 1997. 'Platform and modular concept at Volkswagen – their effect on the assembly process'. In K. Shimokawa, U. Jurgens and T. Fujimoto (eds) *Transforming Auto Assembly*. Berlin: Springer. 146-156.

Williams, G.C. 1966. *Adaptation and Natural Selection*. Princeton, NJ: Princeton University Press.

Williamson, D.J.G., B. Crosson, L.J.G. Rothi, S. Shurla, K.M. Heilman and S.E. Nadeau. 1995. 'Regional cerebral blood flow during the generation of language'. *Journal of the International Neuropsychological Society* 1: 37.

Wilson, D. 2005. 'New directions for research on pragmatics and modularity'. *Lingua* 115: 1129-1146.

Wilson, R.A. 2008. 'The drink you're having when you're not having a drink'. *Mind & Language* 23: 273-283.

Wimsatt, W.C. and J.C. Schank. 2004. 'Generative entrenchment, modularity and evolvability: When genic selection meets the whole organism'. In G. Schlosser and G. Wagner (eds) *Modularity in Development and Evolution*. Chicago: University of Chicago Press. 359-394.

Woodward, A.L. 1998. 'Infants selectively encode the goal object of an actor's reach'. *Cognition* 69: 1-34.

Woollams, A.M., E. Cooper-Pye, J.R. Hodges and K. Patterson. 2008. 'Anomia: A doubly typical signature of semantic dementia'. *Neuropsychologia* 46: 2503-2514.

Wright, R. 2004. 'A review of perceptual cues and cue robustness'. In B. Hayes, R. Kirchner and D. Steriade (eds) *Phonetically Based Phonology*. Cambridge: Cambridge University Press. 34-57.

Zhang, X. 1996. 'Vowel systems of the Manchu-Tungus languages of China. Doctoral Thesis. University of Toronto.